GW - __001___

N & S Aeration Lagoons Closure Certification Report

CLOSURE PLANS

September 2009

Chavez, Carl J, EMNRD

From:Chavez, Carl J, EMNRDSent:Tuesday, April 26, 2011 11:31 AMTo:Kieling, John, NMENVCc:VonGonten, Glenn, EMNRDSubject:March 29, 2011 Public Notice No. 11-01 for the Final Closure Plan North Aeration Lagoons
(Revised January 2011)

Mr. Kieling:

Please find below the New Mexico Oil Conservation Division's (OCD) public comments based on the above subject.

- 1) Should this closure plan reference closure certification, a closure report, and a survey plat?
- 2) The OCD notices that they are sending any waste associated with and excavated from beneath the ponds to the San Juan County Regional Landfill as special waste as long as it is non-hazardous. OCD discharge requirements for approval of the special waste to the San Juan Regional Landfill as Special Wastes are as follows:

19.15.35.8 DISPOSAL OF CERTAIN NON-DOMESTIC WASTE AT SOLID WASTE FACILITIES:

C. The following provisions apply to the types of waste described below as specified.

(3) A person may dispose of the following wastes on a case-by-case basis with the division's approval: (c) contaminated soil other than petroleum contaminated soil.

D. Testing.

(1) The person applying for division approval to dispose of waste in a solid waste facility shall conduct testing required by 19.15.35.8 NMAC according to the Test Methods for Evaluating Solid Waste, EPA No. SW-846 and shall direct questions concerning the standards or a particular testing facility to the division.

(2) The testing facility shall conduct testing according to the test method listed:

(a) TPH: EPA method 418.1 or 8015 (DRO and GRO only) or an alternative, division approved hydrocarbon analysis;

(b) TCLP: EPA Method 1311 or an alternative hazardous constituent analysis approved by the division;

(c) paint filter test: EPA Method 9095A;

(d) ignitability test: EPA Method 1030;

(e) corrosivity: EPA Method 1110;

(f) reactivity: test procedures and standards the division establishes on a case-by-case basis; and

(g) NORM. 20.3.14 NMAC.

(3) To be eligible for disposal pursuant to 19.15.35.8 NMAC, the concentration of substances the testing facility identifies during testing shall not exceed the following limits:

(a) benzene: 9.99 mg/kg;
(b) BTEX: 499.99 mg/kg (sum of all);
(c) TPH: 1000 mg/kg;
(d) hazardous air pollutants: the standards set forth in NESHAP; and
(e) TCLP:
(i) arsenic: 5 mg/l,
(ii) barium: 100 mg/l,
(iii) cadmium: 1 mg/l,
(iv) chromium: 5 mg/l,
(v) lead: 5 mg/l,
(vi) mercury: 0.2 mg/l,
(vii) selenium: 1 mg/l, and
(viii) silver: 5 mg/l.

If the OCD Discharge Permit (GW-001) waste meets the above criteria, the OCD may approve the excavated waste going to the proposed landfill under the OCD Water Quality Control Commission (WQCC) Discharge Permit.

Thank you for your consideration in this matter.

Carl J. Chavez, CHMM New Mexico Energy, Minerals & Natural Resources Dept. Oil Conservation Division, Environmental Bureau 1220 South St. Francis Dr., Santa Fe, New Mexico 87505 Office: (505) 476-3490 Fax: (505) 476-3462 E-mail: CarlJ.Chavez@state.nm.us Website: <u>http://www.emnrd.state.nm.us/ocd/index.htm</u> "Why not Prevent Pollution; Minimize Waste; Reduce the Cost of Operations; & Move Forward with the Rest of the Nation?" To see how, go to "Pollution Prevention & Waste Minimization" at: http://www.emnrd.state.nm.us/ocd/environmental.htm#environmental)



BILL RICHARDSON Governor

DIANE DENISH Lieutenant Governor

NEW MEXICO ENVIRONMENT DEPARTMENT

Hazardous Waste Bureau

2905 Rodeo Park Drive East, Building 1 Santa Fe, New Mexico 87505-6303 Phone (505) 476-6000 Fax (505) 476-6030 www.nmenv.state.nm.us



RON CURRY Secretary

SARAH COTTRELL Deputy Secretary

CERTIFIED MAIL - RETURN RECEIPT REQUESTED

October 14, 2010

Mr. Randy Schmaltz Environmental Manager Western Refining, Southwest, Inc. Bloomfield Refinery P.O. Box 159 Bloomfield, New Mexico 87413

RE: NOTICE OF DISAPPROVAL FINAL CLOSURE PLAN NORTH AND SOUTH AERATION LAGOONS WESTERN REFINING SOUTHWEST, INC., BLOOMFIELD REFINERY EPA ID # NMD089416416 HWB-WRB-10-007

Dear Mr. Schmaltz:

The New Mexico Environment Department (NMED) has reviewed Western Refining Southwest, Inc., Bloomfield Refinery (Western) *Final Closure Plan North and South Aeration Lagoons* (Closure Plan) dated May 2010. NMED hereby issues this Notice of Disapproval (NOD). The Permittee must address the following comments.

Comment 1

In Section 2.2 (ABT Unit Operations), page 6, Western describes the liner system for the Aeration Lagoons (ABT Units) from top to bottom. Western states in bullet 2 that the liner includes "[a] geonet for collecting leaks that drain to a sump equipped with a 6" observation pipe." Western must revise the Closure Plan to include details for the sump, including the design, dimensions, observation pipe and location (e.g., the sump is a concrete structure x feet by x feet deep, and located x feet below ground surface adjacent to the South ABT Unit). See also Comment 4.

Comment 2

In Section 3.3 (Flushing of Leachate Collection System), page 12, Western states "[p]ursuant to the previous Closure Plan, after repairs to the upper liner of the South ABT unit were completed, the geonet between the upper 100-ml liner and the lower 60-ml liner was flushed with clean water. The flush water was sampled using a bailer in the 6" observation pipe. Analytical results indicated that the flush water did not exhibit any hazardous characteristics." Although the flush water did not exhibit hazardous characteristics, Western must indicate if the flush water was analyzed for hazardous constituents, and if so, list the detected constituents. Western must include appropriate documentation (e.g., laboratory reports). Western must revise the Closure Plan accordingly.

Comment 3

In Section 4.1.1 (Sludge/Sediment Removal), page 14, Western states "[t]he sludge /sediment that remains in the ABT units above the top liner after removal [of] the free liquids will be allowed to dry for up to four weeks...At the conclusion of these activities, the material will be sampled for hazardous characteristics in accordance with 40 CFR Part 261, Subpart C - Characteristics of Hazardous Waste, including reactivity, corrosivity, ignitability, TCLP RCRA metals, TCLP semi-volatiles (SVOCs), and total volatiles (VOCs). If the material is non-hazardous, then it may be disposed at a landfill permitted by the NMED to accept Special Waste (e.g., the San Juan County Regional Landfill)." The disposal facility may require analyses in addition to what is discussed above (e.g., total RCRA metals, TCLP volatiles). Western must comply with all waste characterization required by the disposal facility.

The Closure Plan discusses the removal of liquids, sludges, sediments, various liner materials, and investigation derived wastes (media) that will be analyzed for hazardous characteristics in accordance with 40 CFR 261 Subpart C. Western must revise the Closure Plan to clearly indicate that all media determined to be non hazardous and to be disposed offsite will meet the waste acceptance requirements of the disposal facility. The waste disposition must also be documented in the closure report.

Comment 4

In Section 4.1.2 (RCRA Liner Removal), page 14, Western states "[t]he RCRA liners, which include an upper 100 mil HDPE liner, a geonet, and a lower 60 mil HDPE liner, will be removed and disposed at a landfill permitted by the NMED to accept Special Waste (e.g., the San Juan County Regional Landfill)," and on page 15 states "[t]he liners and geonet will be cut into manageable sized pieces and then rolled/folded to facilitate loading into trucks/roll-off boxes for transport off-site disposal." Western does not discuss the disposal of the sump or six inch pipe associated with these liners. Western must revise this section of the Closure Plan to address the removal and disposal of the sump piping and any other associated debris. See also Comment 1.

Comment 5

In Section 4.1.2 (RCRA Liner Removal), page 15, Western states "[i]f the liquid is nonhazardous, then it will be disposed through the on-site permitted discharge system." Western does not identify the on-site permitted discharge system nor address its associated components.

Western must revise the Closure Plan to discuss the on-site permitted discharge system and address its associated components (e.g., fluids will flow through the API separator to the injection well or flow through the API separator and the benzene strippers to the injection well). Western must also identify the sampling requirements associated with the "on-site permitted discharge system," if applicable. This comment applies to all sections of the Closure Plan that reference the on-site permitted discharge system.

Comment 6

In Section 4.1.3 (Non-RCRA Liner/Leachate Collection System Removal), page 15, Western discusses the removal of the "non-RCRA Liner" and states "[t]he uppermost layer beneath the RCRA lower 60 mil liner is a composite geotextile/geonet, which will be cut into manageable pieces and placed into roll-off boxes for off-site disposal. Beneath this composite geotextile/geonet layer is a 6" layer of cement amended sand. The sand layer will be excavated and stockpiled on-site pending waste characterization sampling. Beneath the sand layer is a 100 mil HDPE liner, which will be removed and handled with the previously removed composite geotextile/geonet layer. The lowermost layer consists of approximately 6 inches of bentonite amended soil with a French drain system." Western must revise the Closure Plan to address the removal of the four-inch observation pipe associated with the composite geotextile/geonet described in Section 2.2 (ABT Unit Operations).

Comment 7

In Section 4.1.3 (Non-RCRA Liner/Leachate Collection System Removal), page 15, Western states "[t]he French drain system will be checked for the presence of liquids prior to removal. Any liquids that are present will be containerized. The liquid will be sampled and analyzed for hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste. If the liquid exhibits hazardous characteristics, then it will be disposed offsite as hazardous waste. If the liquid is nonhazardous, then it will be disposed through the facilities on-site permitted discharge." If liquids are detected and determined to be non-hazardous, there is still a potential for the liquids to contain contaminants. If liquids are found in the French drain system and determined to be non-hazardous, Western must meet the discharge limits for the on-site permitted discharge system. Western must revise the Closure Plan to discuss disposal of liquids in the on-site discharge system and the associated discharge concentration limits. Western must also explain the additional analysis requirements if liquids are detected in the French drain system. See also Comment 5.

Comment 8

In Section 4.1.3 (Non-RCRA Liner/Leachate Collection System Removal), page 15, Western states "[i]f the liquid is nonhazardous, then it will be disposed through the facilities on-site permitted discharge. Soil will be physically removed from the drain system piping and if the liquid present in the drain system is characteristically hazardous, then the piping will be triple rinsed with potable water prior to offsite disposal." Western indicates that soil will be physically removed, but does not discuss the disposition of the soil once it has been removed. Western must revise the Closure plan to address the disposition of soils removed from the French drain system and indicate how the soil(s) will be characterized.

Comment 9

In Section 4.1.3 (Non-RCRA Liner/Leachate Collection System Removal), page 16, Western states "[i]f all concentrations of constituents are below the applicable NMED residential soil screening levels, then the soil may be reused for backfill." Meeting the NMED numerical residential soil screening levels does not necessarily allow Western to reuse the soil as backfill. The soils must also meet the cumulative target residential risk of 1E-05 for carcinogens and a hazard index of 1 for noncarcinogens. If the soil is used as backfill, Western must be able to demonstrate it meets the requirements of the NMED Technical Background for Development of Soil Screening Levels, as updated. Western must revise the Closure Plan to state excavated soils will be used as backfill only if it is demonstrated that the soils meet residential cleanup standards, the cumulative target residential risk of 1E-05 for carcinogens.

Comment 10

In Section 4.2 (Soil Investigation), page 16, Western discusses the soil borings and the collection of discrete soil samples from various depths. In addition to the sampling described, Western must also collect a sample from the bottom of each boring for laboratory analyses. Western must revise the Closure Plan accordingly.

Comment 11

In Section 4.2.4 (Collection and Management of Investigation Derived Waste), page 20, Western states "[a]ll decontamination water will be characterized prior to disposal unless it is disposed in the refinery wastewater treatment system upstream of the API Separator." Western also states in Appendix A (Management of Investigation Derived Waste) "[t]he fluids will be pumped directly into suitable storage containers (e.g., labeled 55-gallon drums), which will be located at satellite accumulation areas until the fluids are disposed in the refinery wastewater treatment system upstream of the API separator." With the removal of the ABT Units, Western's wastewater treatment system will not operate as it did in the past. Western must revise all applicable sections of the Closure Plan to describe all components of the wastewater treatment system that will be in operation once the ABT Units are taken out of service. Western must also describe how it will demonstrate that the decontamination water and fluids stored in containers are acceptable for discharge to the wastewater treatment system once the ABT Units are not in operation. Prior to closure of the ABT units, Western must clean and remove all K051 sludges from the API Separator. Western must revise the Closure Plan accordingly.

Comment 12

In Section 4.2.7 (Chemical Analyses), page 21, Western states "[s]oil samples will be analyzed by the following methods..." In the revised Closure Plan, Western must clarify if the methods also apply to the sludge/sediment samples as described in Section 4.1.1 or only the soil samples that will be collected from the borings and excavation limits shown in Figure 3 (Sample Location Map).

Ţ

Comment 13

In Section 4.2.8 (Data Quality Objectives), page 21, Western states "[m]ethod detection limits should be 20 % or less of the applicable background levels, cleanup standards and screening levels." If Western is unable to achieve the method detection limits, an explanation why they were not achieved must be included in the closure report. Western must revise this section indicating that an explanation will be provided if the method detection limits are not achieved.

Comment 14

In Section 4.3 (Soil Removal Action), page 22, Western states "[t]he preferred method to address any such releases from the ABT Units is to remove and dispose the impacted soils at a permitted off-site landfill so as to obtain Corrective Action Complete Without Controls [CACWC] and meet the requirements of 40 CFR§265.228 (a)(1) for clean closure. If the volume or depth of impacted soils is such that a complete removal action is financially or technically impracticable, then a revised Closure Plan will be submitted in accordance with 40 CFR §265.228 (a)(2)." Western must revise the Closure Plan to state that it will provide a demonstration of impracticability and proposed additional phases of work or post-closure care will be discussed in the Closure Report, rather than amending the Closure Plan at the time of discovery. NMED will determine the appropriate course of action after its review of the Closure Plan.

Comment 15

In Section 4.3.1 (Soil Excavation), page 23, Western states "[s]oil containing concentrations of constituents above the applicable NMED residential screening levels will be excavated for off site disposal. The soil will be excavated using back hoes, track hoes, long-reach excavators, or similar equipment. The excavated soil will either be stockpiled on poly sheeting within the footprint of the ABT Units or placed directly into roll-off boxes. The soil will be sampled and analyzed for waste characterization at a minimum of one sample per 20 cubic yards. The samples will be analyzed for hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste, including reactivity, corrosivity, ignitability, TCLP RCRA metals, TCLP SVOCs, and total VOCs." Western must revise the Closure Plan to clarify that samples analyzed for VOC analyses will be collected as discrete samples and include a statement that soil characterization will meet the requirements of the disposal facility.

Comment 16

Western describes confirmation sampling in Section 4.3.2. This Section does not indicate if the samples will be collected as discrete or composite samples, nor does it propose analytical methods. Western must revise the Closure Plan to discuss the sample collection methods as well as the applicable analytical methods.

Comment 17

In Section 4.4 (Closure Certification), page 23, Western states "[u]pon completion of all activities, a Closure Certification Report will be prepared in accordance with 40 CFR §265.115 Certification of Closure. The certification will [describe] how the ABT Units were closed in accordance with the approved Closure Plan." Prior to certification of closure, Western must first submit a closure report that summarizes all work and presents all data related to closure. Once the closure report is approved by NMED, a certification of closure can then be completed in

accordance with 40 CFR 265.115. Western must revise the Closure Plan to clarify that a closure report will be submitted and upon NMED approval, certification of closure will then be completed. Appendix B (Closure Cost Estimate) may need to be revised to include the cost associated with preparation of a closure report. Western must revise the Closure Plan accordingly.

Western must address all comments contained in this NOD and submit a revised Closure Plan to NMED on or before January 14, 2011. The revised Closure Plan must be submitted with a response letter that details where all revisions have been made, cross-referencing NMED's numbered comments. In addition, an electronic version of the revised work plan must be submitted that identifies where all changes have been made in redline strikeout format. If you have any questions regarding this letter, please contact Hope Monzeglio of my staff at (505) 476-6045.

Sincerely,

7-

James P Bearzi Chief Hazardous Waste Bureau

JPB:hm

cc: J. Kieling, NMED HWB
D. Cobrain, NMED HWB
C. Chavez, OCD
A. Hains, Western
File: HWB-WRB-10-007 and Reading 2010



RECEIVED OCD

May 19, 2010

2010 MAY 24 P 1: 19

James Bearzi, Bureau Chief New Mexico Environmental Department Hazardous Waste Bureau 2905 Rodeo Park Drive East, Building 1 Santa Fe, New Mexico 87505-6303

Re: NOTICE OF DISAPPROVAL NORTH ANS SOUTH AERATION LAGOONS CLOSURE CERTIFICATION REPORT WESTERN REFINING SOUTHWEST INC., BLOOMFIELD REFINERY EPA ID# NMD089416416 HWB-GRCB-09-007

Dear Mr. Bearzi:

Western Refining Southwest Inc. - Bloomfield Refinery submits the enclosed plan and cost estimate for final closure of the Aeration Lagoons as required by <u>Comment 8</u> of the March 24, 2010 NOTICE OF DISAPPROVAL letter, and submittal extension that was granted by the Bureau on April 22, 2010.

On April 26, 2010 Western responded under a separate to comments 1-7 of the March 24, 2010 NOTICE OF DISAPPROVAL as required.

If you have any questions or would like to discuss the Aeration lagoon Closure Plan, please contact me at (505) 632-4171.

Sincerely,

James R. Schmaltz Environmental Manager Western Refining Southwest, Inc. Bloomfield Refinery

cc: Hope Monzeglio – NMED HWB (Carl Chavez – NMOCD (w/attachment)) Dave Cobrain – NMED HWB Laurie King – EPA Region 6 (w/attachment) Todd Doyle – Bloomfield Refinery Allen Hains – Western Refining El Paso



404 Camp Craft Rd., Austin, Texas 78746, USA T +1 512 347 7588 F +1 512 347 8243 W www.rpsgroup.com

Final Closure Plan North and South Aeration Lagoons Bloomfield Refinery

Regulated Unit EPA ID# NMD089416416 HWB-GRCB-09-007

Western Refining Southwest, Inc. Bloomfield Refinery Bloomfield, New Mexico

May 2010

James R. Schmaltz

Environmental Manager

Goud

Scott T. Crouch, P.G. Senior Consultant RPS

United Kingdom | Australia | USA | Canada | Russia | Malaysia

Table of Contents

List of Sections

Section 1	Introduction		
Section 2 2.1 2.2 2.3	 N 2 Wastewater Treatment Unit Description and Operation 2.1 Environmental Regulatory Activities. 2.2 ABT Unit Operations. 2.3 Contingency Plan N 3 Completed Partial Closure Activities. 3.1 Sludge Characterization, Removal, and Disposal 3.2 Liner Inspection and Repair. 3.3 Flushing of Leachate Collection System 		
Section 3 3.1 3.2 3.3			
Section 4 4.1 4.2	ABT Unit Closure ABT Units Closure Procedures 4.1.1 Sludge/Sediment Removal 4.1.2 RCRA Liner Removal 4.1.3 Non-RCRA Liner/Leachate Collection System Removal Soil Investigation 4.2.1 Soil Sample Field Screening and Logging 4.2.2 Drilling Activities 4.2.3 Sample Handling 4.2.4 Collection and Management of Investigation Derived Waste 4.2.5 Field Equipment Calibration 4.2.6 Documentation of Field Activities 4.2.7 Chemical Analyses 4.2.8 Data Quality Objectives	13 14 14 14 15 16 17 18 18 19 20 20 20 20 21	
4.3 4.4	Soil Removal Action 4.3.1 Soil Excavation 4.3.2 Confirmation Sampling Closure Certification		
Section 5	Construction Details		
Section 6	References		

List of Figures

Figure 1	Site Location Map	.3
Figure 2	Aeration Lagoons Location Map	9
Figure 3	Sample Location Map	24

List of Appendices

Appendix A Management of Investigation Derived Waste Appendix B Closure Cost Estimate

Section 1 Introduction

The Bloomfield Refinery is located immediately south of Bloomfield, New Mexico in San Juan County. The physical location address is #50 Road 4990, Bloomfield, New Mexico 87413. The Bloomfield Refinery is located on approximately 263 acres. The site is located on a bluff approximately 100 feet above the south side of the San Juan River, a perennial river that flows to the west (Figure 1).

Bordering the facility is a combination of federal and private properties. Public property managed by the Bureau of Land Management lies to the south. The majority of undeveloped land in the vicinity of the facility is used extensively for oil and gas production and, in some instances, grazing. The town of Bloomfield is located to the north of the refinery, across the San Juan River. U.S. Highway 550 is located approximately one-half mile west of the facility. The topography of the site is generally flat with low-lying areas to the east of the process area.

The Bloomfield Refinery is a crude oil refinery currently owned by Western Refining Southwest, Inc., which is a wholly owned subsidiary of Western Refining Company, and it is operated by Western Refining Southwest, Inc. – Bloomfield Refinery. The Bloomfield Refinery generally processed crude oil from the Four Corners area transported to the facility by pipeline or tanker truck and crude from West Texas transported by pipeline.

The Bloomfield Refinery has an approximate refining capacity of 18,000 barrels per day; however, the refinery suspended petroleum refining operations in November 2009. Various process units operated at the facility, included crude distillation, reforming, fluidized catalytic cracking, sulfur recovery, merox treater, catalytic polymerization and diesel hydrotreating. Products produced at the refinery included gasoline, diesel fuels, jet fuels, kerosene, propane, butane, naphtha, residual fuel, fuel oils and LPG.

This Closure Plan addresses the final closure of the North and South Aeration Lagoons. Historical monitoring data of the effluent from the API Separator, which discharges into the South Aeration Lagoon, indicated that concentrations of benzene above the toxicity characteristic (TC) regulatory threshold of 0.5 milligrams per liter (mg/l) entered the aeration lagoons. Modifications to the wastewater treatment system required that the lagoons be cleaned out to remove all hazardous waste, hazardous constituents, decomposition products,

and leachate. These "partial closure" activities were completed in October 2008 through February 2009 pursuant to the North and South Aeration Lagoons Closure Plan dated May 2008 [approved by the New Mexico Environment Department (NMED) on August 7, 2008]. Final closure of the aeration lagoons will be conducted in accordance with an Enforceable Document (July 27, 2007 NMED Order) and this Final Closure Plan.





Section 2 Wastewater Treatment Unit Description and Operation

2.1 Environmental Regulatory Activities

All oil refineries produce process wastewater, which today must be managed in accordance with a variety of environmental requirements intended to assure adequate and appropriate protection of public health and the environment. Three federal regulatory programs [the Clean Water Act (CWA), the Resource Conservation and Recovery Act (RCRA), and the Safe Drinking Water Act (SDWA)] have major significance for Bloomfield Refinery process wastewater. Two of these federal programs at Bloomfield are directly administered by the State of New Mexico, as it has primacy over the RCRA and SDWA Underground Injection Control (UIC) programs. In addition, there are additional State regulatory programs with varying applicability, including those administered by New Mexico Oil Conservation Division (OCD).

Initially, beginning in 1972 under the CWA regulatory program, EPA promulgated petroleum refinery wastewater management requirements pursuant to the National Pollutant Discharge Elimination System (NPDES) permit program. The principal federal regulations implementing this CWA program as it applies to petroleum refineries are found at 40 C.F.R. Parts 122 and 419. The Bloomfield Refinery, like other oil refineries impacted by 40 C.F.R. Part 419, had implemented a series of process wastewater treatment operations, including primary treatment of wastewaters with an oil/water separator followed by aggressive biological treatment in accordance with 40 CFR §261.31(b)(2). The two ponds where such biological treatment oily Water Pond.

A second major regulatory program, the RCRA regulations, affecting hazardous waste was promulgated by EPA on November 19, 1980. Initially, these applied only to certain sludges created by petroleum refinery wastewater management, such as API oil/water separator sludge that was listed as K051 hazardous waste. In November 1980, the Bloomfield Refinery operator applied for a Part A permit as a generator and TSD facility as a protective filing for its so-called oily water ponds. It was later determined that they were not disposing of listed hazardous waste on site since D018 wastewater was not part of the 1980 EP toxicity test (it only became regulated after the 1990 TCLP toxicity test was adopted). In 1982 they petitioned for RCRA



reclassification under a generator only status.¹ In 1982/1983, the liquids and sludge were removed from the oily water ponds and disposed of offsite. Impacted soils were also excavated and the ponds were lined. This activity included the placement of a composite liner consisting of a 33% bentonite/soil bottom liner, a French drain system, and a 100 mill high density polyethylene (HDPE) upper liner.

In 1990, a significant revision to these regulations classified most petroleum refinery process wastewater as D018 benzene characteristic hazardous waste, leading the Bloomfield Refinery to submit a Part B RCRA permit application² in the mid-1990s and to operate its biological treatment impoundments pursuant to RCRA interim status as a regulated unit. To comply with RCRA interim status, the Bloomfield Refinery upgraded and retrofitted the impoundments with an additional set of RCRA double liners and leak detection/leachate collection system over the liner system that the Bloomfield Refinery had installed in 1982/1983.

In 1992, the listing of F037/F038 sludges by EPA as hazardous wastes effectively mandated a certain level of biological treatment and retention time in the biological treatment impoundments at the Bloomfield Refinery.³ Thereafter, the aeration-enhanced impoundments were called the North Aeration Lagoon (NAL) or the South Aeration Lagoon (NAL) [also referred to herein as the North Aggressive Biological Treatment (ABT) Units (two impoundments known as NABT-E and NABT-W) and the South ABT Unit] (Figure 2). The compliance strategy employed aggressive biological treatment followed by disposition through evaporation ponds and a Class I underground injection well permitted consistent with the Safe Drinking Water Act UIC program requirements.⁴

⁴EPA promulgated regulatory requirements to assure that wastewater managed by UIC disposition not pose a risk to public health and the environment (40 C.F.R. Parts 144-146), but those did not apply at the Bloomfield Refinery until 1994 when Bloomfield installed a Class I UIC well for wastewater management.



¹ On November 26, 1985, the Bloomfield Refinery agreed to take an on-site landfill [where some of the materials from the 1982 impoundment cleanout had been placed] through RCRA closure. During 1989, these materials were removed and eventually determined by EPA delisting to be non-hazardous for offsite disposal. See, <u>Hazardous Waste Delisting Petition, Petroleum Contaminated Soil</u>, dated April 15, 1991 (ERM-Rocky Mountain, Inc.)

²This Part B application submitted in the mid-1990s included a RCRA closure plan for the biological treatment impoundments, as discussed later in this document.

³Integral to the operation of the Bloomfield Refinery, as with any oil refinery in the United States, is the operation of an aggressive biological treatment (ABT) unit system for wastewater management, mandated by EPA regulations regarding the listing of certain petroleum refinery wastes (F037/F038) that became effective in May, 1991. EPA regulations, as adopted by NMED, effectively require each petroleum refinery to implement an ABT system to biological treat organics with regulatorily-specified ABT technology to remove organics and eliminate F037/F038 formation. The Bloomfield Refinery has had such advanced organic aeration in place as required since that time, and these EPA-required treatment systems operate as multi-lined ABT wastewater treatment units at Bloomfield, backed up with a double set of leak detection/leachate collection systems, over and above what has been technologically required under EPA regulations.

As a result of an EPA Consent Agreement and Final Order (CAFO) dated May 18, 2006, additional upgrades were made to wastewater treatment operations at the Bloomfield Refinery in the fall of 2007. The upgrades included construction and operation of a benzene stripper/surge tank system that decharacterizes all potentially D018 characteristically hazardous process wastewater prior to further biological treatment in the ABT impoundments. The tank system includes a 10,000 barrel tank to provide surge capacity. As a result, all process wastewater streams, including any contaminated runoff, is decharacterized prior to discharge into the ABT units for aggressive biological treatment.

2.2 ABT Unit Operations

The refinery process wastewater that was generated [approx. 80 gallons per minute (gpm)] at the Bloomfield Refinery prior to suspension of the petroleum refining operations was managed first by treatment in an API oil/water separator, then benzene air strippers to remove the volatile components and the final treatment (biological) in the three ABT impoundments. The impoundments were designed and equipped with aerators sized to prevent F037/F038 waste generation through high rate aeration (i.e., aggressive biological treatment) in accordance with 40 CFR §261.31(b)(2). With the installation of the benzene stripper equipment in October 2007, the wastewater is "decharacterized" below the benzene TC levels prior to discharge into the first (South) ABT unit. The liner system for the ABT units, from top to bottom, includes:

- A 100-mil HDPE top liner;
- A geonet for collecting leaks that drain to a sump equipped with a 6" observation pipe;
- A 60-mil HDPE secondary liner;
- A composite geotextile/geonet with a 4" observation pipe;
- A cement amended sand that was compacted into a 1.5% slope;
- A 100-mil HDPE liner;
- A French drain system, which directs any collected fluids to a central sump; and
- A 6" layer of soil with 33% bentonite mixed into it.

The wastewater discharges from the API separator, passes through the benzene air stripper and into the first (South) ABT unit, which averages 4.4 feet in depth and has a surface area of about 6,652 square feet. The total volume is approximately 216,000 gallons. At 80 gpm, the holding time in the pond was 1.9 days. The South ABT unit is equipped with two, 5-horsepower aerators sized to prevent F037/F038 waste generation through high rate aeration.

Wastewater from the first (South) ABT unit is routed to the second (North) ABT unit through an overflow pipe. The second ABT unit is comprised of two impoundments that are operated together, and are generally referred to together as the North ABT unit. The first of the two

impoundments (which can be referred to as North ABT-W as it is the westernmost of the two portions of the North ABT unit) is separated from the second (the second can be referred to as North ABT-E) by a concrete divider. An overflow pipe from the North ABT-W connects to the North ABT-E. The outflow from North ABT-E goes to a sump, where the non-hazardous wastewater can be pumped for final disposition, either in evaporation ponds or into an SDWA Class I permitted non-hazardous UIC well.

The North ABT-W averages 5.5 feet in depth with a surface area of 10,000 square feet. The total volume is approximately 411,500 gallons. The North ABT-W unit is equipped with two (each) 2-horsepower aerators and wastewater retention time (at 80 gpm) was 3.6 days at 80 gpm.

The North ABT-E (the second of the two in the North area) averages 5.7 feet in depth, with a surface area of 8,440 square feet and a volume of approximately 360,000 gallons. The North ABT-E is equipped with two 2-horsepower aerators and wastewater retention time (at 80 gpm) was 3.1 days

The North and South ABT units have been operated with a minimum freeboard of two feet under normal operating conditions. At the lowest points during operation, the South ABT, North ABT-W and North ABT-E have freeboards of 2.97, 2.54 and 3.08 feet respectively. Influent flow into the South ABT unit is limited by the size of the overflow pipe coming from the API separator/wastewater treatment unit system. Operating personnel monitor pond water levels on a daily basis. The only non-controlled inflow is direct rainfall onto the North and South unit areas.

To manage precipitation, outflow from the ABT unit system is routed to a sump, which has an automatic level control pump. Excess water from process areas generated during a 100-year storm (2.6") is easily handled by this system. The impoundments have 698,000 gallons of additional capacity to the top of the freeboard and the pump can remove 720,000 gallons of water daily. This capacity management total greatly exceeds the 406,000 gallons of water that would be drained from 250,000 square feet of process area. The pump is backed up by two portable diesel backup pumps, which can function in the event of a power outage.

Since termination of the active refining operations, the flow to the ABT units has been reduced to approximately 40 to 60 gpm but this is the only change in operations at the units. The flows



now include primarily ground water recovered from the remediation systems, stormwater and wastewater incidental to the continued terminal operations.

2.3 **Contingency Plan**

In the event of a major failure, the first contingency response is to direct the wastewaters that have not been through the benzene stripping treatment process into the 10,000 barrel surge tank. At a rate of wastewater flow of 60 gpm, that would permit 116 hours of flow to be managed without discharge to the ABT units in the event of a benzene stripper failure. During those 116 hours for repair work, the benzene strippers in most cases could be fixed and returned to operation. In the event the surge tank capacity may be exceeded, it may be possible to make additional surge tank capacity available, depending on other tank usage at the Bloomfield Refinery. Such evaluation would occur if there was a significant likelihood the strippers could not be restored to working order within the 116 hour time frame available for repairs.

Once the benzene strippers are made operational again, wastewaters collected in the surge tank will be appropriately metered back through the wastewater treatment system by being introduced upstream of the API separator consistent with capacity available (in excess of the 40 gpm flow being handled). After the wastewater in the surge tank has been removed, the tank will be inspected to determine if any potentially F037 or F038 listed waste has accumulated. However, it should be noted that since suspension of refining operations, F037 or F038 listed waste should not be present. If residual sludge is present, it will be physically removed from the tank via the manway, and characterized and sent off-site for disposal in accordance with all applicable Hazardous Waste regulations.





Section 3 Completed Partial Closure Activities

NMED approved the previous Closure Plan for the North and South Aeration Lagoons (dated May 2008) on August 7, 2008. The previous Closure Plan required the removal of all materials (water and sludge) from within the North and South Aeration Lagoons and decontamination and repairs, as necessary, of the RCRA liner/leachate collection system. The May 2008 Closure Plan was implemented between October 2008 and May 2009. The previously completed closure activities are discussed in detail in the North and South Aeration Lagoons Closure Certification Report dated September 2009 (supplemented via correspondence dated April 20, 2010) and are summarized below.

3.1 Sludge Characterization, Removal, and Disposal

In accordance with the Closure Plan, sixteen sludge samples were collected in a grid pattern from the South ABT unit on October 7, 2008. Five of the first seven sludge samples analyzed exhibited hazardous waste characteristics for benzene, therefore no further testing was conducted and the entire sludge volume (851,930 pounds) in the South ABT unit was removed from the unit via vacuum transport truck and directly shipped offsite as oil-bearing hazardous secondary materials to be recycled (fuels blending).

On October 9, 2008, twenty sludge samples were collected in a grid pattern from the Northwest ABT unit. Analytical results indicated that the sludge did not exhibit any hazardous characteristics. The sludge in the Northwest ABT unit was removed via vacuum truck, mixed with fly ash, transported, and disposed of at the San Juan County Landfill in Aztec, New Mexico. Approximately 2,476,880 pounds of material, including sludge and fly ash, was transported to the landfill.

On October 21, 2008, seven sludge samples were collected in a grid pattern from the Northeast ABT unit. Analytical results indicated that the sludge did not exhibit any hazardous characteristics. The sludge in the Northeast ABT unit was removed via vacuum truck, mixed with fly ash, transported and disposed of at the San Juan County Landfill in Aztec, New Mexico. Approximately 1,998,780 pounds of sludge and fly ash were transported to the landfill.



3.2 Liner Inspection and Repair

After removal of the sludge, the entire top RCRA liner of each lagoon was power washed with water. The wash water was collected via vacuum truck and off-loaded into the API Separator. The South ABT unit was physically inspected in November 2008 by an independent engineer licensed in the State of New Mexico during closure activities. A crack in the plastic weld was discovered at the crossover piping between South ABT unit and the Northwest ABT unit. During the inspection process, personnel also discovered damage to the boot on the lower RCRA liner and pitting and corrosion on the piping that discharges from the South ABT unit into the Northwest ABT unit. Both sections of discharge piping from the South ABT unit into the Northwest and Northeast ABT units were replaced. After the piping was replaced, the liner was repaired inside the South ABT unit and inspected again. There was no accumulation of fluids in the underlying collection system beneath the lower 60-ml RCRA liner to indicate damage to the secondary liner. The South ABT unit was put back in service by November 18, 2008.

The Northwest ABT unit's top RCRA liner was power washed with water after sludge removal. The wash water was collected via vacuum truck and off-loaded into the API Separator. The liner surrounding the new inlet pipe from the South ABT Unit was repaired. Inspection of the upper liner identified small scrapes and gouges that did not penetrate the liner. A reinforcing plastic weld bead was applied to those areas before water was put back in the Northwest ABT unit on December 29, 2009.

After sludge removal, the top RCRA liner of the Northeast ABT unit was power washed with water. The wash water was collected via vacuum truck and off-loaded into the API Separator. Clean out activities were completed January 20, 2009. Inspection of the upper RCRA line revealed a puncture on the north wall of the liner and a cut on the top east side of the upper liner; neither of these penetrations went through the lower liner. The discharge pipe from the Northeast ABT unit to the suction of P-616 (Transfer Pump from the Aeration Lagoon to the Evaporation Ponds) was found to be pitted and corroded and was replaced. The liner was repaired around the new discharge piping as well as around the new transfer piping from the South ABT unit. The repair of the aforementioned puncture and cut in the upper liner was completed February 4, 2009 and the unit was put back into service.

Liner repairs surrounding the piping replacement consisted of cutting out and removing all three liners and the geonet. The liners and geonet were replaced with new material and new boots were created to tie into the piping.





3.3 Flushing of Leachate Collection System

Pursuant to the previous Closure Plan, after repairs to the upper liner of the South ABT unit were completed, the geonet between the upper 100-ml liner and the lower 60-ml liner was flushed with clean water. The flush water was sampled using a bailer in the 6" observation pipe. Analytical results indicated that the flush water did not exhibit any hazardous characteristics. The flush water was removed from underneath the upper liner via vacuum truck through the 6" observation pipe and off-loaded at the API Separator.

An inspection of the upper liner in the Northwest ABT unit did not reveal any penetrations of the liner. As there were no indications of impacts to the upper liner, the leachate collection system was not flushed with water.

As discussed above, inspection of the Northeast ABT unit revealed a puncture on the north wall of the upper liner and a cut on the top east side of the upper liner. In addition, spongy conditions were observed under the upper liner in the northeast section. This observation prompted sampling and analysis of the water between the upper 100-ml liner and the lower 60-ml liner. The water was sampled through a new incision placed in the liner. Analytical results indicate that the water did not exhibit any hazardous characteristics, thus the leachate collection system was not flushed. The water was removed from underneath the upper liner via vacuum truck through two new incisions placed in the liner and off-loaded at the API Separator. There was no accumulation of fluids in the underlying collection system beneath the 60-ml liner to indicate damage to the RCRA liner. The incisions were repaired along with the other repairs of the liner.







Section 4 ABT Unit Final Closure

This Final Closure Plan sets forth the activities that will be conducted to achieve final closure of the ABT units at the Bloomfield Refinery. These activities will be conducted in compliance with the requirements of the NMED Order dated July 27, 2007 (also referred to as the Enforceable Document) in addition to the applicable closure standards in 40 CFR Part 265.

The applicable closure standard for the North and South ABT Units is provided in 40 CFR §265.111 (Closure Performance Standard), which requires that the owner or operator must close the facility in a manner that:

(a) Minimizes the need for further maintenance, and

(b) Controls, minimizes or eliminates, to the extent necessary to protect human health and the environment, post-closure escape of hazardous waste, hazardous constituents, leachate, contaminated run-off, or hazardous waste decomposition products to the ground or surface waters or to the atmosphere.

The objective of this scope of services is to close the units as Corrective Action Complete Without Controls (CACWOC). The final closure activities are designed to meet the surface impoundment closure requirements of 40 CFR §265.228 (a)(1) – Remove or decontaminate all waste residues, contaminated containment system components (liners, etc.), contaminated subsoils, and structures and equipment contaminated with waste and leachate, and manage them as hazardous waste unless §261.3(d) of the chapter applies.

It is unlikely that the units will contain hazardous waste because of recent upgrades to the wastewater treatment operations discussed above in Section 2.1 and the partial closure activities that removed all hazardous waste from the ABT units as described above in Section 3. Regardless, all waste materials managed during closure will be tested to determine if they are characteristically hazardous in accordance with 40 CFR 261, Subpart C – Characteristics of Hazardous Waste.

4.1 ABT Units Closure Procedures

The steps described below will be implemented to achieve "clean closure" pursuant to 40 CFR §265.228(a)(1). The closure will begin by removing all liquids from the units using the current authorized wastewater treatment and discharge system. It is likely that only a small volume of

sludge/sediment will be present in the units at final closure, based on the fact that the units were completely cleaned out in late 2008 through early 2009 in order to remove all hazardous wastes, hazardous constituents, decomposition products, and leachate. Final closure for the three impoundments will be completed by implementing the steps discussed below.

4.1.1 Sludge/Sediment Removal

The sludge/sediment that remains in the ABT units above the top liner after removal of the free liquids will be allowed to dry for up to four weeks. If necessary, mechanical equipment such as a back hoe, track hoe, long-reach excavators, or similar may be used to facilitate physical drying of the sludge/sediment, moving the sludge/sediment to expose saturated portions to ambient air. At the conclusion of these activities, the material will be sampled for hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste, including reactivity, corrosivity, ignitability, TCLP RCRA metals, TCLP semi-volatiles (SVOCs), and total volatiles (VOCs). If the material is non-hazardous, then it may be disposed at a landfill permitted by the NMED to accept Special Waste (e.g., the San Juan County Regional Landfill).

Samples of the sludge/sediment will be collected for waste characterization at a minimum of one sample per each 20 cubic yards. If the sludge/sediment does not exhibit any hazardous characteristics, it will be removed from the ABT units by a vacuum truck or other mechanical means (e.g., long-reach track hoe) depending upon the consistency of the material for appropriate disposal. Portland cement or fly ash may be added to improve physical strength and reduce moisture content prior to excavation out of the units. If sludge/sediment exhibits hazardous characteristics, then it will be placed into appropriate RCRA containers for disposal offsite as hazardous waste.

4.1.2 RCRA Liner Removal

The RCRA liners, which include an upper 100 mil HDPE liner, a geonet, and a lower 60 mil HDPE liner, will be removed and disposed at a landfill permitted by the NMED to accept Special Waste (e.g., the San Juan County Regional Landfill). Any liquids that are present in the leachate collection system will be containerized. The liquid will be sampled and analyzed for hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste, including reactivity, corrosivity, ignitability, total RCRA metals, SVOCs, and VOCs. If the liquid exhibits hazardous characteristics, then it will be disposed offsite as



hazardous waste. If the liquid is non-hazardous, then it will be disposed through the on-site permitted discharge system. The liners and geonet will be cut into manageable sized pieces and then rolled/folded to facilitate loading into trucks/roll-off boxes for transport for off-site disposal.

4.1.3 Non-RCRA Liner/Leachate Collection System Removal

There are a series of liners/leachate collection systems below the RCRA liner system that will be removed for off-site disposal. The uppermost layer beneath the RCRA lower 60 mil liner is a composite geotextile/geonet, which will be cut into manageable pieces and placed into roll-off boxes for off-site disposal. Beneath this composite geotextile/geonet layer is a 6" layer of cement amended sand. The sand layer will be excavated and stockpiled on-site pending waste characterization sampling. Beneath the sand layer is a 100 mil HDPE liner, which will be removed and handled with the previously removed composite geotextile/geonet layer. The lowermost layer consists of approximately 6 inches of bentonite amended soil with a French drain system.

Prior to removal of the French drain system, the soil dike separating the North and South ABT Units will be removed and stockpiled on-site. The French drain system will be checked for the presence of liquids prior to removal. Any liquids that are present will be containerized. The liquid will be sampled and analyzed for hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste. If the liquid exhibits hazardous characteristics, then it will be disposed offsite as hazardous waste. If the liquid is nonhazardous, then it will be disposed through the facilities on-site permitted discharge. Soil will be physically removed from the drain system piping and if the liquid present in the drain system is characteristically hazardous, then the piping will be triple rinsed with potable water prior to offsite disposal. The wash water will be containerized and analyzed for classification prior to disposal.

The cemented amended sand will be sampled for waste characterization at a minimum of one sample per 20 cubic yards. The samples will be analyzed for hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste, including reactivity, corrosivity, ignitability, TCLP RCRA metals, TCLP SVOCs, and total VOCs. If the material does not exhibit any hazardous characteristics, then it will be transported for off-site disposal to a NMED permitted non-hazardous landfill. If the material exhibits hazardous



characteristics, then it will be placed into appropriate RCRA containers for disposal offsite as hazardous waste.

The soil from the dike that separates the North and South ABT Units will be sampled for waste characterization at a minimum of one sample per 20 cubic yards. The samples will be analyzed for hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste, including reactivity, corrosivity, ignitability, TCLP RCRA metals, and TCLP SVOCs and will also be analyzed for totals concentrations of the constituents set forth below in Section 4.2.7 to evaluate potential reuse. If all concentrations of constituents are below the applicable NMED residential soil screening levels, then the soil may be reused for backfill. If the soil is not suitable to use on-site as backfill and does not exhibit any hazardous characteristics, then it will be transported for off-site disposal to a NMED permitted non-hazardous landfill. If the soil exhibits hazardous characteristics, then it will be placed into appropriate RCRA containers for disposal off-site as hazardous waste.

4.2 Soil Investigation

The purpose of the soil investigation is to determine if a release of contaminants from the ABT units has occurred and if so, to evaluate the nature and extent of the release. Guidance for Choosing a Sampling Design for Environmental Data Collection (EPA, 2000) was utilized to select the appropriate sampling strategy.

Investigation sample locations will be identified by gridding the bottom of the entire area (i.e., footprint of all three ABT Units) into 50 feet by 50 feet grids (Figure 3). Soil borings will be completed at the approximate center of each grid to a minimum depth of ten feet. If soils appear impacted (e.g., petroleum odor, staining, or elevated organic vapor readings) at ten feet, then the soil boring(s) will be drilled deeper until the vertical extent of the impact is reached or ground water is encountered, whichever occurs first.

Discrete soil samples will be collected for laboratory analyses from the soil borings at the following intervals:

- 18-24";
- The sample from each soil boring with the greatest apparent degree of contamination, based on field observations and field screening; and
- Any additional intervals as determined based on field screening results.





Shallow (0-6" and 18-24") samples will be collected from around the perimeter of the ABT Units using hand augers or similar manual tools on a 50-foot spacing. The samples will be collected from approximately half way up the sidewall of the ABT Units (Figure 3).

4.2.1 Soil Sample Field Screening and Logging

Samples obtained from the borings will be screened in the field on two foot intervals for evidence of contaminants. Field screening results will be recorded on the exploratory boring logs. Field screening results will be used to aid in the selection of soil samples for laboratory analysis. The primary screening methods include: (1) visual examination, (2) olfactory examination, and (3) headspace vapor screening for volatile organic compounds.

Visual screening includes examination of soil samples for evidence of staining caused by petroleum-related compounds. Headspace vapor screening targets volatile organic compounds and involves placing a soil sample in a plastic sample bag or a foil sealed container allowing space for ambient air. The container will be sealed and then shaken gently to expose the soil to the air trapped in the container. The sealed container will be allowed to stand for a minimum of 5 minutes while vapors equilibrate. Vapors present within the sample bag's headspace will then be measured by inserting the probe of the VOC screening instrument in a small opening in the bag or through the foil. The maximum value and the ambient air temperature will be recorded on the field boring or test pit log for each sample.

The monitoring instruments will be calibrated each day to the manufacturer's standard for instrument operation. A photo-ionization detector (PID) equipped with a 10.6 or higher electron volt (eV) lamp or a combustible gas indicator will be used for VOC field screening. All conditions capable of influencing the results of field screening will be recorded on the field logs since field screening results may vary with instrument type, the media screened, weather conditions, moisture content, soil type, and type of contaminant,

The physical characteristics of the samples (such as mineralogy, ASTM soil classification, moisture content, texture, color, presence of stains or odors, and/or field screening results), depth where each sample was obtained, method of sample collection, and other observations will be recorded in the field log by a qualified geologist or engineer. Detailed logs of each boring will be completed in the field by a qualified engineer or geologist. Additional information, such as the presence of water-bearing zones and any unusual or noticeable conditions encountered during drilling, will be recorded on the logs.







Quality Assurance/Quality Control (QA/QC) samples will be collected to monitor the validity of the soil sample collection procedures as follows:

- Field duplicates will be collected at a rate of 10 percent; and
- Equipment blanks will be collected at a frequency of one per day.

4.2.2 Drilling Activities

Soil borings will be drilled using either a hand auger, cone penetrometer (CPT), hollow-stem auger or if necessary, air rotary methods including ODEX. The drilling equipment will be properly decontaminated before drilling each boring.

The NMED will be notified as early as practicable if conditions arise or are encountered that do not allow the advancement of borings to the specified depths or at planned sampling locations. Appropriate actions (e.g., installation of protective surface casing or relocation of borings to a less threatening location) will be taken to minimize any negative impacts from investigative borings. Soil samples will be collected continuously and logged by a qualified geologist or engineer.

Both sample information and visual observations of the cuttings and core samples will be recorded on the boring log. Known site features and/or site survey grid markers will be used as references to locate each boring. The boring locations will be measured to the nearest foot, and locations will be recorded on a scaled site map upon completion of each boring.

4.2.3 Sample Handling

At a minimum, the following procedures will be used at all times when collecting samples during investigation, corrective action, and monitoring activities:

- 1. Neoprene, nitrile, or other protective gloves will be worn when collecting samples. New disposable gloves will be used to collect each sample;
- 2. All samples collected of each medium for chemical analysis will be transferred into clean sample containers supplied by the project analytical laboratory with the exception of soil, rock, and sediment samples obtained in Encore® samplers. Sample container volumes and preservation methods will be in accordance with the most recent standard EPA and industry accepted practices for use by accredited analytical laboratories. Sufficient sample volume will be obtained for the laboratory to complete the method-specific QC analyses on a laboratory-batch basis; and
- 3. Sample labels and documentation will be completed for each sample following procedures discussed below. Immediately after the samples are collected,





they will be stored in a cooler with ice or other appropriate storage method until they are delivered to the analytical laboratory. Standard chain-of-custody procedures, as described below, will be followed for all samples collected. All samples will be submitted to the laboratory soon enough to allow the laboratory to conduct the analyses within the method holding times. At a minimum, all samples will be submitted to the laboratory within 48 hours after their collection.

Chain-of-custody and shipment procedures will include the following:

- 1. Chain-of-custody forms will be completed at the end of each sampling day, prior to the transfer of samples off site.
- 2. Individual sample containers will be packed to prevent breakage and transported in a sealed cooler with ice or other suitable coolant or other EPA or industry-wide accepted method. The drainage hole at the bottom of the cooler will be sealed and secured in case of sample container leakage. Temperature blanks will be included with each shipping container.
- 3. Each cooler or other container will be delivered directly to the analytical laboratory.
- 4. Glass bottles will be separated in the shipping container by cushioning material to prevent breakage.
- 5. Plastic containers will be protected from possible puncture during shipping using cushioning material.
- 6. The chain-of-custody form and sample request form will be shipped inside the sealed storage container to be delivered to the laboratory.
- 7. Chain-of-custody seals will be used to seal the sample-shipping container in conformance with EPA protocol.
- 8. Signed and dated chain-of-custody seals will be applied to each cooler prior to transport of samples from the site.
- 9. Upon receipt of the samples at the laboratory, the custody seals will be broken, the chain-of-custody form will be signed as received by the laboratory, and the conditions of the samples will be recorded on the form. The original chain-of-custody form will remain with the laboratory and copies will be returned to the relinquishing party.
- 10. Copies of all chain-of-custody forms generated as part of sampling activities will be maintained on-site.

4.2.4 Collection and Management of Investigation Derived Waste

Drill cuttings, excess sample material and decontamination fluids, and all other investigation derived waste (IDW) associated with soil borings will be contained and characterized using methods based on the boring location, boring depth, drilling method, and type of contaminants





suspected or encountered. All decontamination water will be characterized prior to disposal unless it is disposed in the refinery wastewater treatment system upstream of the API Separator. An IDW management plan is included as Appendix A.

4.2.5 Field Equipment Calibration

Field equipment requiring calibration will be calibrated to known standards, in accordance with the manufacturers' recommended schedules and procedures. At a minimum, calibration checks will be conducted daily, or at other intervals approved by the Department, and the instruments will be recalibrated, if necessary. Calibration measurements will be recorded in the daily field logs. If field equipment becomes inoperable, its use will be discontinued until the necessary repairs are made. In the interim, a properly calibrated replacement instrument will be used.

4.2.6 Documentation of Field Activities

Daily field activities, including observations and field procedures, will be recorded in a field log book. Copies of the completed forms will be maintained in a bound and sequentially numbered field file for reference during field activities. Indelible ink will be used to record all field activities. Photographic documentation of field activities will be performed, as appropriate. The daily record of field activities will include the following:

Site or unit designation;

- 1. Date;
- 2. Time of arrival and departure;
- 3. Field investigation team members including subcontractors and visitors;
- 4. Weather conditions;
- 5. Daily activities and times conducted;
- 6. Observations;
- 7. Record of samples collected with sample designations and locations specified;
- 8. Photographic log, as appropriate;
- 9. Field monitoring data, including health and safety monitoring;
- 10. Equipment used and calibration records, if appropriate;
- 11. List of additional data sheets and maps completed;
- 12. An inventory of the waste generated and the method of storage or disposal; and
- 13. Signature of personnel completing the field record.

4.2.7 Chemical Analyses

All samples collected for laboratory analysis will be submitted to an accredited laboratory. The laboratory will use the most recent standard EPA and industry-accepted analytical methods for target analytes as the testing methods for each medium sampled. Chemical analyses will be

performed in accordance with the most recent EPA standard analytical methodologies and extraction methods.

Soil samples will be analyzed by the following methods:

- SW-846 Method 8260 volatile organic compounds;
- SW-846 Method 8270 semi-volatile organic compounds; and
- SW-846 Method 8015B gasoline range (C5-C10), diesel range (>C10-C28), and motor oil range (>C28-C36) organics.
- Soil samples will also be analyzed for the following Skinner List metals using the indicated analytical methods.

Analyte	Analytical Method
Antimony	SW-846 method 6010/6020
Arsenic	SW-846 method 6010/6020
Barium	SW-846 method 6010/6020
Beryllium	SW-846 method 6010/6020
Cadmium	SW-846 method 6010/6020
Chromium	SW-846 method 6010/6020
Cobalt	SW-846 method 6010/6020
Cyanide	SW-846 method 335.4/335.2 mod
Lead	SW-846 method 6010/6020
Mercury	SW-846 method 7470/7471
Nickel	SW-846 method 6010/6020
Selenium	SW-846 method 6010/6020
Silver	SW-846 method 6010/6020
Vanadium	SW-846 method 6010/6020
Zinc	SW-846 method 6010/6020

Inorganic Analytical Methods

4.2.8 Data Quality Objectives

The Data Quality Objectives (DQOs) were developed to ensure that newly collected data are of sufficient quality and quantity to address the projects goals, including Quality Assurance/Quality Control (QA/QC) issues (EPA, 2006). The project goals are established to determine and evaluate the presence, nature, and extent of releases of contaminants from the ABT Units. The type of data required to meet the project goals includes chemical analyses of soil to determine if there has been a release of contaminants. Method detection limits should be 20% or less of the applicable background levels, cleanup standards and screening levels.







Additional DQOs include precision, accuracy, representativeness, completeness, and comparability. Precision is a measurement of the reproducibility of measurements under a given set of circumstances and is commonly stated in terms of standard deviation or coefficient of variation (EPA, 1987). Precision is also specific to sampling activities and analytical performance. Sampling precision will be evaluated through the analyses of duplicate field samples and laboratory replicates will be utilized to assess laboratory precision.

Accuracy is a measurement in the bias of a measurement system and may include many sources of potential error, including the sampling process, field contamination, preservation, handling, sample matrix, sample preparation, and analysis techniques (EPA, 1987). An evaluation of the accuracy will be performed by reviewing the results of matrix spikes and laboratory QC samples.

Representativeness is an expression of the degree to which the data accurately and precisely represent the true environmental conditions. Sample locations and the number of samples have been selected to ensure the data is representative of actual environmental conditions.

Completeness is defined as the percentage of measurements taken that are actually valid measurements, considering field QA and laboratory QC problems. EPA Contract Laboratory Program (CLP) data has been found to be 80-85% complete on a nationwide basis and this has been extrapolated to indicate that Level III, IV, and V analytical techniques will generate data that are approximately 80% complete (EPA, 1987). As an overall project goal, the completeness goal is 85%; however, some samples may be critical based on location or field screening results and thus a sample –by-sample evaluation will be performed to determine if the completeness goals have been obtained.

Comparability is a qualitative parameter, which expresses the confidence with which one data set can be compared to another. Industry standard sample collection techniques and routine EPA analytical methods will be utilized to help ensure data are comparable to historical and future data. Analytical results will be reported in appropriate units for comparison to historical data and cleanup levels.

4.3 Soil Removal Action

Removal actions will be conducted in the event that there are concentrations of constituents present in soils beneath the ABT Units that exceed the NMED residential soil screening levels.



The preferred method to address any such releases from the ABT Units is to remove and dispose the impacted soils at a permitted off-site landfill so as to obtain Corrective Action Complete Without Controls and meet the requirements of 40 CFR§265.228 (a)(1) for clean closure. If the volume or depth of impacted soils is such that a complete removal action is financially or technically impracticable, then a revised Closure Plan will be submitted in accordance with 40 CFR §265.228 (a)(2).

4.3.1 Soil Excavation

Soil containing concentrations of constituents above the applicable NMED residential screening levels will be excavated for off-site disposal. The soil will be excavated using back hoes, track hoes, long-reach excavators, or similar equipment. The excavated soil will either be stocked piled on poly sheeting within the footprint of the ABT Units or placed directly into roll-off boxes. The soil will be sampled and analyzed for waste characterization at a minimum of one sample per 20 cubic yards. The samples will be analyzed for hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste, including reactivity, corrosivity, ignitability, TCLP RCRA metals, TCLP SVOCs, and total VOCs. If the material does not exhibit any hazardous characteristics, then it will be transported for off-site disposal to a NMED permitted non-hazardous landfill. If the material exhibits hazardous characteristics, then it will be placed into appropriate RCRA containers for disposal offsite as hazardous waste.

4.3.2 Confirmation Sampling

After removal of impacted soils, the underlying soils will be sampled along all faces of the excavations with an approximate spacing of 20 feet between sample grid locations. Sample results will be compared to NMED residential soil screening levels. Locations exhibiting constituent concentrations in excess of NMED residential soil screening levels will be further excavated and the excavated soils will be stockpiled within the footprint of the ABT Units or placed directly into roll-off boxes in anticipation of characterization, transport and off-site disposal. This process will be repeated until impacted soils with concentrations exceeding the NMED residential Soil Screening Levels have been removed from beneath the ABT Units.

4.4 Closure Certification

Upon completion of all activities, a Closure Certification Report will be prepared in accordance with 40 CFR §265.115 Certification of Closure. The certification will described how the ABT Units were closed in accordance with the approved Closure Plan.






Section 5 Construction Schedule

The schedule for closure of three ABT Units is as follows:

Description	<u>Duration</u>
Removal of liquids in ABT Units Drying of residual solids Testing of residual solids Removal of RCRA liners Removal of Non-RCRA lines/leachate collection Soil Investigation (including analyses) Soil excavation Final confirmation sampling and Analyses	1 week 4 weeks 1 week 2 weeks 3 weeks 6 weeks 8 weeks <u>4 weeks</u>
Total time required	29 weeks

The current cost of closure for the aeration ABT Units is estimated at \$302,800, based on the estimated volumes and other assumptions as detailed in the cost estimate tables presented in Appendix B.







Section 6 References

- EPA, 1987, Data Quality Objectives for Remedial Response Activities; United States Environmental Protection Agency, Office of Emergency and Remedial Response and Office of Waste Programs Enforcement, OSWER Directive 9355.0-7B, 85p
- EPA, 2000, Guidance on Choosing a Sampling Design for Environmental Data Collection, EPA/240/R-02/005, EPA QA/G-5S, 168 p.
- EPA, 2006, Guidance on Systematic Planning Using the Data Quality Objectives Process, United States Environmental Protection Agency, Office of Environmental Information; EPA/240/B-06/001, p. 111.

Appendix A

Management of Investigation Derived Waste



Investigation-Derived Waste Management Plan

All investigation-derived waste (IDW) will be properly characterized and disposed of in accordance with all federal, State, and local rules and regulations for storage, labeling, handling, transport, and disposal of waste. The IDW may be characterized for disposal based on the known or suspected contaminants potentially present in the waste. It is assumed that there are no listed wastes present in environmental media at any of the planned investigation areas.

A dedicated decontamination area will be setup prior to any sample collection activities. The decontamination pad will be constructed so as to capture and contain all decontamination fluids (e.g., wash water and rinse water) and foreign materials washed off the sampling equipment. The fluids will be pumped directly into suitable storage containers (e.g., labeled 55-gallon drums), which will be located at satellite accumulation areas until the fluids are disposed in the refinery wastewater treatment system upstream of the API separator. The solids captured in the decontamination pad will be shoveled into 55-gallon drums and stored at the designated satellite accumulation area pending proper waste characterization for off-site disposal.

Drill cuttings generated during installation of soil borings and monitoring wells will be placed directly into 55-gallon drums and staged in the satellite accumulation area pending results of the waste characterization sampling. The portion of soil cores, which are not retained for analytical testing, will be placed into the same 55-gallon drums used to store the associated drill cuttings.

The solids (e.g., drill cuttings and used soil cores) will be characterized by testing to determine if there are any hazardous characteristics in accordance with 40 Code of Federal Regulations (CFR) Part 261. This includes tests for ignitability, corrosivity, reactivity, and toxicity. If the materials are not characteristically hazardous, then further testing will be performed pursuant to the requirements of the facility to which the materials will be transported. Depending upon the results of analyses for individual investigation soil samples, additional analyses may include TPH and polynuclear aromatic hydrocarbons (PAHs).

Purge water generated during groundwater sampling activities will be containerized in 55-gallons drums and then disposed in the refinery wastewater treatment system upstream of the API separator. All miscellaneous waste materials (e.g., discarded gloves, packing materials, etc.) will be placed into the refinery's solid waste storage containers for off-site disposal.



Appendix B

Closure Cost Estimate













TABLE 1 Final Closure Cost Estimate Western Refining - Bloomfied Refinery North and South Aeration Lagoons May 4, 2010

ltem	Description	Quantity	Units	Unit Cost	Cost			
Professional Services								
1	Analyses for waste characterization & investigation/soil confirmation sampling (Table 2)	1	LS	\$118,000	\$118,000			
2	Final closure report	1	LS	\$20,000	\$20,000			
3	Project administration (engineering, bidding, construction administration, etc.)	1	LS	\$18,700	\$18,700			
	Construction							
5	Mobilization	1	LS	\$6,200	\$6,200			
6	Administrative costs (office facilities & staff, H&S plan, SWPPP, insurance, eqpmt decon, QA/QC, etc.)	1	LS	\$12,500	\$12,500			
7	Dewater lagoons (1 ft water over 25,092 sq. ft.) Dispose water at authorized on-site discharge	188,000	Gal	\$0.011	\$2,100			
8	Excavate and load sludge from aeration lagoons for disposal at local NMED permitted landfill. ⁽¹⁾	310	СҮ	\$4	\$1,200			
9	Transfer sludge from aeration lagoons to local NMED permitted landfill. ⁽²⁾	403	CY	\$12.5	\$5,000			
10	Dispose of sludge at local landfill as Special Waste	403	CY	\$16.5	\$6,600			
11	Remove and dispose of RCRA liners at local landfill ⁽³⁾	1	LS	\$5,340	\$5,300			
12	Remove and dispose of non-RCRA composite geotextile/geonet layer and 100 mil liner at local landfill; stockpile cemented amended sand ^{(4) (5)}	1	LS	\$7,780	\$7,800			
13	Transport and dispose of cemented amended sand at local NMED permitted landfill as special waste ⁽⁵⁾	605	СҮ	\$29	\$17,500			
14	Excavate upper two feet of soils across all lagoons ⁽⁶⁾	1,859	CY	\$5	\$9,300			
15	Transport and dispose of excavated soils at local landfill as Special Waste	2,416	СҮ	\$29	\$70,100			
16	Demobilization	1	LS	\$2,500	\$2,500			
	TOTAL				\$302,800			

Notes

- Assumed dried sludge in-place volume = 25,092 sq. ft. x 0.333ft = 310 cy (special waste). Estimated truck yards = 310 cy x 1.3 (fluff) = 403 cy. Estimated excavation cost = \$4/cy
- 2 Estimated transportation cost to NMED permitted landfill in Aztec, NM = \$12.50/cy (\$125/hr @ 2hrs per trip & 20 yd. truck)
- 3 Assume three 20-yd trucks @ \$16.50/cy; \$750 transporation & 72 hours labor @ \$50/hr = \$5,340
- 4 Assume four 20-yd trucks @ \$16.50/cy, \$1,000 transporation, 72 hours labor @ \$50/hr, & stockpile cemented amended sand (\$4/cy x 465 cy) = \$7,780
- 5 Estimated in-place volume of cemented amended sand = 25,092 sq. ft. x .5 ft. x 1.3 = 465 cy. Estimated truck yards = 465 cy x 1.3 (fluff) = 605 cy
- 6 Estimated in-place volume of excavated soils beneath lagoons = 25,092 sq.ft. x 2 ft. = 1,859 cy. Estimated truck yards = 2,203 cy x 1.3 (fluff) = 2,416 cy
 - LS Lump Sum
 - CY cubic yard
 - Gal gallon





TABLE 2 Investigation & Confirmation Sampling Cost Estimate Western Refining - Bloomfied Refinery North and South Aeration Lagoons

Analysis	# of Samples	Cost/Sample	Costs
Wast	e Characterizatior	n Samples ¹	
VOCs 8260B	155	\$90	\$13,950
TCLP SVOCs 8270C	155	\$220	\$34,100
Haz. Characteristics	155	\$140	\$21,700
TCLP Skinner List Metals	155	\$185	\$525
Sampling Labor	40 hours	\$75/hour	\$3,000
······································		Subtotal	\$73,275
Investi	gation/Confirmation	on Samples ²	
VOCs 8260B	71	\$90	\$6,390
SVOCs 8270C	71	\$220	\$15,620
TPH 8015B (GRO, DRO, MRO)	71	\$90	\$6,390
Skinner List Metals	71	\$185	\$13,135
Sampling Labor	40 hours	\$75/hour	\$3,000
Subcontract drilling			\$12,000
		Subtotal	\$44,535
		Total	\$117,810

1 - sludge samples (25,092 sq. ft. x .33 ft. = 310 yds / 20 yds/sample) = 16 samples; cement amended sand samples (25, 092 sq. ft. x .5 ft = 465 yds / 20 yds/sample) = 24 samples; excavated soil samples (25,092 sq. ft. x 2 ft. x 1.2 (fluff factor) / 27 (cu. ft/yd.) = 2,230 yds / 20 yds/sample) = 112 samples; potential leachate samples (RCRA liner, non-RCRA liner & French drain) = 3 samples; estimated total of 155 characterization samples

2 - assumes two samples (0-6" & 18-24") at each of 15 soil borings & 15 sidewall samples, six duplicate samples, and five equipment blanks

TPH - total petroleum hydrocarbons

GRO - Gasoline Range Organics

DRO - Diesel Range Organics

MRO - Motor Oil Range Organics

VOCs - volatile organic compounds

SVOCs - semi-volatile organic compounds







BILL RICHARDSON Governor

DIANE DENISH Lieutenant Governor

NEW MEXICO ENVIRONMENT DEPARTMENT

Hazardous Waste Bureau

2905 Rodeo Park Drive East, Building 1 Santa Fe, New Mexico 87505-6303 Phone (505) 476-6000 Fax (505) 476-6030 www.nmenv.state.nm.us



RON CURRY Secretary

SARAH COTTRELL Deputy Secretary

CERTIFIED MAIL - RETURN RECEIPT REQUESTED

May 20, 2010

James R. Schmaltz Environmental Manager Western Refining Southwest, Inc Bloomfield Refinery. P.O. Box 159 Bloomfield, NM 87413

RE: APPROVAL

PARTIAL CLOSURE OF THE NORTH AND SOUTH AERATION LAGOONS CLOSURE CERTIFICATION REPORT WESTERN REFINING SOUTHWEST INC., BLOOMFIELD REFINERY EPA ID # NMD089416416 HWB-GRCB-09-007

Dear Mr. Schmaltz:

The New Mexico Environment Department (NMED) has received Western Refining Southwest Inc., Bloomfield Refinery's (Western) *Response to March 24, 2010 Notice of Disapproval North and South Aeration Lagoons Closure Certification Report* (NOD Response), dated April 19, 2010. NMED hereby approves partial closure of the Aeration Lagoons.

Western has completed partial closure of the North and South Aeration Lagoons (N & S ALs). The Aeration Lagoons are no longer treating hazardous waste, all sludges and liquids were removed, and the liners were cleaned and repaired as necessary. The Aeration Lagoons are currently operating, but are only used to treat non-hazardous waste water.

Although NMED has approved partial closure, Western is still required to complete final closure of the N & S ALs when they cease operation. The final closure process must include complete removal of the lagoons and all underlying contaminated soils. Final closure may be completed as part of corrective action under the July 27, 2007 Order (Order) with Group 9, SWMU 13 and in

Mr. Schmaltz May 20, 2010 Page 2 of 2

accordance with 40 CFR 265 Subpart G. The cost estimate associated with the closure process has been addressed by NMED in a separate letter dated April 5, 2010 (*Response to Western Refining March 15, 2010 Financial Assurance for the Gallup Refinery EPA ID* #NMD000333211 and the Bloomfield Refinery EPA ID #NMD089416416 Western Refining Southwest, Inc.).

In addition, Western's response to Comment 7 in the NOD Response states "[t]he approved IDW Management Plan mentioned in the North and South Aeration Lagoon Closure Certification Report was intended to reference the same IDW Management Plan that was included in the earlier approved Group 2 Investigation Work Plan, and has since been included in the earlier approved Group 2 Investigation Work Plans. A copy of the IDW management Plan was followed during the aeration lagoon monitoring well installation activities [and] is provided in (Attachment D)." Attachment D is written in the future tense indicating what will be conducted instead of stating what was conducted. In the future, any IDW management reporting must address what was actually completed during the associated field activities, regardless of whether it was cited in an earlier plan. No revision is necessary because the IDW activities are complete and chemical analyses was conducted on the soils, which were accepted by the disposal facility.

Please contact Hope Monzeglio of my staff at (505) 476-6045, should you have any questions.

Sincerely,

Whn E. Kieling

Program Manager Permits Management Program Hazardous Waste Bureau

cc: J. Bearzi, NMED HWB
D. Cobrain, NMED HWB
H. Monzeglio, NMED HWB
C. Chavez, OCD
L. King, EPA Region 6
N. Stone, EPA Region 6
A. Hains, Western Refining El Paso File: Reading File and GRCB 2010





RECEIVED OCD

2010 APR 21 P 1:15

James Bearzi, Bureau Chief New Mexico Environment Department Hazardous Waste Bureau 2905 Rodeo Park Drive East, Bldg 1 Santa Fe, NM 87505

Certified Mail #: 7007 2560 0002 5890 7277

April 19, 2010

Re: Response to March 24, 2010 NOTICE OF DISAPPROVAL North and South Aeration Lagoons Closure Certification Report Western Refining Southwest, Inc., Bloomfield Refinery EPA ID# NMD089416416 HWB-GRCB-09-007

Dear Mr. Bearzi:

Western Refining Southwest, Inc., Bloomfield Refinery has prepared the following responses to comments 1-7 on the referenced Closure Certification Report. A response to comment 8 will be submitted separately.

Comment 1

In Section 4.1 (Sludge Characterization, Removal, and Disposal) on pages 5 and 6, Western references the aeration lagoons as follows: South Aeration Lagoon (#1 Aeration Lagoon), the Northwest Aeration Lagoon (#2 Aeration Lagoon), and the Northeast Aeration Lagoon (#3 Aeration Lagoon). The Report does not include a figure that identifies each aeration lagoon by name, Western must submit a figure that labels the aeration lagoons as described above.

Response 1

A figure labeling the aeration lagoons as described above is provided (Attachment A).

Comment 2

In Section 4.1 (Sludge Characterization, Removal, and Disposal), pages 5 and 6, Western describes the sludge removal and disposal process for each Aeration Lagoon (AL). In the response letter, Western must describe how the sludge samples were collected (e.g., shovel, Encore® sampler) and indicate for each AL the volume of sludge removed and transported off-site for recycling or disposal.

Response 2

Samples were collected from the #1 Aeration Lagoon (AL) after water was removed from the lagoon and the sludge was allowed to partially dry. Sludge samples were collected via a shovel by a Western employee who was suspended over the grid sections

of the #1 AL in a JLG Lift (a man-basket). Samples were placed in eight ounce sample jars that were supplied by the analytical laboratory. The samples were sent to Hall Environmental Analytical Laboratory for analysis. Approximately 851,930 pounds of sludge was removed for disposal from the #1 AL during the clean out.

Sludge samples from #2 AL and #3 AL were collected while both lagoons were in service. Western employees used a rowboat to traverse the lagoons and position over each respective sample location. A sludge judge was used to collect each sludge sample from #2 AL and #3 AL. The sludge judge is used similar to that of a well bailer. It is constructed of 3-inch piping approximately 10 feet in length, and equipped with a ball-check valve on one end. At each sample location, the sludge judge (ball check valve end first) was extended down into the sludge to the bottom of the lagoon. As the sludge judge is pulled up, the check ball closes the bottom of the tube. Excess liquid was decanted from the 3-inch sludge tube, and sludge was then placed in a stainless steel bucket. Samples were collected from the bucket and placed in eight ounce sample jars that were supplied by the analytical laboratory. The samples were then properly packaged and sent to Hall Environmental Analytical Laboratory for analysis. The bucket was rinses of sludge using pond water after each sample was collected.

Approximately 2,476,880 pounds of material (including flyash) were removed from the #2 AL during the clean out. Approximately 1,998,780 pounds of material (including flyash) were removed from #3 AL.

Comment 3

t

Ť.

In Section 4.1 (Sludge Characterization, Removal, and Disposal), pages 5 and 6, Western states the sludges from the Northwest Aeration Lagoon and the Northeast Aeration Lagoon were "removed via vacuum truck, mixed with fly ash, and transported and disposed of at the San Juan County Landfill in Aztec, New Mexico." In the response letter, Western must discuss if the San Juan County Landfill required chemical analyses for the sludge and any analyses related to the fly ash before acceptance for disposal. Western must provide documentation that demonstrates the landfill accepted the waste and verify that all waste manifests are available for review at the refinery.

Response 3

San Juan County Landfill required chemical analysis for the sludge which was also provided to NMED in the *North and South Aeration Lagoons Closure Certification Report* in Attachment 6.0. Chemical analysis of the flyash was provided to Waste Management (WM) by Clean Harbors. The approved WM profile for the non-hazardous sludge is provided in **Attachment B**. Also provided in **Attachment B**) are copies of e-mail correspondance from Clean Harbors to WM providing the MSDS and analysis of the flyash, and a letter from WM to Clean Harbors confirming that San Juan County Landfill can accept the non-hazardous waste.

All waste manifests from the project are kept on-site and available for review at the refinery.

Comment 4

ý.

In Section 4.1 (Sludge Characterization, Removal, and Disposal), page 6, Western states "[o]n October 21, 2008, eight sludge samples were collected in a grid pattern from the Northeast ABT unit (#3 Aeration Lagoon)" and in the following paragraph states "[t]the sludge sample location map is presented in Attachment 5." The sludge location map show and the laboratory results report seven sample locations, not eight as indicated in the text. In the response letter, Western must clarify this discrepancy and submit replacement pages with modified text and revise the sludge location map accordingly.

Response 4

The discrepancy in the text was a typographical error. There were seven samples collected from the #3 Aeration Lagoon. A replacement page for page 6, Section 4.1 is provided (Attachment C).

Comment 5

In Section 4.3 (Collection of Soil and Flush Water Samples), page 8, Westerns states "[s]tained soil was discovered underneath the discharge piping from the South ABT unit to the Northwest ABT unit when the piping was replaced. The soil was removed, placed on containment, and sampled. The characterization samples were placed on ice for preservation, and shipped to HEAL for analysis of TCLP Metals (EPA Method 6010B) and Benzene (EPA Method 8260B). Analytical results indicate non-detect on all analyses. The soil was used as backfill. Stained soil was found under the discharge piping from the Northeast ABT unit to the suction of P-606. The soil was removed, placed on containment, and sampled. The characterization samples were placed on ice for preservation, and shipped to HEAL for analysis of TCLP Metals (EPA Method 6010B) and Benzene (EPA Method 8260B). Analytical results indicate non-detect on all analyses. The soil was used as backfill. Laboratory reports are provided in Attachment 6."

Because the stained soil was used as backfill and not disposed of off-site, additional chemical analysis should have been conducted in addition to the toxicity characteristic leaching procedure (TCLP) analytical method for metals and benzene. Even though a sample is not hazardous, it does not mean the sample is not contaminated; before the soil was used as backfill, the soil samples should have been analyzed for diesel range organics (DRO), gasoline range organics (GRO), and volatile organic compounds (VOCs), semi-volatile organic compounds (SVOCs), and total metals. In the future, if soils are considered for reuse on-site, these additional chemical analyses must be conducted and reported to NMED prior to use as backfill. The analytical results must be compared to NMED's Soil Screening Levels in order to determine if the soil can be reused on-site and the location of the soil reuse must be approved by NMED. In the response letter, Western must indicate how much soil was removed and used as backfill and identify where the soil was placed.

Response 5

Approximately one yard of material was removed from each site. In order to replace the discharge piping that extends from the South ABTUT to the Northwest ABTU, the surrounding soil was excavated and placed in a stockpile adjacent. Once the piping was

replaced, the same soil was put back in-place. Similarly, the soil removed from under the discharge piping from the Northeast ABTU to the suction of P-616 was used to backfill the excavation created by replacing the discharge piping from the Northeast ABTU to the suction of P-616.

Comment 6

ì.

The Closure Report did not address well development or describe the collection of soil samples during the installation of the monitoring wells. In the response letter, Western must provide a description of the well development process and explain how the soil samples were collected during the installation of the monitoring wells (e.g., new monitoring wells were developed within 10 days of installation and developed by pumping/surging; soil samples placed in four ounce jars provided by the laboratory, no headspace was left in the jar to avoid volatilization of VOCs).

Response 6

Soil samples were collected using a split-spoon sampler. At each selected sample interval, two soil samples were collected for VOC analysis. An Encore® Sampler was used for collection of soil samples for low-level VOC analysis, and the second sample was placed in a laboratory-prepared VOA container with a methanol preservative. The remaining soil sample material was placed in pre-cleaned four ounce glass jars provided by the laboratory for SVOC and metals analysis. The glass jars were filled completely to avoid headspace for potential volatilization.

Field activities associated with the installation of the Group 1 North and South Aeration Lagoon monitoring wells were coordinated with the commencement of the activities associated with Group 3. Therefore, well development of the new monitoring wells near the aeration lagoons commenced during the last week of April 2009, which immediately followed well completion activities associated with the Group 3 monitoring wells. This consolidated schedule was necessary in order to ensure that groundwater samples from all newly constructed wells could be collected no later than five days after completion of well development, as stated in Section VIII.B.2 of the Order.

Each monitoring wells was developed using a combination of mechanical surging and air-lift techniques. The following well development activities were conducted at each new monitoring well prior to groundwater sampling activities.

Using a surge block attached to the end of the drill rod, groundwater was forced to flow in and out of the well screen by the repeated upward and downward motion of the surge block along the entire length of the well screen. The repeated plunging motion drew filter pack fines and loosened sediment into the well casing, improving the water quality within the surrounding formation and filter pack.

Once the well was surged for a minimum of 20-minutes, the surge block was removed and the air-lift apparatus was used to remove the loosened sediment and fines from inside the well casing. Using an air compressor and dedicated 1-inch PVC eductor piping, compressed air was injected into the well. The air flow rate was manually adjusted to produce a continuous flow of water/sediment mixture out of the top of the well casing via the 1-inch eductor piping. The groundwater/sediment mixture discharged directly into a 55-gallon drum. A glass jar was used to capture a sample of the purge water every 15 minutes to monitor the improving clarity of the purge water. Air lifting ceased once the purge water was relatively clear.

Comment 7

î

In Section 5.7 (Investigation Derived Waste management), page 18, Western states "[f]oliowing completion of all drilling activities, all soil and sampling fluids were disposed of in compliance with the approved IDW Management Plan. A composite sample of the soil cuttings was collected and sent to Hall Environmental Analytical Laboratories for analysis. The soil cuttings were characterized as non-hazardous and disposed of off-site in accordance with the approved IDW management plan."

- a. It is not clear what "approved IDW Management Plan" Western is referring to. The Closure Plan North and Sough Aeration Lagoons (Closure Plan), dated May 2008, does not have an IDW Management Plan. In the response letter, Western must discuss the IDW Management Plan and what protocols were followed.
- b. In reference to the composite sample, Western must provide a list of laboratory analyses conducted and include the laboratory results. Western must also describe how the composite sample was collected (e.g., a small amount of soil was collected from five drums and placed in a four ounce jar provided by the laboratory or soil was collected from five drums and homogenized before placement in a sample container). This information must be included in the response letter.

Response 7

The approved IDW Management Plan mentioned in the North and South Aeration Lagoon Closure Certification Report was intended to reference the same IDW Management Plan that was included in the earlier approved Group 2 Investigation Work Plan, and has since been included in subsequent RCRA Investigation Work Plans. A copy of the IDW Management Plan that was followed during aeration lagoon monitoring well installation activities is provided (**Attachment D**).

Drill cuttings and excess sample material associated with the soil borings were containerized in DOT certified 55-gallon drums, which amounted to a total of approximately five drums of material. A small amount of soil was collected from each of the drums and placed in a Ziplock bag. One composite sample was collected from the Ziplock bag. The soil sample was placed in four ounce jars and sent to the laboratory for waste characterization analysis. The composite sample was analyzed for the following:

- BTEX and MTBE by EPA Method 8021B;
- Polycyclic Aromatic Hydrocarbons (PAHs) by EPA Method 8310;
- TCLP RCRA 8 Metals by EPA Method 6010B; and
- Ignitability, Corrosivity, and Reactivity.

The associated analytical is provided in **Attachment E.** All analytical results for waste characterization purposes were submitted to Waste Management, Inc. for review and approval of acceptance. Soils were disposed of at the Painted Desert Landfill in Joseph City, Arizona. IDW fluid, which includes groundwater purge water, was disposed of through the refinery on-site wastewater treatment system.

If you have questions regarding the above response or would like to discuss the revised Investigation Report, please contact me at (505) 632-4171.

6

Sincerely,

٦

Sames R. Schmaltz Environmental Manager Western Refining Southwest, Inc., Bloomfield Refinery

cc: Hope Monzeglio – NMED HWB Dave Cobrain – NMED HWB John Kieling – NMED HWB Laurie King – EPA Region 6 Nick Stone – EPA Region 6 Carl Chavez - NMOCD Allen Hains – Western Refining El Paso

ATTACHMENT A

1

North and South Aeration Lagoon Map

.



ATTACHMENT B

Sludge Characterization and Disposal Documents

\$

Fax sent by : 3032809848

1

WASTE MANAGEMENT

11-05-09 11:24a Pg: 2/8



November 5, 2008

National Logistics Clean Harbors Environmental Services 42 Longwater Drive Norwell, MA 02061 FAX #: 781-792-5930 RE: Waste Approval for Settling pond sludge Generator: Western Refining Southwest Inc. WM Profile Number: 100951NM Expiration Date: 02/05/2009

Dear National Logistics:

This letter shall serve as written confirmation that San Juan County has all necessary permits and licenses to accept the waste materials described on the above referenced profile. Limitations on acceptance if any are noted at the base of this letter.

The profile processing approval fees for the above waste stream is **the disposal rate is** currently **the disposal rate is** minimum. (This pricing does not include any applicable fuel surcharges, environmental fees, state or local fees.)

A Non-Hazardons Waste Manifest must accompany each load. Please be sure to enter your profile number on the Non-Hazardous Waste Manifest. Your account representative is Dan Callaghan. He can be reached at 303-886-9694.

Visit Waste Management's updated web page at WMDisposal.com. We've made profiling your waste easier and more efficient!

Thank you for allowing Waste Management to assist you with your waste disposal needs.

Sincerely,

Michelle Matzke Waste Management Technical Service Center

If not otherwise specified herein, no special conditions exist:

PO NUMBER REQUIRED BEFORE WASTE CAN BE ACCEPTED AT LANDFILL.

Please see attached waste acceptance conditions.

From overyday collection to environmental protection. Think Green? Think Waste Management.

🛞 Polaiki wa 1645 puni-ramanne neoniki paper

NOV-05-2008 01:21PM FAX:3032809848

ID: CH CORP MAINTENANCE PAGE: 002 R-94%

Waste Management

Western Gtoup Industrial Sales Office 7780 East 96th Avenee Henderson, CO 80640 (800) 963-4776 (666) 365-0175 Fax VASTE MANAGEMENT



1

INDUSTRIAL WASTE & DISPOSAL SERVICES AGREEMENT

Exhibit A

WM Profile #	100951NM
WM Facility ID	San Juan County

CUSTOMER BILLING ADDRESS	CUSTOMER CONTRACTING ADDRESS (It citizent from Billing Acidises)	CUSTOMER SERVICE LOCATION (If different from Belling Address)
Clean Harbora Environmental Services		Western Refining Southwest Inc.
42 Longwater Drive		#50 Road 4990.
Norwell, MA		Bloomfield, NM
Contact Phone: (781) 792-5000	Contact Phane:	Contact Phone:

WM.Customer Service Phone:	800-963-4776	WM Contact:	Dan Calleghan	WM Contact Phone:	303-885-9694
			••••••••••••••••••••••••••••••••••••••	· ·	
		Smith	A Fraterroundings		

Generator:	Western Refim	Western Refining Southwest Inc.				
Ground Transporter:	1					
PO#, SO#, Job #.	REQUIRED PR	NOR TO SHIPMENT TO L	ANDFILL	1		
Waste Description:	Setting pond si	ludge				
Disposal Cost:	0					
Profile Fee:		444 ¹⁴⁴	······			
Additional Cost: (describe)	Applicable fuel	surcharges, environmenta	fées, state and local taxes.			
Additional Cost: (describe)						
Transportation Fee:	N/A		······································	an a		
	Quantity:	Size:	Quantity:	Size:		
Containers provided by WM:	Quantity:	Size:	Quantity:	Size:		
Thats over the survey of the	Quantity:	·.\$120;	Quantity:	Size:		
Pick-up Frequency;						
Contract Expiration Date:	As per agreeme	ent				
Additional information:	An Industrial W	aste Disposal Manifast is F	lecuired for all loads.			
Silver Color State	Star 2 ded	an air l in san a	TERIBA FIEN	NUE TWINE FURE FURE FURE		

THE WORK CONTEMPLATED BY THIS EXHIBIT A IS TO BE DONE IN ACCORDANCE WITH THE TERMS AND CONDITIONS OF THE INDUSTRIAL SERVICES AGREEMENT OR OTHER CONTRACTUAL AGREEMENT BETWEEN THE PARTIES DATED:

COMPA	NΥ			CUSTOMER	
By:	· · · · · · · · · · · · · · · · · · ·		By:		
Name	Dan Caliaghan	Date	Name:		Date
Title	Industrial Account Manager		Title:	,	
				· · ·	

Waste Management

المصيفة المتحد فالمتحم والمتعالي

	Generator's Nonka	zaráous Wa	aste Profil;	Shoot	National Account C	Ustomer
	Requested Disposal Facility San Juan Con	inty Landill	Profile Xu	mber 1009	51NM	
versy's master because	Renewal for Profile Number	Waste	Approval Expiratio	n Date	02/05/09	Say
R. Waste Gene	natur Facility Sufermation (mast	enterof location	n of waste gra	n mitum (cr	kepins)	- 4/05/02
1. Generator Name:	Western Retining Souther	ust the -	Blaza Field	Rafin	r.¥	
2. Site Address: H	50 Reed 4990	7 Email Addre	54:		-	
3. City/DP: Blan	sheld 87.413	. 8. Alloria: 525	5-632-4161	9. 642	505-632-3111	
4. State: Milit	Marilo	10. NAICS Code			······	
5 Sounds 54N	Than	1. Generator U	SEPA ID T NA D	089416	41. ta	
6. Contact Name/Th	te Cindy Huckey minimum	12. State 108 (1	Fapplicable):	· · · · ·		
B. Customor L	domination () same as above	P. O. Nuraber:				
I. Furtherner Name:	Clean Haritors Environmental Services	6 Ph-car 781-70	2,5090	CLES VRS.		in the second se
2: Billing Midness	2 Longwater Drive/ P.O. Box 8149	2 Franking Parties	*****	A	******	-
- Christen and T	D. Norwell MA 02081	 Transporter star Transporter star 	R ALZ manual Al			Mara
S. Cantack Nomes N	mining) 1. selection	C. Ingraporter IV	P. P. 9999.31	175 - 187 - 18 - 62 ft, re voter - 197 -		*
R. Charlent Emple	Brighing Science and the second	SA Peter Conta and	1100);			
. Compost surgers		ROUCING SCALE RIG	Lir's any age in the second			
C. Watth Stress	nn frefer i rektistere					
1. DESCRIPTION	Caller Paul SI					-2
a. Los) mon wasz Santu Monte C	WARNEL WAR AND A LONG AND A	ege	يل 196 بر	niningara ang ikana		
h. Berrita Pm	MC(3), OTT			ikhandre lannan menetaigungu.		
Nau Halles	chous Setting Popul Stud	47	1 17	7.1		n i
Tacked and	or to step liketion - No;	J-riaBardon	s before	B PP20han	g With thy A	s.a.
ALTING	The Alter has a march & it	Stor Eldner	low thomas 1	يدد مينو برز	The lover to a 2-	1 A A
States FL	Parl	~sc~ . ? [. (. 7)	100 WI MER		11.7. Marcar. (77)	
c. Typical Coloris	7 1 Vin 184 10 10 10 10 10 10 10 10	Ai9				and the second second
. d. strong unors	La ves va ko pescribe: <u>Marca</u>	Charles and the second s		a a de la companya d La companya de la comp		SV23
e roysical sidle	at roma in some in a ciquite in the	wast a seigi-sa	ମହାର ମାସପ୍ରିବ 🛁	Osner:		(
f. Jyerst Lt.	angle layer of Multi- layer of NA					:
G. Water Reaction	er ut ver ut no it ves, Describet	مىسىرىيە بىلەرتېيىسىرە ت ەت بىر دېرە يەرىدۇ ، بىلە بە تەت				
h. Free Liquid Ra	npe (%): 10 14 NA(3000)				
i. pH Range:	4 52 W 2.1-23.4 W 222.5 W NA(satid) G Actua	li			
j. Ljovid Resh Po	i)ot: Gr < 148°F G ≥ 143°F	us NA(solid)	C Actual:	······		
k, fiaminable 50	10: Lat Yes BCT NO	e thá của Nama	SHALL A BART	E. Bran Asses	(hade	-
C. FIIVSKER CONSU	EXCISES LINE BE LUTISLEVITIES OF WORLD SLEEP	1 - ee.g. 302 O-por	a, versou unconori	Landt Exect	they of Famer	**-
2. Satting P	and Student	#0%		50%		
12 fight	H	30 9a		60%	د. ووديد بدوير ² ن مس	
			الدينا ومشاهده ومساعيه ومعاوية والمعار			-
G		· · · · · · · · · · · · · · · · · · ·		; ; ; ; ;	and an and a second	-* ; ; ?
We wanted and the second				الادرادية فأستحصب		-
2. ESTIMATED QUANT	TTY OF WASTE AND SHIPPING THFORMATION	· · · · · · · · · · · · · · · · · · ·	ار - <mark>- با الم</mark> الية الم	s serbinare	aren de Klannen i in de services sources i s	
2. D One Trine Eve	in Diffuse Diseperat Event					
b. Estimated And	est Quantity: 1000 O Jons X	Cubic-Vards 20.0	arm 🖸 Gallans	⊐-0ther(sp	ecify):	
c: Snipping Frequ	ieng:unit*'p	N- O Komb 3	Uvarter 🔟 Yes	er 🗶 One	Jime Q Other	
d. Is this a U.S. I	Separtment of Transportation (USDOT) Haze	indous, Haterial? (11	yes, answer es)	🗆 Yes 😿	No	
e. USDOT Shippin	g Description (if apolicable):					
3. SAFETY REQUIREN	ENTS (Handling, PPE, etc.): NIA	and the second second second second				in J
ASPAR Spicta Same	mania ina mining and a second s	Pace F of 2		a that data from the Bask may wrong a series		3.96
- X & X & 17 (13 L & WW1) [2 (12 C	NOWP by Alterna	1 12 2 1 1 A AT 1 11			- 103 - 1	

VAVAN

ĩ

and the second s	*	an and the second second			
venerator'	23	Romitazariare	WATCH	These	£21.
And a state of the second s	447	A R. MALE AND COLUMN AND A R.	ANY 201 15 12 07		** 6. 5

the state of the second st	A &	10	OPSINA	····
S this a USEPA (ACTER Part 281) Atoms to another in the Part 281	Capullecs)		~ 3	<u></u>
Is this easte included in one or more of rategories below (Check al) D botted Hazardous Waste Treated Hazardous Waste Operis. In the waste from a federal (40 CFR 300, Appendix B) or state mond. Does the waste from a federal (40 CFR 300, Appendix B) or state mond. Does the waste represented by this waste profile sheet contain radio a. If yes, is disposed regulated by the Nuclear Regulatory Commission 5. If yes, is disposed regulated by the Nuclear Regulatory Commission	Inan (your seas representative) Inan (your seas representative) Inan (your seas of the 261,2 and Charlenteristic Hazardous Waster sted dean-up? If yes, see instruction active material? If in the season of the season of the season of the active material?	a docomentation. Lu d'Ila 15 e.e. Par Carla da Added 11/01P es Q. Na		2 340 2 440 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
bes the white represented by this waste multile sheet contain convert	BARDENS Strandsted Billuthlandsta	25 July 143 A Distance of DOD 1977	Day X	na i
e. If yes, is disposal regulated under ISCA?	amidewild of regulatory rosychich hade	s D'No	للكو 19 <u>25</u> استا	. 43
Does the waster contain untreated, requitated, modical or infectious is	aste?		OB S	Ne
TO THE MONTH BAC ROMANDING MALES CALL THE LAL NO	and the second	If Yes, Cl. Friable	J Noo Frie	o₩e
AD FER AS ENDERED ACKER 12	of source of Hazardous Air Pollutar	es (Site Remediatio	acineshap; Asto	
16 and the state in a second and the	enstaan aster die ookleise bewerken van	Li Yes	XNO	
E Concenter Corrigation (Manus and and and	Donat of Octomination?	Li Yes	LI NO	
signing this isometrary Watthe Profile Sham I harming rathing that all	y by something melows		· · · · · · · · · · · · · · · · · · ·	<u>ڊ</u> ئيند
Internation submitted in this profile and all attached documents or relevant information within the possession of the Longrapy regardle	tale tive and accurate descriptions of known of suspected hazards perial	t the waste meterial		
discused to WM/the Contractor				
Analytical data attached pertaining to the profiled waste was derived 40 CFR 252.20(c) of equivalent rules; and	From testing a representative sample	in accordance with		
Changes that occur in the character of the waste (i.e. changes in the and discussed to Whi (and the Contractor # applicable) prior to provide	process or new analytical) will be ide any the waste to WM (and the Contro	ntified by the Gener sctor (Fapplicable).	ator	
Check all that apply:	Extension And American Contraction Contraction	(P ^r	chillen	S.A.
pall Environmental Analysis Laboratory 25 + 1 AL- 1 Maria	-1. +241-20- 621-821-8	RCI : Pa	pro let_	1 F.
Anacheron 4	ac coardry by leocratory a sample	D \$ 2 900 parameter	s testee).	
Additional information necessary to characterize the profiled was	te has been attached (other than any	and a second		Ma r ¹
indicate the number of estactived pages	FFINASI Salt Livera	laterals Con	2 p 4 p 2	in .
1 I am an agent signing on benalf of the Generator, and the dalaga	bon of authority to me from the Gen	mator for this signar	્યું ન્ડ્સ હ્યવ્યું ક	1
available upon trocess.				,
By Generator process knowledge, the following wasteris not a lisu	ed waste and is below all MLP recula	lory fimits.		
roffication Signatures C inder Guid 200	The Environmental Coe	pdinator_		ې د دنوه
mpany Names Western Returns - Bloom field Kelengen	Name (Print): Ciridy Hu	rtado		
	5 7			
Por V	Pm use only			
nagement Nethod: 🕺 Landfili 🗆 Bioremediation	Approval Decision:	itoved Divor	Approved	
Non nozardous solidification O Other:	Baste Apportal Explicition date:	_02105;	icn.	-
nagament Facility Precautions, Special Handling Procedures or	L'Imitation El Saalt act contais	freedoudet		
approvale .	El Shinment most i	e scheduled into d	ionosi, britti	т 4
frence ettriked	Li Accrevat Numbe	TOBIST SCEOGDSBY	arn shirener	
and a second	Ti Water Nanther	user accompany of	and more come	•
Authorization Name / Title: Die E Andre	Contraction of the second seco	Dates (/	05/08	
the Arth main the 188 Delander to		(Inte)	- -	-
Contraction (1) Notion (1)		19		تر مرجعه مناجع
C. AHMCHNACHUM, (11 NG(4/3 GX)+				
P2008 Vience Histopsonial and Page alla provide i TCCS motals of fluxan (A	lesarce materials Fl	ring, me, k	动物	0:5}:
22008 Vince Hanagerräch 2000 - Prige alle provide 1: Tour metals & figson (A	losarie makrixed Fe	rting me. k	nto sung d	0:5};

الربيد الدار الالمتوريين الدامية وتابا فتتحاج فال

WASTE MANAGEMENT

المراجع والمراجع والمتحد والمحاف المراجع المراجع

1

INDUSTRIAL WASTE Waste Acceptance Conditions

الاراب وحد بدموها والتعديدية المدر المعرامة المحار واللا

				DATE:	21/05/08
GENERATOR:	Western Re	fining South	west, Inc Bloc	mfield Refinery	
WM WASTE COD	E/PROFILE#	10095100	GX AR	EXPIRATION DATE:	02/05/05
			Arf Gar		• • • •

Precautions, Conditions, and Limitations on Approval

Generator Requirements

- Waste Management must be informed if the waste characteristics or process change. Changes to be approved on a case by case basis.
- Generator or its authorized agent has certified that no hazardous waste codes, PCBs, asbestos containing materials; or other prohibited wastes are associated with this waste stream.
- Waste must be absent free liquids.
- 7 Tota waste is a New Mexico Special Waste (industrial solid waste). The following additional requirements apply:
 - A manifest must accompany all loads of special waste in accordance with NMAC 20.9.8.19
 - All drums or containers must be clearly labeled or marked indicating the name and address of the Generator contents, potential health, safety, and environmental hazards associated with the waste (NMAC 20.8.8.10(d)).
- Contact Weste Management to schedule waste for disposal at least 24 nours prior to shipping.
 San Juan Landfill: 505.334,1121

Waste Management Handling Conditions

- Direct bury. Material must be covered immediately. Cover with refuse or soll pitor to compaction. Do not generate dust.
- Specific health and safety concerns associated with this material; <u>Avoid all contact</u>. Pollow general MSS procedures including good hydiene if done in contact with wasse material.

Required PPE is indicated below.

~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~		
$\Box$	aspirator and filters/cartildges:	
$\otimes$	remicel resistant gloves:	
	andard Work Brillorm	
	vek Suit	
X	Il sideshield safety glasses, or	•
X	iemical goggles	
	her PPE depending on conditions of management and disposat	2

Ensure eyewash station is functionet.

🕐 diperial wayabizek polatalishi usazilishimi SW.dor

Generator's Nor	hazardous l	Naste Profi	le Sheet ^N	ational Account Customer				
Requested Disposal Facility San Jua	n County Landfill	Profile	Number 100951N	M .				
WASTE MANAGEMENT   Renewal for Profile Number		ste Approval Expira	tion Date 02	405/09 STRA-				
A. Waste Generator Facility Information (m	nust reflect locat	ion of waste ge	eneration/origin	u/0570				
1. Generator Name: Western Refining Seu	thurst, Inc.	- Bloom fie	ld Refinery					
2. Site Address: #50 Read 4990		iress:						
3. City/ZIP: Bloom Field 87413	2 8. Phone: 5	25-632-410	01 9. FAX: 505	632-3911				
4. State: New Maxilo	10. NAICS Co	de:						
5. County: <u>SAN Juan</u>	11. Generato	г USEPA ID #: <u>ММ</u>	008941641	6				
6. Contact Name/Title: Cindy Huctado Coordin	12. State ID	# (if applicable):		NOV 0 2008				
B. Customer Information 🛛 same as above	e P. O. Number		l By	······				
1. Lustomer Name: Clean Harbors Environmental Service	s, 6. Phone: 781	-792-5000	FAX: 781-792-5	930 -				
2. Billing Address: 42 Longwater Drive/ P.O. Box 9149	7. Transporter	Name:						
3. City, State and ZIP: Norwell, MA, 02061	8. Transporter	ID # (if appl.):	,					
4. Contact Name: National Logistics	9. Transporter	Address:						
5. Contact Email: Directships@cleanharbors.com	10. City. State	and ZIP:						
C. Waste Stream Information								
1. DESCRIPTION	Studan							
a. common waste Name. <u>CCI.IIII J J J J J J J J J J J J J J J J J</u>	<u> </u>	**************************************						
b. Describe Process Generating Waste or Source of Co	ntamination:			······				
NON HAZardous Settling Pond 5	uolgo ,	haf-re	hlonding	14 8 11				
Toged prior to stabilization -	NON - Hazard	ous before	o on any i	a rig Ash				
office attribe A email	I tated (1/1	14/08 FOR a	d detunio	bachemin				
C Tunical (olor(s): Black	1 20-1- 17			File				
d Strong Odge? D Yes M No Describe: Mild	ODOR	**************************************		ulatio				
a Physical State at 70°F: D Solid D Liquid	D Powder Sem	-Solid or Sludge						
f Lavers? IP Single laver (1) Multi-laver [		-source of storage	G Other.					
a Water Reactive? These Principal Strategy of the section of the s	ribe							
b Erec Liquid Pance (%):	NA(solid)							
i of Range: $\Box < 2 \Box 21-124 \Box > 125$	A NA(solid)	tual.						
i Liquid Elach Point: $\square = 140\%$	°F (D MA/solid)	Actual:						
J. Elquid Flash Funct. $\Box < 140$ $\Box \ge 140$								
L. Physical Constituents: List all constituents of waste	e stream - (e.g. Soil O-	80%, Wood 0-20%	): 🛄 (See Attached	1)				
Constituents (Total Compasition Must be > 100%)	Lower Range	Unit of Measure	Upper Range	Unit of Measure				
1. <u>Settling Pourl Studge</u>	<u> 40%</u>	<b></b>	50%					
3.								
4								
5		[						
2. ESTIMATED QUANTITY OF WASTE AND SHIPPING INFOR	MAILON							
a, KO One Time Event La Base La Repeat Event								
D. Estimated Armual quantity: 1000 - 1005 A cubic rards - Drums - Gallons - Uther (specify):								
c. Shipping Frequency:	Inits per 🔟 Month		Year <b>X</b> One Time	U Other				
d. Is this a U.S. Department of Transportation (USDO	T) Hazardous Material	? (If yes, answer e.)	) 🗆 Yes 🙀 No					
e. USDOT Shipping Description (if applicable):				]				
3. SAFETY REQUIREMENTS (Handling, PPE, etc.): N A								

©2008 Waste Management, Inc.

1 1

May 2008

WASTE MANAGEMENT	Status (Please check approp	riate responses)	e Promie Sheet	100951NM
1 Is this a USEPA (AO (	(FR Part 261) (State bazardous waste? T	Eves contact your cales ran		
<ol> <li>Is this waste include</li> </ol>	ed in one or more of categories below (C	heck all that apply? If yes	attach supporting documer	tation Dives Du
Delisted Hazardov	us Waste	Excluded Wastes Under	40 CFR 261.4 A	15 event
Treated Hazardou	us Waste Debris	Treated Characteristic	Hazardous Waste Da	THE Chedenial
3. Is the waste from a l	Federal (40 CFR 300, Appendix B) or sta	te mandated clean-up? If ye	es, see instructions. Ante	I where the Yes D
4. Does the waste repre	esented by this waste profile sheet cont	ain radioactive material?	nance	Yes X
a. If yes, is disposal	regulated by the Nuclear Regulatory Co	mmission?	C) Yes C) M	io C
b. If yes, is disposal	regulated by a State Agency for radioac	tive waste/NORM?	U Yes U N	
<ol> <li>Does the waste representation</li> </ol>	esented by this waste profile sheet cont	ain concentrations of regula	ted Polychlonnated Bipheny	Ls (PCBs)? Li Yes yall
6. Does the waste contr	ain untreated, regulated, medical or inf	ectious waste?		Ves N
<ol> <li>Does the waste cont</li> </ol>	tain asbestos? 🖸 Yes 🕱 No		If Yes,	G Friable O Non Friab
8. Is this profile for r	remediation waste from a facility that	is a major source of Haza	rdous Air Pollutants (Site F	Remediation NESHAP.
40 CFR 63 subna	art GGGGG)?	улт тал же ет тами		Ves No
If ves does t	he waste contain <500 nnmw VOHAP	s at the point of determina	ition?	
E. Generator C	ertification (Please read and	a certify by signatur	e below)	
By signing this Generati	or's Waste Profile Sheet, I hereby certify	/ that all:		
1. Information submitt	ted in this profile and all attached docu	ments contain true and accu	rate descriptions of the was	te material;
2. Relevant information	n within the possession of the Generato	r regarding known or suspec	ted hazards pertaining to th	is waste has been
disclosed to WM/the	e Contractor;			
3. Analytical data attac	ched pertaining to the profiled waste w	as derived from testing a rep	presentative sample in accord	lance with
40 CFR 261.20(c) or	equivalent rules; and			
4. Changes that occur	in the character of the waste (i.e. chan	ges in the process or new an	alytical) will be identified by	the Generator
and disclosed to WM	M (and the Contractor if applicable) prio	r to providing the waste to	NM (and the Contractor if ap	plicable).
5. Check all that apply	<i>n</i> :			Decholow
Attached analyti	ical pertains to the waste. Identify labo	ratory & sample ID #'s and p	arameters tested: RCI	<u> </u>
Hall Environmental	RANalysis LAboratory-ID=2A	1-1 through #2A2-20; 1	CLP-BALLO Meta	19 # Pages:
Only the analyse	es identified on the attachment pertain	to the waste (identify by lai	poratory & sample ID #'s and	parameters tested).
Attachment #:	·			
Additional infon	mation necessary to characterize the pr	ofiled waste has been attach	ed (other than analytical).	1.0.0.4.00
Indicate the nur	mber of attached pages:	Mass Frigash,	save river materia	als Grong - Cpg2
Lam an agent s	agning on behalf of the Generator, and t	the delegation of authority t	o me from the Generator for	this signature is
available upon r	request,	not a listed weeks and is to	In all TCID and the second	_
By Generator pro	ocess knowledge, the following waste is	not a listed waste and is be	low all ICLP regulatory limit	s.
Certification Signature	e: Child Aut abo	Title: Environ	mental Coperin	aron
Company Name: Weste	ern Kenning - Bloom field /K	Hunen Name (Print):	Cindy Hurtad	<u>о</u>
Date:		······		
		FOR WIN USE ONLY		
Management Method:	Eandfill U Bioremediation	Approval Dec	sion: Approved	Not Approved
Q Non-hazardous soli	idification 🗅 Other:	Waste Approva	l Expiration Date:	2105/09
Management Facility	Precautions, Special Handling Proc	edures or Limitation (	Chall not contain free lig	uid
on approval:	dian for all		C Shipment must be sched	uled into disposal facility
	privice see attach	e4 i	Approval Number must a	company each shipment
	_	<b>A</b> 5	3 Waste Manifest must are	omnany load
WM Authorization Na	ame / Title: Stan E /	hdom	Date	: _1/05/08
State Authorization 4	(if Required):		D-+-	
	In neglenced/			•

.

**j** 1



DATE:

**EXPIRATION DATE:** 

02/05/09

GENERATOR:	Western Refining	Southwest,	Inc	Bloomfield	Refinery

-A

## Precautions, Conditions, and Limitations on Approval

100951NM

### **Generator Requirements**

WM WASTE CODE/PROFILE#

- ✓ Waste Management must be informed if the waste characteristics or process change. Changes to be approved on a case-by-case basis.
- ✓ Generator or its authorized agent has certified that no hazardous waste codes, PCBs, asbestos containing materials, or other prohibited wastes are associated with this waste stream.
- ✓ Waste must be absent free liquids.
- This waste is a New Mexico Special Waste (industrial solid waste). The following additional requirements apply:
  - A manifest must accompany all loads of special waste in accordance with NMAC 20.9.8.19.
  - All drums or containers must be clearly labeled or marked indicating the name and address of the Generator, contents, potential health, safety, and environmental hazards associated with the waste (NMAC 20.9.8.10(d)).
- ✓ Contact Waste Management to schedule waste for disposal at least 24 hours prior to shipping.
  - San Juan Landfill: 505.334.1121

## Waste Management Handling Conditions

Direct bury. Material must be covered immediately. Cover with refuse or soil prior to compaction. Do not generate dust.

$\checkmark$	Specific health and safety concerns associated with this material: Avoid all contact.
	Follow general H&S procedures including good hygiene if come in contact
	with waste material.
~	Required PPE is indicated below:
	Respirator and filters/cartridges:
	Chemical resistant gloves:
	Standard Work Uniform
	Tyvek Suit
	⊠ Full sideshield safety glasses, or
	Chemical goggles;
	Other PPE depending on conditions of management and disposal

Ensure eyewash station is functional.

### Anderson, Stacy-E

From: Sent: To: Subject:

81.13

- 1

Matzke, Michelle Tuesday, November 04, 2008 7:58 AM Anderson, Stacy-E 100951NM - Western Refining

-) **4**7008





Fly-Ash SRMG - Fly Ash.pdf SRMG - Fly Ash -Analytical.pdf p5.pdf

----Original Message----From: Tessier, Kevin M [mailto:tessier.kevin@cleanharbors.com] Sent: Tuesday, November 04, 2008 6:04 AM To: Callaghan, Dan Cc: Matzke, Michelle Subject: RE: Western Refinery Sludge

Attached is the MSDS and Analytical for the bottom ash solidification material.

- A sample of the fly-ash and pond 2 liquid (representative of the impending materials) was hand-delivered this morning by our onsite crew.

1. See attached MSDS. Yes it is exempt. Yes, make change.

2. Settling pond from refinery process

3. Required cleaning due to build-up of solids and mud

Let me know if we've got enough here for approval. If so, any way you could get this approved as quickly as possible would be great. Customer is very anxious for approval.

Thanks! Kevin

----Original Message----From: Callaghan, Dan [mailto:DCallagh@wm.com] Sent: Monday, November 03, 2008 5:22 PM To: Tessier, Kevin M Cc: Matzke, Michelle Subject: Western Refinery Sludge

Kevin,

My approvals manager has sent back a few questions regarding Western Refinery's settling pond sludge. Could you please respond at your earliest convenience?

1) Can you tell me where the flyash came from? Is the flyash exempt under RCRA (see 40 CFR 261.4(b)(4))? The exemption reads as follows:
"The following solid wastes are not hazardous wastes.... Fly ash waste, bottom ash waste, slag waste, and flue gas emission control waste, generated primarily from the combustion of coal or other fossil fuels, except as provided by 266.112 of this chapter for facilities that burn or process hazardous waste."
If exempt, please email authorization to make this change on the profile (Section D.2)
If not exempt, do they have metals analytical for the ash itself?

2. Where did the material in the settling pond come from?

3. Why cleaning out the settling pond?

Thank you Kevin, please let me know if you have any questions.

Sincerely,

1

1

Dan Callaghan Industrial Account Manager Waste Management 7780 E 96th Avenue Henderson, CO 80640 303-886-9694

27 9 4 2000

The information transmitted is intended only for the person or entity to which it is addressed and may contain confidential and/or privileged material. Any review, retransmission, dissemination or other use of, or taking of any action in reliance upon, this information by persons or entities other than the intended recipient is prohibited. If you received this in error, please contact Clean Harbors Environmental Services at 781.792.5555 and delete the material from any computer.

## ATTACHMENT C

ø

1

,

# North and South Aeration Lagoon Closure Certification Report (Replacement Page)

shipped to HEAL for analysis of hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste. Analytical results indicate that the sludge did not exhibit any hazardous characteristics. The sludge load from the Northwest ABT unit was removed via vacuum truck, mixed with fly ash, transported and disposed of at the San Juan County Landfill in Aztec, New Mexico.

On October 21, 2008, seven sludge samples were collected in a grid pattern from the Northeast ABT unit (#3 Aeration Lagoon). The sludge samples were placed on ice for preservation and shipped to HEAL for analysis of hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste. Analytical results indicate that the sludge did not exhibit any hazardous characteristics. The sludge load from the Northeast ABT unit was removed via vacuum truck, mixed with fly ash, transported and disposed of at the San Juan County Landfill in Aztec, New Mexico.

The sludge sample location map is presented in Attachment 5. Laboratory reports are provided in Attachment 6.

#### 4.2 Site Inspection and Repair

1

1

After sludge removal, the entire top liner was power washed with water. The liner/residue wash water was collected via vacuum truck and off-loaded into the API Separator. The South ABT unit was physically inspected in November 2008 by an independent engineer licensed in the State of New Mexico during closure activities. A crack in the plastic weld was discovered at the crossover piping between #1 and #2 Aeration Lagoon. During the investigation process, personnel discovered damage to the boot on the second liner and serious pitting and corrosion on the piping that discharges from the South ABT unit into the Northwest ABT unit. Both sections of discharge piping from the South ABT unit into the Northwest and Northeast ABT units were replaced. After the piping was replaced, the liner was repaired inside the South ABT unit and inspected again. There was no accumulation of fluids in the underlying collection system beneath the 60-ml liner to indicate damage to the secondary liner. The South ABT unit was put back in service by November 18, 2008.

The Northwest ABT unit's top liner was wash power washed with water after sludge removal. The liner/residue wash water was collected via vacuum truck and off-loaded into the API Separator. The liner surrounding the new inlet pipe from the South ABT Unit was repaired. Inspection of the liner showed small scrapes and gouges that did not penetrate the liner. A

## ATTACHMENT D

;

1

# IDW Management Plan

.

## **IDW Management Plan**

All IDW will be properly characterized and disposed of in accordance with all federal, State, and local rules and regulations for storage, labeling, handling, transport, and disposal of waste. The IDW may be characterized for disposal based on the known or suspected contaminants potentially present in the waste. It is assumed that there are no listed wastes present in environmental media at any of the planned investigation areas.

A dedicated decontamination area will be setup prior to any sample collection activities. The decontamination pad will be constructed so as to capture and contain all decontamination fluids (e.g., wash water and rinse water) and foreign materials washed off the sampling equipment. The fluids will be pumped directly into suitable storage containers (e.g., labeled 55-gallon drums), which will be located at satellite accumulation areas until the fluids are disposed in the refinery wastewater treatment system upstream of the API separator. The solids captured in the decontamination pad will be shoveled into 55-gallon drums and stored at the designated satellite accumulation for off-site disposal.

Drill cuttings generated during installation of soil borings and monitoring wells will be placed directly into 55-gallon drums and staged in the satellite accumulation area pending results of the waste characterization sampling. The portion of soil cores, which are not retained for analytical testing, will be placed into the same 55-gallon drums used to store the associated drill cuttings.

The solids (e.g., drill cuttings and used soil cores) will be characterized by testing to determine if there are any hazardous characteristics in accordance with 40 Code of Federal Regulations (CFR) Part 261. This includes tests for ignitability, corrosivity, reactivity, and toxicity. If the materials are not characteristically hazardous, then further testing will be performed pursuant to the requirements of the facility to which the materials will be transported. Depending upon the results of analyses for individual investigation soil samples, additional analyses may TPH and polynuclear aromatic hydrocarbons.

Purge water generated during groundwater sampling activities will be containerized in 55-gallons drums and then disposed in the refinery wastewater treatment system upstream of the API separator. All miscellaneous waste materials (e.g., discarded gloves, packing materials, etc.) will be placed into the refinery's solid waste storage containers for off-site disposal.

# ATTACHMENT E

# IDW Soil Cuttings Analytical

•

.



## COVER LETTER

Tuesday, July 28, 2009

Kelly Robinson Western Refining Southwest, Inc. #50 CR 4990 Bloomfield, NM 87413

TEL: (602) 908-6617 FAX (505) 632-3911

RE: RCRA Investigation-Group 1

Dear Kelly Robinson:

Order No.: 0907367

Hall Environmental Analysis Laboratory, Inc. received 1 sample(s) on 7/21/2009 for the analyses presented in the following report.

These were analyzed according to EPA procedures or equivalent. Below is a list of our accreditations. To access our accredited tests please go to www.hallenvironmental.com or the state specific web sites.

Reporting limits are determined by EPA methodology. No determination of compounds below these (denoted by the ND or < sign) has been made.

Please don't hesitate to contact HEAL for any additional information or clarifications.

Sincerely,

Andy Freeman, Business Manager Nancy McDuffie, Laboratory Manager

NM Lab # NM9425 AZ license # AZ0682 ORELAP Lab # NM100001 Texas Lab# T104704424-08-TX



4901 Hawkins NE ■ Suite D ■ Albuquerque, NM 87109 505.345.3975 ■ Fax 505.345.4107 www.hallenvironmental.com

CLIENT:	Western Refining Sout	hwest, Inc.		Clien	t Sample ID:	IM No. 1-	- IDW
Lab Order:	0907367	0907367			llection Date:	7/20/2009	) 11:10:00 AM
Project:	RCRA Investigation-G	roup 1		Da	ate Received:	7/21/2009	)
Lab ID:	0907367-01			•	Matrix:	SOIL	
Analyses		Result	PQL	Qual	Units	DF	Date Analyzed
EPA METHOD 8	021B: VOLATILES						Analyst: NSB
Methyl tert-butyl	ether (MTBE)	ND	0.10		mg/Kg	1	7/22/2009 5:19:58 PM
Benzene		ND	0.050		mg/Kg	1	7/22/2009 5:19:58 PM
Toluene		ND	0.050		mg/Kg	1	7/22/2009 5:19:58 PM
Ethylbenzene		ND	0.050		mg/Kg	1	7/22/2009 5:19:58 PM
Xylenes, Total		ND	0.10		mg/Kg	1	7/22/2009 5:19:58 PM
Surr: 4-Bromo	fluorobenzene	89.0	66.8-139		%REC	· <b>1</b>	7/22/2009 5:19:58 PM
EPA METHOD 8	310: PAHS						Analyst: JMP
Naphthalene		ND	0.25		mg/Kg	1	7/27/2009 8:44:14 PM
1-Methylnaphthal	lene	ND	0.25		mg/Kg	1	7/27/2009 8:44:14 PM
2-Methylnaphthal	lene	ND	0.25		mg/Kg	1	7/27/2009 8:44:14 PM
Acenaphthylene		ND	0.25		mg/Kg	1	7/27/2009 8:44:14 PM
Acenaphthene		ND	0.25		mg/Kg	1	7/27/2009 8:44:14 PM
Fluorene		ND	0.030		mg/Kg	1	7/27/2009 8:44:14 PM
Phenanthrene		ND	0.015		mg/Kg	1	7/27/2009 8:44:14 PM
Anthracene		ND	0.015		mg/Kg	1	7/27/2009 8:44:14 PM
Fluoranthene		ND	0.020		mg/Kg	1	7/27/2009 8:44:14 PM
Pyrene		ND	0.025		mg/Kg	1	7/27/2009 8:44:14 PM
Benz(a)anthracer	ne	ND.	0.010		mg/Kg	1	7/27/2009 8:44:14 PM
Chrysene		ND	0.011		mg/Kg	1	7/27/2009 8:44:14 PM
Benzo(b)fluoranth	nene	ND	0.010		mg/Kg	1	7/27/2009 8:44:14 PM
Benzo(k)fluoranth	lene	ND	0.010		mg/Kg	1	7/27/2009 8:44:14 PM
Benzo(a)pyrene		ND	0.010		mg/Kg	1	7/27/2009 8:44:14 PM
Dibenz(a,h)anthra	аселе	ND	0.010		mg/Kg	1	7/27/2009 8:44:14 PM
Benzo(g,h,i)peryle	ene	ND	0.010		mg/Kg	1	7/27/2009 8:44:14 PM
Indeno(1,2,3-cd)p	yrene	ND	0.10		mg/Kg	1	7/27/2009 8:44:14 PM
Surr: Benzo(e)	pyrene	106	31.5-75.9	S	%REC	1	7/27/2009 8:44:14 PM
MERCURY, TOLI	P						Analyst: SNV
Mercury		ND	0.020		mg/L	1	7/24/2009 2:11:47 PM
EPA METHOD 60	010B: TCLP METALS						Analyst: TES
Arsenic		ND	5.0	1	mg/L	1	7/24/2009 11:25:52 AM
Barium		ND	100	I	mg/L	1	7/24/2009 11:25:52 AM
Cadmium		ND	1.0	I	mg/L	1	7/24/2009 11:25:52 AM
Chromium		ND	5.0		mg/L	1	7/24/2009 11:25:52 AM
Lead		ND	5.0	1	mg/L	1	7/24/2009 11:25:52 AM
Selenium		ND	1.0	I	mg/L	1	7/24/2009 11:25:52 AM
Silver		ND	5.0	. 1	mg/L	1	7/24/2009 11:25:52 AM

# Hall Environmental Analysis Laboratory, Inc.

8

Date: 28-Jul-09

B Analyte detected in the associated Method Blank

H Holding times for preparation or analysis exceeded

MCL Maximum Contaminant Level

RL Reporting Limit

Page 1 of 1

Qualifiers: *

- E Estimated value
- J Analyte detected below quantitation limits
- ND Not Detected at the Reporting Limit
- S Spike recovery outside accepted recovery limits
- .
- Value exceeds Maximum Contaminant Level B Analyte



ENERGY LABORATORIES, INC. * 1120 S 27th St * PO Box 30916 * Billings, MT 59107-0916 Toll Free 800.735.4489 * 406.252.6325 * FAX 406.252.6069 * ell@energylab.com

#### LABORATORY ANALYTICAL REPORT

 Client:
 Hall Environmental

 Project:
 0907367

 Lab ID:
 B09071973-001

 Client Sample ID:
 0907367-01C, IM No. 1-IDW

 Report Date:
 07/24/09

 Collection Date:
 07/20/09 11:10

 DateReceived:
 07/22/09

 Matrix:
 Soil

Analyses	Result	Units	Qualifiers	RL	MCL/ QCL	Method	Analysis Date / By
IGNITABILITY Flash Point (Ignitability)	>200	۰F		30.0	• .	SW1010M	07/23/09 15:00 / clr
CORROSIVITY pH of Soil and Waste	7.97	s.u.		0.10		SW9045D	07/23/09 00:00 / cir
REACTIVITY Cyanide, Reactive Sulfide, Reactive	ND ND	mg/kg mg/kg		0.05 20	250 500	SW846 Ch 7 SW846 Ch 7	07/23/09 10:31 / kjp 07/22/09 13:00 / pwc

Report Definitions: RL - Analyte reporting limit. QCL - Quality control limit. MCL - Maximum contaminant level. ND - Not detected at the reporting limit.


ENERGY LABORATORIES, INC. * P.O. Box 30916 + 1120 South 27th Street (Billings, MT 59107-0916 Toll Free 800.735.4489 * 406.252.6325 * Fax 406.252.6069 * ell@energylab.com * www.energylab.com

### **QA/QC Summary Report**

Client: Hall Environmental

Project: 0907367

Report Date: 07/24/09 Work Order: B09071973

Analyte	Result Units	RL %	%REC Low Limit: High Limit RPD	RPDLimit Qual
Method: SW846 Ch 7		••••••••••••••••••••••••••••••••••••••	<b>,</b>	Batch: 40262
Sample ID: MB-40262 Cyanide, Reactive	Method Blank ND mg/kg	0.05	Bun: AUTOAN201-B_090723A	07/28/09 10:32
Method; SW846 Ch 7		<u></u>		Batch: R133149
Sample ID: MB-R133143 .Sullide, Reactive	Method Blank ND mo/kg	10	Run: MISC-H2W_090722A	07/22/09 13:00
Sample ID: LCS-R133143 Sullide, Reactive	Laboratory Control Sample 22 mg/kg	20	Run: MISC-HZW_090722A 76 50 150	07/22/09 13:00
Sample ID: 509071471-003CDUP Sulfide, Reactive	Sample Duplicate 40 mg/kg	20	Ruh: MISC-HZW_090722A 40	07/22/09 13:00 20 R
Method SW9045D	a <u></u>		<u> </u>	Batch: R133202
Sample ID: B09071972-001ADUP pH of Soll and Waste	Sample Dupilcate 8,28 s.u.	0.10	Run: PH METER_090723A	07/23/09·00:00 10

Qualifiers:

RL - Analyte reporting limit.

R - RPD exceeds advisory limit.

ND - Not detected at the reporting limit.

1.079

3.243

mġ/Kg

mg/Kg

0.050

0.10

108

108

69.3

73

125

128

## QA/QC SUMMARY REPORT

Client:

Western Refining Southwest, Inc.

**RCRA** Investigation-Group 1 **Project:** Work Order: 0907367 PQL %RPD Analyte Result Units %Rec LowLimit HighLimit RPDLimit Qual EPA Method 8021B: Volatiles Method: Sample ID: 0907367-01A MSD MSD Batch ID: 19668 Analysis Date: 7/22/2009 7:21:35 PM Methyl tert-butyl ether (MTBE) 0.9971 mg/Kg 0.10 99.7 67.9 135 2.78 28 0.9790 97.2 132 Benzene mg/Kg 0.050 78.8 0.338 27 Toluene 0.9994 0.050 78.9 112 2.50 19 mg/Kg 96.4 Ethylbenzene 1.039 0.050 104 69.3 125 3.77 10 mg/Kg Xylenes, Total 3.065 mg/Kg 0.10 102 73 128 5.66 13 Sample ID: MB-19668 MBLK Batch ID: 19668 Analysis Date: 7/22/2009 8:22:31 PM Methyl tert-butyl ether (MTBE) ND mg/Kg 0.10 Benzene ND mg/Kg 0.050 Toluene NÐ 0.050 mg/Kg Ethylbenzene ND mg/Kg 0.050 Xvienes, Total ND mg/Kg 0.10 Sample ID: LCS-19668 LCS Batch ID: 19868 Analysis Date: 7/22/2009 7:51:58 PM Methyl tert-butyl ether (MTBE) 1.009 67.9 mg/Kg 0.10 100 135 Benzene 0.9934 mg/Kg 0.050 97.4 78.8 132 Toluene 1.025 mg/Kg 0.050 101 78.9 112 Ethylbenzene 1.070 mg/Kg 0.050 107 69.3 125 Xylenes, Total 3.165 0.10 73 mg/Kg 106 128 Sample ID: 0907367-01A MS MS Batch ID: 19668 Analysis Date: 7/22/2009 6:51:20 PM Methyl tert-butyl ether (MTBE) 0.9698 mg/Kg 0.10 97.0 67.9 135 Benzene 0.9757 mg/Kg 0.050 96.9 78.8 132 Toluene 1.025 mg/Kg 0.050 98.9 78.9 112

Qualifiers:

Ethylbenzene

Xylenes, Total

E Estimated value

J Analyte detected below quantitation limits

R RPD outside accepted recovery limits

H Holding times for preparation or analysis exceeded

ND Not Detected at the Reporting Limit

S Spike recovery outside accepted recovery limits

Page 1

### QA/QC SUMMARY REPORT

Client:

Western Refining Southwest, Inc.

**RCRA** Investigation-Group 1 **Project:** Work Order: 0907367 %RPD Result Units PQL %Rec LowLimit HighLimit RPDLimit Qual Analyte Method: EPA Method 8310: PAHs Sample ID: MB-19682 MBLK Batch ID: 19682 Analysis Date: 7/27/2009 6:22:57 PM Naphthalene ND mg/Kg 0.25 1-Methvinaphthalene ND ma/Ka 0.25 ND mg/Kg 0.25 2-Methylnaphthalene Acenaphthylene ND mg/Kg 0.25 Acenaphthene ND mg/Kg 0,25 0.030 ND mg/Kg Fluorene Phenanthrene ND mg/Kg 0.015 ND 0.015 Anthracene mg/Kg Fluoranthene ND mg/Kg 0.020 Pyrene ND mg/Kg 0.025 ND mg/Kg 0.010 Benz(a)anthracene ND 0.011 Chrysene mg/Kg ND 0.010 Benzo(b)fluoranthene mg/Kg Benzo(k)fluoranthene ND mg/Kg 0.010 Benzo(a)pyrene ND mg/Kg 0.010 Dibenz(a,h)anthracene ND mg/Kg 0.010 ND mg/Kg 0.010 Benzo(g,h,i)perylene ND mg/Kg 0.10 Indeno(1,2,3-cd)pyrene Batch ID: LCS 19682 Analysis Date: 7/27/2009 6:43:09 PM Sample ID: LCS-19682 Naphthalene 1.328 mg/Kg 0.25 66.4 35.4 86.1 1.401 0.25 70.0 38.4 90.1 1-Methylnaphthalene mg/Kg 2-Methylnaphthalene 1.336 mg/Kg 0.25 66.8 36.2 91.9 39.6 88.1 Acenaphthylene 1.505 mg/Kg 0.25 75.3 Acenaphthene 1.419 mg/Kg 0.25 70.9 38.8 91.6 Fluorene 0.1160 mg/Kg 0.030 58.0 19.9 102 0.015 26.2 103 0.06675 66.4 Phenanthrene mg/Kg 0.07200 mg/Kg 0.015 31 95.3 Anthracene 71.6 0.1535 0.020 37.2 90.5 Fluoranthene mg/Kg 76.5 Pyrene 0.1243 mg/Kg 0.025 62.1 29.2 92.4 34.6 97.5 Benz(a)anthracene 0.01425 mg/Kg 0.010 71.3 0.07125 0.011 35.6 94.3 Chrysene mg/Kg 70.8 29.9 97.4 Benzo(b)fluoranthene 0.02075 mg/Kg 0.010 83.0 Benzo(k)fluorantherie ND mg/Kg 0.010 78.0 36.9 95.7 Benzo(a)pyrene ND mg/Kg 0.010 70.0 35.3 97 Dibenz(a,h)anthracene 0.01875 mg/Kg 0.010 72.0 37.7 90.7 Benzo(g,h,i)perviene 0.01775 mg/Kg 0.010 61.0 35.1 94.1 Indeno(1,2,3-cd)pyrene ND mg/Kg 0.10 80.5 34.6 89.2 LCSD Batch ID: Sample ID: LCSD-19682 19682 Analysis Date: 7/27/2009 7:03:21 PM Naphthalene 1.239 mg/Kg 0.25 62.0 35.4 86.1 6.90 26.2 1-Methylnaphthalene 1.310 mg/Kg 0.25 65.5 38.4 90.1 6.70 23.5 2-Methylnaphthalene 1.300 mg/Kg 0.25 36.2 91.9 2.71 22.7 65.0 Acenaphthylene 1.453 mg/Kg 0.25 39.6 88.1 18.8 72.6 3.55 Acenaphthene 1.380 mg/Kg 0.25 69.0 38.8 91.6 2.77 19 Fluorene 0.1120 mg/Kg 0.030 56.0 19.9 102 3.51 21.4

Qualifiers:

E Estimated value

J Analyte detected below quantitation limits

R RPD outside accepted recovery limits

H Holding times for preparation or analysis exceeded

Not Detected at the Reporting Limit

S Spike recovery outside accepted recovery limits

Page 2

ND

## QA/QC SUMMARY REPORT

Client:

Western Refining Southwest, Inc. RCRA Investigation-Group 1

**Project:** Work Order: 0907367 PQL %RPD Units %Rec LowLimit HighLimit RPDLimit Qual Analyte Result EPA Method 8310: PAHs Method: Sample ID: LCSD-19682 Batch ID: 19682 Analysis Date: LCSD 7/27/2009 7:03:21 PM mg/Kg 26.2 103 0.374 31.7 0.06700 0.015 66.6 Phenanthrene 95.3 0.347 18.3 71.8 31 Anthracene 0.07225 mg/Kg 0.015 0.020 79.0 37.2 90.5 3.21 23.8 Fluoranthene 0.1585 mg/Kg 63.0 29.2 92.4 1.40 18.9 Pyrene 0.1260 mg/Kg 0.025 Benz(a)anthracene 0.01400 mg/Kg 0.010 70.0 34.6 97.5 1.77 40 Chrysene 0.06950 mg/Kg 0.011 69.1 35.6 94.3 2.49 33 2.44 Benzo(b)fluoranthene 0.02025 mg/Kg 0.010 81.0 29.9 97.4 38.2 Benzo(k)fluoranthene ND mg/Kg 0.010 76.0 36.9 95.7 0 26.2 ND mg/Kg 0.010 68.0 35.3 97 0 35.5 Benzo(a)pyrene 90.7 2.70 mg/Kg 0.010 70.0 37.7 25.1 Dibenz(a,h)anthracene 0.01825 Benzo(g,h,i)perylene 0.01725 mg/Kg 0.010 59.0 35.1 94.1 2.86 20.5 ND mg/Kg 0.10 78.5 34.6 89.2 0 23.1 Indeno(1,2,3-cd)pyrene Method: MERCURY, TCLP MBLK Batch ID: 19696 Analysis Date: Sample ID: MB-19696 7/24/2009 2:04:46 PM Mercury ND mg/L 0.020 Analysis Date: LCS Batch ID: 19696 7/24/2009 2:06:31 PM Sample ID: LCS-19696 Mercury ND mg/L 0.020 102 80 120 EPA Method 6010B: TCLP Metals Method: Sample ID: MB-19690 MBLK Batch ID: 19690 Analysis Date: 7/24/2009 11:11:59 AM ND mg/L 5.0 Arsenic Barium ND mg/L 100 Cadmium ND mg/L 1.0 ND 5.0 Chromium mg/L ND 5.0 Lead mg/L ND mg/L 1.0 Selenium ND mg/L 5.0 Silver Sample ID: LCS-19690 LCS Batch ID: 19690 Analysis Date: 7/24/2009 11:14:35 AM 5.0 Arsenic ND mg/L 114 80 120 Barium ND mg/L 100 98.9 80 120 Cadmium ND mg/L 1.0 107 80 120 Chromium ND mg/L 5.0 100 80 120 Lead ND mg/L 5.0 99.1 80 120 Selenium ND mg/L 1.0 112 80 120 Silver ND mg/L 5.0 103 80 120

Qualifiers:

E Estimated value

J Analyte detected below quantitation limits

R RPD outside accepted recovery limits

H Holding times for preparation or analysis exceeded

ND Not Detected at the Reporting Limit

S Spike recovery outside accepted recovery limits

Page 3

mail Environmental Analysis Laboratory, il		•.		
Sar	nple Receipt C	hecklist	۰.	
Client Name WESTERN REFINING SOUT		Date Receiv	ed:	7/21/2009
Work Order Number 0907367		Received t	y: ARS	
Checklist completed by:		Sample ID	labels checked by:	Initials
Matrix: Carrier na	ame: <u>UPS</u>			
Shipping container/cooler in good condition?	Yes 🗹	No 🗌	Not Present	
Custody seals intact on shipping container/cooler?	Yes 🗹	- No 🗌	Not Present	Not Shipped
Custody seals intact on sample bottles?	Yes 🗌	No 🗌	N/A 🗹	
Chain of custody present?	Yes 🗹	No 🗔		
Chain of custody signed when relinquished and received?	Yes 🗹	No 🗌		. •
Chain of custody agrees with sample labels?	Yes 🗹	No 🗌		
Samples in proper container/bottle?	Yes 🗹	No 🗌		•
Sample containers intact?	Yes 🗹	No		
Sufficient sample volume for indicated test?	Yes 🗹	No 🗌		
All samples received within holding time?	Yes 🗹	No 🗔		Number of preserved
Water - VOA vials have zero headspace? No VOA vials	submitted 🗹	Yes 🗌	No 🗌	bottles checked for pH:
Water - Preservation labels on bottle and cap match?	Yes	No 🗌	N/A 🗹	
Water - pH acceptable upon receipt?	Yes 🗍	No 🗍	N/A 🗹	<2 >12 unless noted
Container/Temp Blank temperature?	<b>4.0</b> °	<6° C Accepta	ble nt time to cool.	Deiow.
COMMENTS:				
Client contacted Date contacted:	<u></u>	Per	son contacted	
Contacted by: Regarding:				
·		····		
Comments:				
Corrective Action				

-

all Environmental Analysis Laboratory, Inc.

7

CCC   LCC   LCC     HALLENVIRONMENTAL   ANALYSIS LABORATORY     ANALYSIS LABORATORY     www.hallenvironmental.com     4901 Hawkins NE - Albuquerque, NM 87109     Tel. 505-345-3975     Fax 505-345-4107-	BTEX + MTBE + TMB's (8021) BTEX + MTBE + TPH (Gas only) TPH Method 8015B (Gas/Diesel) TPH (Method 504.1) EDB (Method 504.1) B310 (PNA or PAH) Anions (F,CI,NO ₃ ,NO ₂ ,PO ₄ ,SO ₄ ) B310 (PNA or PAH) Anions (F,CI,NO ₃ ,NO ₂ ,PO ₄ ,SO ₄ ) B260B (VOA) Anions (F,CI,NO ₃ ,NO ₂ ,PO ₄ ,SO ₄ ) B260B (VOA) B260B (VOA) Anions (F,CI,NO ₃ ,NO ₂ ,PO ₄ ,SO ₄ ) B260B (VOA) B260B (VOA) Anions (F,CI,NO ₃ ,NO ₂ ,PO ₄ ,SO ₄ ) B270 (Semi-VOA) B260B (VOA) B260B (VOA)	V V X X X X X X X X X X X X X X X X X X
Client: Der Lostody Record Turn-Around Time: Client: Der Losteux Berline, Butwert lie Standard Rush 5 day Bloomfield Read 4996 RCBA Investigation - Group 1 Bloomfield NNA 87413 Project #:	email or Fax#: Kellu, Rebinson Cunr.com Project Manager:   QA/QC Package: QA/QC Package: Project Manager:   QA/QC Package: Date Date   Matrix Level 4 (Full Validation) Kullu, Robinson   Date Time Matrix   Date Time Matrix   Date Time Matrix	120/4 IID Soil IM No. 1-IDU (3) Jaw Now II Parent Providence III Parent Providence III Parent



OOMFIELD REFINERY



-,0

September 1, 2009

James Bearzi, Bureau Chief New Mexico Environmental Department Hazardous Waste Bureau 2905 Rodeo Park Drive East, Building 1 Santa Fe, New Mexico 87505-6303

Re: Giant Refining Company, Bloomfield Refinery (currently know as Western Refinery Southwest, Inc.. – Bloomfield Refinery) Order No. HWB 07-34 (CO) Solid Waste Management Unit (SWMU) Group No. 1 North and South Actation Lagoons Closure Report.

Dear Mr. Bearzi:

Western Refining Southwest Inc. - Bloomfield Refinery submits the referenced Closure Report pursuant to Section IV.B.7 of the July 2007 HWB Order. The Closure Report summarizes the closure activities of SWMU Group No. 1, North and South Aeration Lagoons. The Closure Report was developed and formatted to meet the requirements of Section X.C of the July 2007 HWB Order.

If you have any questions or would like to discuss the Investigation Work Plan, please contact me at (505) 632-4171.

Sincerely, Vames R. Schmaltz

Environmental Manager Western Refining Southwest, Inc. Bloomfield Refinery

cc: Hope Monzeglio – NMED HWB Carl Chavez – NMOCD (w/attachment) Dave Cobrain – NMED HWB Laurie King – EPA Region 6 (w/attachment) Todd Doyle – Bloomfield Refinery Allen Hains – Western Refining El Paso



404 Camp Craft Rd., Austin, Texas 78746 Tel: (512) 347 7588 Fax: (512) 347 8243 Internet: www.rpsgroup.com/energy

## North and South Aeration Lagoons Closure Certification Report

Western Refining Southwest, Inc. Bloomfield Refinery Bloomfield, New Mexico

September 2009

Prepared by:

Western Refining Southwest, Inc. Bloomfield Refinery 50 Road 4990 Bloomfield, NM 87413

> RPS 404 Camp Craft Road Austin, Texas 78746

## **Table of Contents**

## **List of Sections**

Section 1	Introduction	1
Section 2	Background	2
Section 3	Closure Objectives	4
Section 4 4.1 4.2 4.3	Closure Procedures Sludge Characterization, Removal, and Disposal Site Inspection and Repair Collection of Soil and Flush Water Samples	5 
Section 5 5.1 5.2 5.3 5.4 5.5 5.6 5.7	Ground Water Monitoring System Soil Boring Drilling and Soil Sampling Activities Monitoring Well Installation Monitoring Well Completions Survey Soil Sampling Analytical Results Ground Water Sampling Investigation Derived Waste Management	9 9 12 14 15 15 15 16 17
Section 6	Conclusions	19



## List of Attachments

- Attachment 1 Closure Plan and NMED Approval with Modifications Letter
- Attachment 2 Photographs of Closure (Before)
- Attachment 3 Photographs of Closure (After)
- Attachment 4 Liner Repair Photos
- Attachment 5 Sludge Sample Locations
- Attachment 6 Chain-of-Custoday Forms and CD of Laboratory Analytical Reports
- Attachment 7 Soil Boring and Monitoring Well Location Map
- Attachment 8 Soil Boring Logs and Well Construction Diagrams
- Attachment 9 Field Monitoring Parameter Results
- Attachment 10 Well Survey Information
- Attachment 11 Soil and Ground Water Analytical Summaries
- Attachment 12 Quality Assurance/Quality Control Report

This is to certify that closure of regulated unit EPA ID #NMD089416416 has been completed and that closure has been completed in accordance with and in compliance with good engineering practices and the provisions of the NMED Enforceable Document dated July 27, 2007.

I certify under penalty of law that this document was prepared under my direction or supervision in accordance with a system designed to assure that qualified personnel properly gather and evaluate the information submitted. Based on my inquiry of the person or persons who manage the system, or those persons directly responsible for gathering the information, the information submitted is, to the best of my knowledge and belief, true, accurate, and complete. I am aware that there are significant penalties for submitting false information, including the possibility of fine and imprisonment for knowing violations.



Tode Doyle, Plant Manager

8-27-09

Date

8/26/09

· · ·

## Section 1 Introduction

The Bloomfield Refinery is located in northwestern New Mexico, approximately 1 mile south of the City of Bloomfield in San Juan County. It is further located approximately 1/2 mile east of State Route 44 on County Road 4990 (a.k.a. Sullivan Road).

The refinery is a crude oil refinery that incorporates various processing units that convert crude into finished products. The facility can receive and process up to 18,000 barrels per day of crude oil and produce propane, butane, gasoline, naphtha, kerosene, diesel fuel, fuel oil, and residual fuel.

The essential function of the North and South Aeration Lagoons is aggressive biological treatment (ABT) of process wastewater. Monitoring data of the effluent from the API Separator, which discharges into the South Aeration Lagoon, has indicated that concentrations of benzene above the toxicity characteristic regulatory threshold of 0.5 milligrams per liter (mg/L) have entered the aeration lagoons. Bloomfield Refinery does not want to operate these lagoons as hazardous waste treatment units. In that regard, pretreatment in the form of benzene strippers and a 10,000 barrel surge tank has been installed to ensure that wastewater with hazardous levels of benzene does not enter the aeration lagoons in the future. After the installation of the pretreatment equipment, the lagoons were cleaned out to remove all hazardous waste, hazardous constituents, decomposition products, and leachate.

This report addresses all activities related to the modified closure of the aeration lagoons. Modified closure is defined as the process where each lagoon is removed from service, the existing water and sludge is removed and the liner is cleaned, inspected, and, if necessary, repaired before being returned for service.

This report provides the results of the closure activities conducted for the aeration lagoons, which will bring Bloomfield Refinery into compliance with both the requirements of the EPA CAFO from May 18, 2006 and the requirements of the NMED Order dated July 27, 2007.

## Section 2 Background

Process wastewater is generated at various refinery processing units, storage tanks, utility systems, and maintenance activities. This water is collected in a segregated sewer system located throughout the refinery processing and tankage areas. Process wastewater flows to the API Separator where solids, sludge, and floating scum are removed. API Separator effluent is then pumped through the Benzene Strippers and then flows onward through a series of three lined aeration lagoons. Wastewater is then either evaporated at the evaporation ponds or injected underground at the Class I injection well.

Three federal regulatory programs [the Clean Water Act, the Resource Conservation and Recovery Act (RCRA), and the Safe Drinking Water Act (SWDA)] have affected the evolution of the North and South Aeration Lagoons to their current operation and status.

In 1974, the impoundments were constructed with bentonite-treated bottoms for fresh water holding. After the initiation of the Clean Water Act (40CFR Part 419), the ponds were converted to manage API Separator water as a secondary biological treatment of the wastewater. In 1982, RCRA regulations triggered the first clean out of these biological treatment oily water ponds. In 1982/83 a liner and leachate system was installed that consisted of a 33% bentonite composite liner equipped with a French drain system, with a 100-ml high density polyethylene (HDPE) liner on top. Around 1990, two key RCRA regulatory changes (the listing of F037/F038 and the adoption of the D018 benzene TCLP standards) lead Bloomfield Refinery to submit a Part B RCRA permit application. To comply with RCRA interim status, the lagoons were upgraded and retrofitted with an additional set of double liners and leak detection/leachate collection system over and above the cleaned 1982/83 system. In 2007, a benzene stripper/tank system was constructed and put into service to decharacterize all wastewater prior to entering the first aeration lagoon.

The Aeration Lagoons from top to bottom, include a 100-mil HDPE top liner, a geonet for collecting leaks to a sump equipped with a 6" observation pipe, a 60-ml HDPE secondary liner, a composite geotextile/geonet with a 4" observation pipe, a cement amended sand layer that was compacted into a 1.5% slope, a 100-ml HDPE liner, a French drain system which directs any collected fluids to a central sump, and a 6" layer of soil with 33% bentonite mixed into it.

2

The South Lagoon averages 4.4 feet in depth and has a surface area of about 6,652 square feet. The total volume is approximately 216,000 gallons. At a wastewater flow rate of 80 gpm, the holding time in the pond is 1.9 days. This lagoon is equipped with two, 5 horsepower aspirating aerators sized to prevent F037/F038 waste generation.

The Northwest Lagoon averages 5.5 feet in depth with a surface area of 10,000 square feet. This lagoon is equipped with two 2-horsepower aerators and wastewater retention time (at 80 gpm) is 3.6 days. The Northeast Lagoon averages 5.7 feet in depth, with a surface area of 8,440 square feet and a volume of approximately 360,000 gallons. This lagoon is equipped with two 2-horsepower aerators and wastewater retention time (at 80 gpm) of 3.1 days.

## Section 3 Closure Objectives

The objective is to bring the Bloomfield Refinery into compliance with both the requirements of the EPA CAFO from May 18, 2006 and the requirements of the NMED Order dated July 27, 2007. The applicable closure standard for the North and South Aeration Lagoons is 40 CFR 264.111. This regulation states:

The owner or operator must close the facility in a manner that:

- (a) Minimizes the need for further maintenance; and
- (b) Controls, minimizes, or eliminates, to the extent necessary to protect human health and the environment, post-closure escape of hazardous waste, hazardous constituents, leachate, contaminated runoff, or hazardous waste decomposition products to the ground or surface waters or to the atmosphere.

The closure procedures that are specified for meeting these objectives are outlined in the NMED-approved Closure Plan for the North and South Aeration Lagoons, and included installation of four additional monitoring wells as requested by NMED in the Approved with Modification letter dated August 7, 2008. The Closure Plan and NMED correspondence letter are included as Attachment 1 to this report.



The modified closure procedures for the North and South Aeration Lagoons are described in the NMED-approved Closure Plan dated May 2008. Generally, the modified closure plan is defined as the process by which each aeration lagoon is removed from service, the existing water and sludge is removed and appropriately disposed of, the liner is cleaned, inspected, and if necessary, repaired before being returned to service. In accordance with guidance from NMED, the South ABT unit was taken out of service in October 2008, to perform the modified closure process, and the unit became operational again on November 18, 2008. At that time, the Northwest ABT unit was bypassed and modified closure procedures were initiated. This unit was restored to service on December 29, 2008 after completion of modified closure at which time the Northeast ABT unit was taken out of service. Clean out activities of the Northeast ABT unit was taken out of service. Service and and infinite the unit were completed January 20, 2009. Piping replacement and liner repair was completed by February 4, 2009 at which time the unit was put back into service.

Photographs documenting the closure procedures are provided in Attachments 2 and 3 with those in Attachment 2 reflecting the unit before closure and those in Attachment 3 reflecting the unit after closure.

#### 4.1 Sludge Characterization, Removal, and Disposal

In accordance with the modified Closure Plan, sixteen sludge samples were collected in a grid pattern from the South ABT unit (#1 Aeration Lagoon) on October 7, 2008. The sludge samples were placed on ice for preservation and shipped to Hall Environmental Analysis Laboratory (HEAL) for analysis of hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste. Five of the first seven sludge samples analyzed exhibited hazardous waste characteristics for benzene, therefore no further testing was conducted and the entire sludge load in the South ABT unit was removed from the unit via vacuum transport truck, then directly shipped offsite as oil-bearing hazardous secondary materials to be recycled (fuels blending). There was no speculative accumulation of this material. The receiving facility was Motiva Enterprises LLC - Norco Refinery, Norco, LA. EPA ID #LAD008186579.

On October 9, 2008, twenty sludge samples were collected in a grid pattern from the Northwest ABT unit (#2 Aeration Lagoon). The sludge samples were placed on ice for preservation and

shipped to HEAL for analysis of hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste. Analytical results indicate that the sludge did not exhibit any hazardous characteristics. The sludge load from the Northwest ABT unit was removed via vacuum truck, mixed with fly ash, transported and disposed of at the San Juan County Landfill in Aztec, New Mexico.

On October 21, 2008, eight sludge samples were collected in a grid pattern from the Northeast ABT unit (#3 Aeration Lagoon). The sludge samples were placed on ice for preservation and shipped to HEAL for analysis of hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste. Analytical results indicate that the sludge did not exhibit any hazardous characteristics. The sludge load from the Northeast ABT unit was removed via vacuum truck, mixed with fly ash, transported and disposed of at the San Juan County Landfill in Aztec, New Mexico.

The sludge sample location map is presented in Attachment 5. Laboratory reports are provided in Attachment 6.

#### 4.2 Site Inspection and Repair

After sludge removal, the entire top liner was power washed with water. The liner/residue wash water was collected via vacuum truck and off-loaded into the API Separator. The South ABT unit was physically inspected in November 2008 by an independent engineer licensed in the State of New Mexico during closure activities. A crack in the plastic weld was discovered at the crossover piping between #1 and #2 Aeration Lagoon. During the investigation process, personnel discovered damage to the boot on the second liner and serious pitting and corrosion on the piping that discharges from the South ABT unit into the Northwest ABT unit. Both sections of discharge piping from the South ABT unit into the Northwest and Northeast ABT units were replaced. After the piping was replaced, the liner was repaired inside the South ABT unit and inspected again. There was no accumulation of fluids in the underlying collection system beneath the 60-ml liner to indicate damage to the secondary liner. The South ABT unit was put back in service by November 18, 2008.

The Northwest ABT unit's top liner was wash power washed with water after sludge removal. The liner/residue wash water was collected via vacuum truck and off-loaded into the API Separator. The liner surrounding the new inlet pipe from the South ABT Unit was repaired. Inspection of the liner showed small scrapes and gouges that did not penetrate the liner. A

6

reinforcing plastic weld bead was applied to those areas before water was put back in the Northwest ABT unit on December 29, 2009.

After sludge removal, the entire top liner of the Northeast ABT unit was power washed with water. The liner/residue wash water was collected via vacuum truck and off-loaded into the API Separator. Clean out activities were completed January 20, 2009. Inspection revealed a puncture on the north wall of the liner and a slice on the top east side of the liner. The discharge pipe from the Northeast ABT unit to the suction of P-616 (Transfer Pump from the Aeration Lagoon to the Evaporation Ponds) was found to be pitted and corroded and was replaced. Liner repair around that new discharge piping as well as around the new piping from the South ABT unit, and the two tears was completed February 4, 2009 and the unit was put back into service.

Liner repair surrounding the piping replacement consisted of cutting out and removing all three liners and the geonet. The liners and geonet were replaced with new material and new boots were created to tie into the piping.

Photographs documenting piping replacement and liner repair can be found in Attachment 4.

#### 4.3 Collection of Soil and Flush Water Samples

In keeping with the modified Closure Plan, after repairs to the liner of the South ABT unit were completed, the geonet between the top 100-ml liner and the second 60-ml liner was flushed with clean water. The flush water was sampled using a bailer in the 6" observation pipe, placed on ice for preservation, and shipped to HEAL for analysis of hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste. Analytical results indicate that the flush water did not exhibit any hazardous characteristics. The flush water was removed from underneath the liner via vacuum truck through the 6" observation pipe and off-loaded at the API Separator.

Inspection after cleaning the Northeast ABT unit revealed a puncture on the north wall of the liner and a slice on the top east side of the liner and squishiness under the liner in the northeast section. This discovery prompted sampling and analysis of the water between the top 100-ml liner and the second 60-ml liner. The liner water was sampled through a new incision in the liner, placed on ice for preservation, and shipped to HEAL for analysis of hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste. Analytical results indicate that the liner water did not exhibit any hazardous

characteristics. The liner water was removed from underneath the liner via vacuum truck through two new incisions in the liner, and off-loaded at the API Separator. There was no accumulation of fluids in the underlying collection system beneath the 60-ml liner to indicate damage to the secondary liner. The incisions were repaired during the other repair of the liner.

Stained soil was discovered underneath the discharge piping from the South ABT unit to the Northwest ABT unit when the piping was replaced. The soil was removed, placed on containment, and sampled. The characterization samples were placed on ice for preservation, and shipped to HEAL for analysis of TCLP Metals (EPA Method 6010B) and Benzene (EPA Method 8260B). Analytical results indicate non-detect on all analyses. The soil was used as backfill.

Stained soil was found under the discharge piping from the Northeast ABT unit to the suction of P-616. The soil was removed, placed on containment, and sampled. The characterization samples were placed on ice for preservation, and shipped to HEAL for analysis of TCLP Metals (EPA Method 6010B) and Benzene (EPA Method 8260B). Analytical results indicate non-detect on all analyses. The soil was used as backfill.

Laboratory reports are provided in Attachment 6.

## Section 5 Ground Water Monitoring System

As requested by NMED in a comment letter dated August 7, 2008, four soil borings (IM-1-1, IM-1-2, IM-1-3, and IM-1-4) were drilled around the outer perimeter of the North and South Aeration Lagoons. Soil cuttings from the soil borings were field screened using a PID, and soil samples were collected for analytical analysis. Each boring was completed as a permanent monitoring well (MW-55, MW-56, MW-57, and MW-58, respectively). A figure showing the location of the wells with respect to nearby site features is included in Attachment 7.

#### 5.1 Soil Boring Drilling and Soil Sampling Activities

Drilling activities were conducted between March 31st and April 3th, 2009. The soil boring locations were initially hydro-excavated to approximately 7 ft below ground surface (bgs) to identify potential utility lines near the proposed drilling locations. The soil borings were drilled using hollow-stem auguring (HSA) method or air rotary-ODEX method.

Soil samples were collected using split-spoon samplers. Soil samples were collected continuously and logged by a qualified geologist. The soil sample descriptions were made in accordance with USGS nomenclature and recorded on the individual field boring logs. As seen on the boring logs (Attachment 8), the data recorded included the lithologic interval, symbol, percent recovery, and a sample description of the cuttings and core samples.

Samples obtained from the borings were screened in the field on 2 foot intervals for evidence of contaminants. Field screening results were recorded on the soil boring logs. Field screening results were used to aid in the selection of soil samples for laboratory analysis. The primary screening methods include: (1) visual examination, (2) olfactory examination, and (3) headspace vapor screening for volatile organic compounds.

Visual screening included examining the soil samples for evidence of staining caused by petroleum-related compounds or other substances that may have caused staining of natural soils such as elemental sulfur or cyanide compounds. Headspace vapor screening was conducted and involved placing a soil sample in a plastic sealable bag allowing space for ambient air. The bag was sealed, labeled and then shaken gently to expose the soil to the air trapped in the container. The sealed bag was allowed to rest for a minimum of 5 minutes while the vapors equilibrated. Vapors present within the sample bag's headspace were then measured by inserting the probe of

9

a MiniRae 2000 in a small opening in the bag. The maximum value and the ambient air temperature were recorded on the field boring log for each sample. The screening results are presented in Attachment 9. The MiniRae 2000 was calibrated to 100 ppm isobutylene each day to the manufacturer's standard for instrument operation. Field screening results and any conditions that were considered to be capable of influencing the results of the field screening were recorded on the field logs.

#### Soil Boring: IM-1-3

On March 31, 2009, drilling commenced at soil boring location IM-1-3, which was located near the northeast corner of South Aeration Lagoon. Soil samples were collected using HSA drilling method and split-spoon samplers until soil appeared saturated. The drill rig was then modified to drill using the ODEX drilling method to extend the soil boring down to the Nacimiento Formation surface.

Soils encountered between 0 to 6 ft bgs appeared to be fill material and exhibited an odor. Elevated PID readings were observed in the soils with readings of 237 ppm (2 -4 ft bgs) and 567 ppm (4 – 6 ft bgs). A soil sample from the 4 -6 ft interval was collected for laboratory analysis. The soils exhibited an odor from 7 to 15 feet bgs. Elevated PID readings were observed in the soils from 7 to 13 feet bgs with the highest reading occurring at the 7 to 9 foot interval (415 ppm). The soils (gravelly sand) at 17 feet bgs appeared to be saturated. A soil sample was collected for laboratory analysis from the interval 15 to 17 bgs.

The rig was modified to use the ODEX drilling method and resumed drilling and sampling activities. Gravelly sand was encountered from 17 to 23 ft bgs. The sand was dark gray, saturated and exhibited a strong hydrocarbon odor. The Nacimiento was encountered at 23 ft bgs and consisted of high plastic, very stiff, damp, light yellowish orange clay. The borehole was sampled to a depth of 24 ft bgs. An elevated PID reading of 856 ppm was collected from the 17 to 19 foot interval. A soil sample was not collected since the sample was saturated. The borehole was deepened to a final depth of 24.25 feet bgs to accommodate the well setting.

#### Soil Boring: IM-1-2

On April 1, 2009, drilling commenced at soil boring IM-1-2, which was located near the northeast corner of the North Aeration Lagoon. Soil samples were collected using HSA drilling method and split-spoon samplers until soil appeared saturated. The drill rig was then modified

to drill using the ODEX drilling method to extend the soil boring down to the Nacimiento Formation surface.

The soils encountered from 0 to 6 feet bgs consisted of fill material. Elevated PID readings were observed in the soils with readings of 116 ppm (4 - 6 ft) and 40 ppm (6 - 8 ft). A soil sample from the 4 -6 foot interval was collected for laboratory analysis. The soils from 8 to 16 ft bgs did not exhibit an odor nor were there any elevated PID readings from 8 to 16 feet bgs. The PID reading from 16 to 18 ft bgs was recorded as 50 ppm and a hydrocarbon odor was noted. The soils (gravelly sand) at 18 feet bgs appeared to be saturated. A soil sample for laboratory analysis was collected from the interval 16 to 18 bgs.

The rig was modified to use the ODEX drilling method and resumed drilling and sampling. Gravelly sand was encountered from 18 to 20 feet bgs. The sand was black, saturated and exhibited a hydrocarbon odor. The Nacimiento was encountered at 20 feet bgs and consisted of a weathered sandstone/sand that was damp to moist, gray to brown with a faint odor. The borehole was sampled to a depth of 22 ft bgs and was deepened to 23.25 ft bgs to accommodate the well setting.

#### Soil Boring: IM-1-1

On April 2, 2009 drilling commenced at soil boring IM-1-1, which was located near the northwest corner of the North Aeration Lagoon. Soil samples were collected using HSA drilling method and split-spoon samplers until soil appeared saturated. The drill rig was then modified to drill using the ODEX drilling method to extend the soil boring down to the Nacimiento Formation surface.

Fill material was encountered from 0 to 12 ft bgs. Gravelly sand was encountered from 12 to 22 ft bgs. The fill material and gravelly sand from 4 to 22 ft bgs did exhibit a hydrocarbon odor and elevated PID readings. Soil samples for laboratory analysis were collected from the following intervals:

- 8 10 ft bgs Fill material (sand), dark brown, hydrocarbon odor, 891 ppm
- 18 20 ft bgs Gravelly sand, black, hydrocarbon odor, 1085 ppm
- 20 22 ft bgs Gravelly sand, black, hydrocarbon odor, 1047 ppm.

The augers were removed from the borehole and the rig was modified to use the ODEX drilling method. Drilling and sampling resumed with gravelly sand encountered from 22 to 25.75 ft bgs. The sand was black, saturated/oily and exhibited a hydrocarbon odor. The Nacimiento was

encountered at 25.75 ft bgs and consisted of a weathered sandstone/sand that was damp, greenish gray and exhibited a faint odor. The borehole was sampled to a depth of 26 ft bgs and then deepened to 27.25 ft bgs to accommodate the well setting.

#### Soil Boring: IM-1-4

On April 3, 2009, drilling commenced at soil boring location IM-1-4, located near the northwest corner of the South Aeration Lagoon. Soil samples were collected using HSA drilling method and split-spoon samplers until soil appeared saturated. The drill rig was then modified to drill using the ODEX drilling method to extend the soil boring down to the Nacimiento Formation surface.

Fill material (clay) was encountered from 0 to 10 ft bgs and from 2 to 10 ft bgs did exhibit a hydrocarbon odor and elevated PID readings. A soil sample for laboratory analysis was collected from the 2 - 4 foot interval (395 ppm).

The augers were removed from the borehole and the rig was modified to use the ODEX drilling method. Drilling and sampling resumed with gravelly sand encountered from 10 to 19.5 ft bgs. The gravelly sand exhibited a hydrocarbon odor; however, the PID readings were decreasing in intensity with depth from 43 ppm (10-12 ft bgs) to 17 ppm (14 -16 ft bgs). At the interval 18 to 19.5 ft bgs, the PID reading increased to 312 ppm in a damp to moist, gravelly sand. A soil sample from this interval was collected for laboratory analysis.

The lithology of the soil changed at 19.5 ft bgs and became a medium grained, moist to very moist sand that was black and exhibited a strong hydrocarbon odor. The sand was considered to be weathered Nacimiento and extended to a depth of at 22.5 ft bgs. There was no recovery of the core sample from 22.5 to 25 ft bgs. The core sample collected from the interval 25 to 27 feet bgs was clay with high plasticity, damp to moist in sand seams and was olive brown. The drilling and sampling terminated in this clay at 27 ft bgs.

#### 5.2 Monitoring Well Installation

The four soil borings were completed as permanent monitoring wells, extending down to the top of bedrock (Nacimiento Formation). Slotted (0.01 inch) rigid PVC well screen was placed at the bottom of the well and extended for 10 to 15 feet to ensure that the entire saturated zone is open to the well. Rigid PVC with threads was utilized for the well casing and no glues/solvents

were utilized. A 10/20 sand filter pack was installed to a minimum of two feet over the top of the well screen. A six-inch sand bed was also installed at the base of the monitor well.

Since the top of the well screens in this area were near the surface, a minimum of three feet of bentonite seal was placed over the filter pack and hydrated. The remaining annular space in the wells ranged from 2 to 6 feet bgl and was filled with concrete during the installation of the pad. Well construction diagrams are located in Attachment 8 of this report.

#### MW-57 (IM-1-3)

Following completion of the soil boring drilling at IM-1-3 on April 1, 2009, the soil boring was converted to a permanent monitoring well. Slotted (0.01 inch) rigid PVC well screen was placed at the bottom of the well and extended for 15 feet (8 to 23 ft bgs) to ensure that the entire saturated zone was open to the well. Rigid Schedule 40 PVC with threads was utilized for the well casing. A six-inch sand bed was placed at the bottom of the well bore. The 10/20 sand filter pack was installed approximately two feet over the top of the well screen. As the sand was installed in the well bore the outer casing of the ODEX drilling system was removed. Three feet of bentonite was placed over the filter pack and hydrated. After allowing the bentonite to hydrate the surface pad and protective casing were installed.

#### MW-56 (IM-1-2)

Following completion of the soil boring drilling at IM-1-2 on April 1, 2009, slotted (0.01 inch) rigid PVC well screen was placed at the bottom of the well and extended for 15 ft bgs (7 to 22 ft bgs) to ensure that the entire saturated zone was open to the well. Rigid Schedule 40 PVC with threads was utilized for the well casing. A six-inch sand bed was placed at the bottom of the well bore. The 10/20 sand filter pack was installed approximately two feet over the top of the well screen. As the sand was installed in the well bore, the outer casing of the ODEX drilling system was removed. Three feet of bentonite was placed over the filter pack and hydrated. After allowing the bentonite to hydrate the surface pad and protective casing were installed.

#### <u>MW-55 (IM-1-1)</u>

Following completion of the soil boring drilling at IM-1-1 on April 2, 2009, slotted (0.01 inch) rigid PVC well screen was placed at the bottom of the well and extended for 15 ft (11 to 26 ft bgs) to ensure that the entire saturated zone was open to the well. Rigid Schedule 40 PVC with

threads was utilized for the well casing. A six-inch sand bed was placed at the bottom of the well bore. The 10/20 sand filter pack was installed approximately two feet over the top of the well screen. As the sand was installed in the well bore the outer casing of the ODEX drilling system was removed. Three feet of bentonite was placed over the filter pack and hydrated. After allowing the bentonite to hydrate a flush mount surface completion was installed.

#### <u>MW-58 (IM-1-4)</u>

Following the completion of soil boring drilling at IM-1-4 on April 3, 2009, slotted (0.01 inch) rigid PVC well screen was placed at the bottom of the well and extended for 15 ft bgs (10.75 to 25.75 ft bgs) to ensure that the entire saturated zone was open to the well. Rigid Schedule 40 PVC with threads was utilized for the well casing. A six-inch sand bed was placed at the bottom of the well bore. The 10/20 sand filter pack was installed approximately 2.75 ft over the top of the well screen. As the sand was installed in the well bore the outer casing of the ODEX drilling system was removed. Three feet of bentonite was placed over the filter pack and hydrated. After allowing the bentonite to hydrate a flush mount surface completion was installed.

#### 5.3 Monitoring Well Completions

The surface completions used consisted of either flush mount completions or stickup completions. The flush mount completions consisted of an 8-inch well vault centered within a concrete pad measuring 4 ft by 4 ft wide by 6-inches thick. The concrete pad was wire reinforced.

The stickup completions consisted of a protective aluminum enclosure with cap that was secured in a concrete pad measuring 4 ft by 4 ft wide by 6-inches thick. The concrete pad was wire reinforced. The aluminum protective casing extends approximately 4 ft above the top surface of the concrete pad.

Four-inch diameter steel bollards were installed approximately six-inches from each corner of the concrete pad to protect the stickup wells. The bollards were installed two feet below grade and extend approximately three feet above grade. The holes for the bollards were dug by hand with the diameter of the borehole measured a minimum of 6-inches. Each bollard was cemented into the ground with the cement extending from the bottom of the hole to the surface. The bollard was filled with cement. Each bollard was pretreated to remove rust and painted with two coats of safety-yellow paint.

#### 5.4 Survey

Upon completion of the monitoring well installation activities, each monitoring well was surveyed by a registered New Mexico professional land surveyor in accordance with the State Plane Coordinate System (NMSA 1978 47-1-49-56). Horizontal position of each monitoring well was measured to the nearest 0.1-ft, and vertical elevations were measured to the nearest 0.01-ft. In addition, a hand held GPS receiver was used to record the coordinates of each soil boring. These coordinates were recorded on the boring logs. A copy of the well survey information is included in Attachment 10.

#### 5.5 Soil Sampling Analytical Results

Soil samples were analyzed by Hall Environmental Analysis Laboratory in Albuquerque, New Mexico using the following methods for organic constituents.

- SW-846 Method 8260 volatile organic compounds; and
- SW-846 Method 8015B gasoline, diesel, and motor oil range petroleum hydrocarbons.

For each sample that reported a diesel concentration greater than 2,000 mg/kg, the sample was analyzed for the following:

• SW-846 Method 8270 semi-volatile organic compounds.

In addition, soil samples were analyzed for the following metals using the indicated analytical methods.

Analyte	Analytical Method
Arsenic	SW-846 Method 6010/6020
Barium	SW-846 Method 6010/6020
Cadmium	SW-846 Method 6010/6020
Chromium	SW-846 Method 6010/6020
Lead	SW-846 Method 6010/6020
Mercury	SW-846 Method 7470/7471
Selenium	SW-846 Method 6010/6020
Silver	SW-846 Method 6010/6020

A summary of the soil analytical results is included in Attachment 11. Chain-of-custody copies and pdf copies of the analytical data reports are included in Attachment 6. There were no conditions observed during the sample collection efforts that are thought to have had any impact on the analytical results. The respective regulatory cleanup levels as outlined in the Order are included in the table to facilitate a comparison between the reported concentrations and the applicable cleanup levels. Concentrations that exceed the applicable cleanup levels are bolded.

#### 5.6 Ground Water Sampling

During the week of May 4th, 2009, ground water samples were collected from the four permanent monitoring wells around the Aeration Lagoons. The ground water samples were analyzed for organic constituents by the following methods:

- SW-846 Method 8260 volatile organic compounds;
- SW-846 Method 8270 semi-volatile organic compounds; and
- SW-846 Method 8015B gasoline, diesel, and motor oil range petroleum hydrocarbons.

Ground water samples were also analyzed for the following metals and general chemistry analytes using the indicated analytical methods:

Analyte	Analytical Method
Arsenic	SW-846 Method 6010/6020
Barium	SW-846 Method 6010/6020
Cadmium	SW-846 Method 6010/6020
Chromium	SW-846 Method 6010/6020
Iron	SW-846 Method 7470/7471
Lead	SW-846 Method 6010/6020
Magnesium	SW-846 Method 7470/7471
Manganese	SW-846 Method 7470/7471
Mercury	SW-846 Method 7470/7471
Potassium	SW-846 Method 7470/7471
Selenium	SW-846 Method 6010/6020
Silver	SW-846 Method 6010/6020
Sodium	SW-846 Method 7470/7471
Total Dissolved Solids (TDS)	SM 2540C
Chloride	EPA Method 300.0
Fluoride	EPA Method 300.0
Nitrogen, Nitrate (as N)	EPA Method 300.0
Nitrogen, Nitrite (as N)	EPA Method 300.0
Phosphorus, Orthophosphate (as P)	EPA Method 300.0
Sulfate	EPA Method 300.0
Alkalinity, total (as CaCO3)	SM Method 2320B
Bicarbonate	SM Method 2320B
Carbonate	SM Method 2320B
Specific Conductance	EPA Method 120.1



A summary table and a CD with lab reports of the ground water analytical results is provided in Attachment 11. Chain-of-custody copies are provided in Attachment 6. Soil and ground water sample results that pertain to the installation of the four permanent monitoring wells were subject to a Level II data review. A summary report of the Level II data review is included in Attachment 12 of this report.

Each well was purged a minimum of three well volumes prior to sample collection. Field measurements of ground water stabilization parameters included pH, specific conductance, dissolved oxygen concentrations, oxidation-reduction potential, and temperature were monitored during purging of each well. The calibration solutions used at the beginning of each day are as follows:

- 4.0 pH solution
- 7.0 pH solution
- 10.0 pH solution
- 1.413 mS/cm conductivity solution
- 220 for ORP

Following parameter stabilization and 3-volume well purging, ground water samples were immediately poured directly into clean laboratory supplied containers. A summary of the ground water field parameters is included in Attachment 9 of this report.

Prior to well purging, depth to product, depth to ground water, and soil gas samples were collected. The soil gas samples were field monitored for VOCs, O₂, and CO₂ using a MiniRae 2000 PID and multi-gas meter. Dedicated Teflon tubing was used that extended down into the well, approximately 1 ft above the detected fluid level. The top of the well was sealed off to prevent ambient air from entering the well casing during soil-gas monitoring. A portable vacuum pump was used to purge a minimum of three well volumes before the vapor sample was collected. A summary of the vapor sampling results are included in Attachment 9.

#### 5.7 Investigation Derived Waste Management

Drill cuttings, excess sample material decontamination fluids, and all other investigation derived waste (IDW) associated with drilling and sampling activities were contained and placed in 55-gallon drums.

Drilling equipment was decontaminated between each borehole using a high pressure potable water wash. The decon water collected in a mobile decon trailer and was subsequently placed in open top 55-gallon drums. Split-spoon samplers used to collect soil samples were

17

decontaminated between each use using a potable water rinse, an Alconox was and then a distilled water rinse. The decon water from the sampling equipment was collected in 5-gallon buckets and placed in open top 55-gallon drums. All IDW drums were sealed and labeled at the end of each work day.

Following completion of all drilling activities, all soil and sampling fluids were disposed of in compliance with the approved IDW Management Plan. A composite sample of the soil cuttings was collected and sent to Hall Environmental Analytical Laboratories for analysis. The soil cuttings were characterized as non-hazardous and disposed of off-site in compliance with the approved IDW management plan. All decontamination and purged ground water fluid was disposed of at the Refinery through the wastewater treatment system, upstream of the API Separator.

## Section 6 Conclusions

The North and South Aeration Lagoons, which are the subject of this closure certification report, were evaluated to determine compliance with applicable state and federal regulatory criteria that prescribe standards for the closure of hazardous waste management units and the approved Closure Plan. The results of this evaluation indicate that the closure of the unit conforms to these regulations and the approved Closure Plan and no additional reporting activities related to the service of this unit will be required.

## **Attachment 1**

## **Closure Plan and NMED Approval with Modifications Letter**

Title	Tab Number
Approved Closure Plan	1
NMED Approval with Modifications	2

## **Approved Closure Plan**



404 Camp Craft Rd., Austin, TX 78746 Tel: (512) 347 7588 Fax: (512) 347 8243 Internet: www.rpsgroup.com/energy

## Closure Plan North and South Aeration Lagoons Bloomfield Refinery

## Regulated Unit EPA ID# NMD089416416 HWB-GRCB-07-002

Giant Refining Company Bloomfield, New Mexico

May 2008

James R. Schmaltz Environmental Manager

out

Scott T. Crouch, P.G. Senior Consultant RPS JDC, Inc.

United Kingdom Australia USA Canada Ireland Netherlands Malaysia

## **Table of Contents**

### List of Sections

Section 1 Introduction	Ï
Section 2 Wastewater Treatment Unit Description and Operation	3
2.1 Environmental Regulatory Activities	3
2.2 ABT Unit Operations	5
Section 3 WWTU Upgrades	7
3.1 Contingency Plan	7
Section 4 ABT Unit Closure	8
4.1 Closure Procedures	8
Section 5 Construction Details	2

Introduction

The Bloomfield Refinery is located immediately south of Bloomfield, New Mexico in San Juan County. The physical location address is #50 Road 4990, Bloomfield, New Mexico 87413. The Bloomfield Refinery is located on 285 acres (0.45 square miles). The site is located on a bluff approximately 100 feet above the south side of the San Juan River, a perennial river that flows to the west.

Bordering the facility is a combination of federal and private properties. Public property managed by the Bureau of Land Management lies to the south. The majority of undeveloped land in the vicinity of the facility is used extensively for oil and gas production and, in some instances, grazing. The town of Bloomfield is located to the north of the refinery, across the San Juan River. U.S. Highway 44 is located approximately one-half mile west of the facility. The topography of the site is generally flat with low-lying areas to the east of the process area.

The Blöomfield Refinery is a crude oil refinery currently owned by the San Juan Refining Company and operated by Giant Industries Arizona, Inc., which is a wholly owned subsidiary of Western Refining Company. The Bloomfield Refinery generally processes crude oil from the Four Corners area transported to the facility by pipeline or tanker truck and crude from West Texas transported by pipeline.

The Bloomfield Refinery has an approximate refining capacity of 18,000 barrels per day. Various process units are operated at the facility, including crude distillation, reforming, fluidized catalytic cracking, sulfur recovery, merox treater, catalytic polymerization and diesel hydrotreating. Current and past operations have produced gasoline, diesel fuels, jet fuels, kerosene, propane, butane, naphtha, residual fuel, fuel oils and LPG.

This Closure Plan addresses the "closure" of the North and South Aeration Lagoons. Monitoring data of the effluent from the API Separator, which discharges into the South Aeration Lagoon, has indicated that concentrations of benzene above the toxicity characteristic (TC) regulatory threshold of 0.5 milligrams per liter (mg/l) have entered the aeration lagoons. Western Refining does not desire to operate these lagoons as hazardous waste treatment units and thus the ponds will be cleaned out to remove all hazardous waste, hazardous constituents, decomposition products, and leachate. Closure of the aeration lagoons will be conducted in accordance with an Enforceable Document (July 27,

1

2007 NMED Order). Additional pretreatment in the form of benzene strippers and a 10,000 barrel surge tank has been installed to ensure that wastewater with hazardous levels of benzene does not enter the aeration lagoons in the future.

# Wastewater Treatment Unit Description and Operation

#### 2.1 Environmental Regulatory Activities

2

All oil refineries produce process wastewater, which today must be managed in accordance with a variety of environmental requirements intended to assure adequate and appropriate protection of public health and the environment. Three federal regulatory programs [the Clean Water Act, the Resource Conservation and Recovery Act (RCRA), and the Safe Drinking Water Act (SWDA)] have major significance for Bloomfield Refinery process wastewater. Two of these federal programs at Bloomfield are directly administered by the State of New Mexico, as it has primacy over the RCRA and SDWA UIC programs. In addition, there are additional state regulatory programs with varying applicability, including those administered by New Mexico Oil Conservation Division (OCD).

Initially, beginning in 1972 under the Clean Water Act regulatory program, EPA promulgated petroleum refinery wastewater management requirements pursuant to the Clean Water Act NPDES permit program. The principal federal regulations implementing this CWA program as it applies to petroleum refineries are found at 40 C.F.R. Parts 122 and 419. The Bloomfield Refinery, like other oil refineries impacted by 40 C.F.R. Part 419, had implemented a series of process wastewater treatment operations, including primary treatment of wastewaters with an oil/water separator and secondary biological treatment in wastewater ponds to further reduce organics in the petroleum refinery wastewater. These two ponds where such biological degradation of organics occurred were referred to at the time as the North Oily Water Pond and the South Oily Water Pond.

A second major regulatory program, the RCRA regulations, affecting hazardous wastewaters was promulgated by EPA on November 19, 1980. Initially, these applied only to certain sludges created by petroleum refinery wastewater management, such as API oil/water separator sludge which was listed as K051 hazardous waste. In November, 1980, the Bloomfield Refinery operator applied for a Part A permit as a generator, and TSD as a protective filing for its so-called oily water ponds. It was later determined they were not disposing of listed hazardous waste on site and since D018 wastewater was not part of the 1980 EP toxicity test (it only became regulated after the 1990 TCLP toxicity test was adopted), in 1982 they petitioned for reclassification under a generator only status.¹ In



3.

¹ On November 26, 1985, the Bloomfield Refinery agreed to take an on-site landfill [where some of the materials from the 1982 impoundment cleanout had been placed] through RCRA closure: During 1989, these materials were removed and eventually determined by EPA delisting to be non-hazardous for offsite disposal. See, <u>Hazardous Waste Delisting</u> Petition, Petroleum Contaminated Soil, dated April 15, 1991 (*ERM-Rocky Mountain, Inc.*)
1982/1983, the liquids and sludge were removed from the oily water ponds and disposed of offsite. Impacted soils were also excavated and the ponds were lined. This activity included the placement of a 33% bentonite composite liner equipped with a French drain system, with a 100 mill high density polyethylene (HDPE) liner on top.

In 1990, a significant revision to these regulations made most petroleum refinery process wastewater into D018 benzene characteristic hazardous waste, leading the Bloomfield Refinery to submit a Part B RCRA permit application² in the mid-1990s and to operate its biological treatment impoundments pursuant to RCRA interim status as a regulated unit. To comply with RCRA interim status, the Bloomfield Refinery upgraded and retrofitted the regulated unit with an additional set of RCRA double liners and leak detection/leachate collection system over and above what the Bloomfield. Refinery had initially installed in 1982/1983.

The listing of F037/F038 sludges by EPA as hazardous (effective in 1991) effectively mandated a certain level of biological treatment and retention time in the biological treatment impoundments at the Bloomfield Refinery.³ Thereafter, the aeration-enhanced impoundments were called the North Aeration Lagoon (NAL) or the South Aeration Lagoon (NAL) [also referred to herein as the North Aggressive Biological Treatment (ABT) Units (two impoundments known as NABT-E and NABT-W) and the South ABT Unit]. The compliance strategy employed aggressive biological treatment of wastewaters to make them safe for the environment, followed by disposition through evaporation ponds and a Class I underground injection well permitted consistent with the Safe Drinking Water Act UIC program requirements.⁴ As discussed in Section 3.0, additional upgrades to the wastewater treatment system were recently completed in the fall of 2007.

⁴EPA promulgated regulatory requirements to assure that wastewater managed by UIC disposition not pose a risk to public health and the environment (40 C.F.R. Parts 144-146), but those did not apply at the Bloomfield Refinery until 1994 when Bloomfield installed a Class I UIC well for wastewater management.

²This Part B application submitted in the mid-1990s included a RCRA closure plan for the biological treatment impoundments, as discussed later in this document.

³Integral to the operation of the Bloomfield Refinery, as with any oil refinery in the United States, is the operation of an aggressive biological treatment (ABT) unit system for wastewater management, mandated by EPA regulations regarding the listing of certain petroleum refinery wastes (F037/F038) that became effective in May, 1991. EPA regulations, as adopted by NMED, effectively require each petroleum refinery to implement an ABT system to biological treat organics with regulatorily-specified ABT technology to remove organics and eliminate F037/F038 formation. The Bloomfield Refinery has had such advanced organic aeration in place as required since that time, and these EPA-required treatment systems operate as multi-lined ABT wastewater treatment units at Bloomfield, backed up with a double set of leak detection/leachate collection systems, over and above what has been technologically required under EPA regulations.

#### 2.2 ABT Unit Operations

The refinery process wastewater currently generated (approx. 80 gallons per minute (gpm)) at the Bloomfield Refinery is managed first by treatment in an API oil/water separator, then the volatile components are removed via benzene air strippers and the final treatment (biological) occurs in three ABT impoundments. The ABT units, from top to bottom, include:

- a 100-mil HDPE top liner;
- a geonet for collecting leaks that drain to a sump equipped with a 6" observation pipe;
- a 60-mil HDPE secondary liner;
- a composite geotextile/geonet with a 4" observation pipe;
- a cement amended sand that was compacted into a 1.5% slope;
- a 100-mil HDPE liner;
- a French drain system, which directs any collected fluids to a central sump; and
- a 6" layer of soil with 33% bentonite mixed into it.

The wastewater is currently discharged from the API separator, passes through the benzene air stripper and into the South ABT unit, which averages 4.4 feet in depth and has a surface area of about 6,652 square feet. The total volume is approximately 216,000 gallons. At 80 gpm, the holding time in the pond is 1.9 days. The impoundment is equipped with two, 5-horsepower aerators sized to prevent F037/F038 waste generation through high rate aeration. The system was designed to reduce benzene concentrations from approximately 10 ppm to less than 0.5 ppm. With the installation of the benzene stripper equipment in October 2007, the wastewater is now "decharacterized" below the benzene TC levels prior to discharge into this first (South) ABT unit. As a result, this unit has received its final volume of hazardous wastewater and no longer will be required to treat hazardous wastewater.

Wastewater from the first (South) ABT unit, which has already been reduced below TC levels by design, is routed to the North ABT unit through an overflow pipe from the South ABT unit. The second ABT unit is comprised of two impoundments that are operated together, and these are generally referred to as the North ABT unit. The first of the two impoundments in the North area (which can be referred to as North ABT-W as it is the westernmost of the two portions of the North ABT unit) is separated from the second of the two in the North area (the second can be referred to as North ABT-E) by a concrete divider. An overflow pipe from the North ABT-W connects to the North ABT-E. The outflow from North ABT-E goes to a sump, where the non-hazardous wastewater can be pumped for final disposition, either in evaporation ponds or into an SDWA Class I permitted non-hazardous UIC well.

The North ABT-W averages 5.5 feet in depth with a surface area of 10,000 square feet. The total volume is approximately 411,500 gallons. The North ABT-W unit is equipped with two (each) 2-horsepower aerators and wastewater retention time (at 80 gpm) is 3.6 days

The North ABT-E (the second of the two in the North area) averages 5.7 feet in depth, with a surface area of 8,440 square feet and a volume of approximately 360,000 gallons. The North ABT-E is equipped with two 2-horsepower aerators and wastewater retention time (at 80 gpm) is 3.1 days

The North and South ABT units have been operated with a minimum freeboard of two feet under normal operating conditions. At the lowest points during operation, the South ABT, North ABT-W and North ABT-E have freeboards of 2.97, 2.54 and 3.08 feet respectively. Influent flow into the South ABT unit is limited by the size of the overflow pipe coming from the API separator/wastewater treatment unit system. Operating personnel monitor pond water levels on a daily basis. The only non-controlled inflow is direct rainfall onto the North and South unit areas.

To manage precipitation, outflow from the ABT unit system is routed to a sump, which has an automatic level control pump. Excess water from process areas generated during a 100-year storm (2.6") is easily handled by this system. The impoundments have 698,000 gallons of additional capacity to the top of the freeboard and the pump can remove 720,000 gallons of water daily. This capacity management total greatly exceeds the 406,000 gallons of water that would be drained from 250,000 square feet of process area. The pump is backed up by two portable diesel backup pumps, which can function in the event of a power outage.

6

# 3 WWTU Upgrades

As a result of an EPA Consent Agreement and Final Order (CAFO) dated May 18, 2006, upgrades were made to wastewater treatment operations at the Blöomfield Refinery. This EPA-mandated change at the Bloomfield Refinery was accomplished through construction and operation of a benzene stripper/tank system that will decharacterize all hazardous process wastewater prior to further biological treatment in the ABT impoundments. The tank system is equipped with an additional 10,000 barrel tank to provide surge capacity. As a result, all process wastewater streams, including any contaminated runoff, will be decharacterized prior to discharge into the ABT units for aggressive biological treatment.

#### 3.1 Contingency Plan

In the event of a major failure, the first contingency response is to direct the wastewaters that have not been through the benzene stripping treatment process into the 10,000 barrel surge tank. At a rate of wastewater flow of 80 gpm, that would permit 87.5 hours of flow to be managed without discharge to the ABT units in the event of a benzene stripper failure. During those 87.5 hours for repair work, the benzene strippers in most cases could be fixed and returned to operation. In the event the surge tank capacity may be exceeded, it may be possible to make additional surge tank capacity available, depending on other tank usage at the Bloomfield Refinery. Such evaluation would occur if there was a significant likelihood the strippers could not be restored to working order within the 87.5 hour time frame available for repairs.

Once the benzene strippers are made operational again, wastewaters collected in the surge tank will be appropriately metered back through the wastewater treatment system by being introduced upstream of the API separator consistent with capacity available (in excess of the 80 gpm flow being handled). After the wastewater in the surge tank has been removed, the tank will be inspected to determine if any potentially F037 or F038 listed waste) is present, it will physically removed from the tank via the manway, and characterized and sent off-site for disposal in accordance with all applicable Hazardous Waste regulations.

7



This modified Closure Plan is submitted to bring the Bloomfield Refinery into compliance with both the requirements of the EPA CAFO from May 18, 2006 and the requirements of the NMED Order dated July 27, 2007 (also referred to as the Enforceable Document). Because the hazardous characteristic (D018 benzene) will be removed from the wastewater prior to placement in the ABT units (as the result of the wastewater treatment upgrades discussed in Section 3) no further treatment of hazardous waste will occur in the ABT units. Instead, the ABT units will continue to perform their essential function of biologically treating/aerating the non-hazardous wastewater. Such aggressive biological treatment of non-hazardous wastewater in the ABT units will be essential for the operation of the Bloomfield Refinery to assure that F037/F038 formation does not occur at Bloomfield and to achieve water quality required for wastewater disposition pursuant to the Class I UIC permit.

The applicable closure standard for the North and South Aeration Lagoons is 40 CFR §265.111 (Closure Performance Standard), which requires that the owner or operator must close the facility in a manner that:

(a) Minimizes the need for further maintenance, and

(b) Controls, minimizes or eliminates, to the extent necessary to protect human health and the environment, post-closure escape of hazardous waste, hazardous constituents, leachate, contaminated run-off, or hazardous waste decomposition products to the ground or surface waters or to the atmosphere.

#### 4.1 Closure Procedures

When the ABT impoundments became RCRA units as a result of the TC regulations, the Bloomfield Refinery became obligated to prepare and maintain a closure plan for the regulated unit. The previous closure plan for the ABT units was submitted on December 21, 1995 as a portion of the Part B RCRA permit application for this facility.

This modified closure plan coordinates retention of the environmental safeguards of the liners and leachate collection systems for the ABT units with corrective action that includes removal and appropriate disposition of all hazardous wastes, hazardous constituents, decomposition products, and leachate above that liner system while addressing any historic contamination (below the liners/leachate collection system) under the corrective action portion of the NMED Order (Enforceable Document) and any post-closure monitoring. In order to implement these requirements consistent with the guidance provided by NMED and EPA, the original 1995 closure plan for the ABT Units is modified as set forth below. "Modified closure" is defined as the process by which each aeration lagoon is removed from service, the existing water and sludge is removed and the liner is cleaned, inspected, and, if necessary, repaired before being return to service.

In accordance with guidance from NMED, the South ABT unit will be initially taken through this modified closure process, followed by the North ABT unit once the South ABT unit is placed back into service for the nonhazardous wastewaters coming from the upgraded wastewater treatment system. To accomplish closure of the South ABT Unit, nonhazardous wastewaters will flow directly from the API separator/benzene stripper system to the North ABT units, bypassing the South ABT unit.

After completion of the modified closure of the South ABT unit, the aggressive biological treatment system in the South ABT unit will become operational and the wastewater will be routed from the API separator/benzene stripper system back to the South ABT unit. Following appropriate aggressive biological treatment conducted in accordance with 40 CFR §261.31(b)(2)(i), the treated wastewaters will be then routed from the South ABT unit directly for disposition via evaporation and/or UIC-permitted injection, bypassing the North ABT. After completion of the modified closure for the North ABT unit, it will be restored to service as an additional wastewater treatment unit.

Each ABT unit will be decontaminated following the procedures discussed below. After the flow of decharacterized wastewater to an ABT unit is shut off as part of the closure process, the wastewater in the ABT unit will be pumped back to the WWTU to a location upstream of the API separator. The sludges (including some attendant watery solution entrained with the sludges) in the ABT unit above the liner will be sampled for hazardous characteristics in accordance with 40 CFR Part 261, Subpart C – Characteristics of Hazardous Waste. Sample(s) of the sludge will be collected for waste characterization at a minimum of one sample per each 10 cubic yards. If the sludges do not exhibit any hazardous characteristics, they will be removed from the ABT units by a vacuum truck for appropriate disposal. Additional wastes not amenable to vacuum removal may be removed through, careful shovel or other similar small-scale operations in such a manner as to assure protection of the 100 mil liner. The remaining materials [after vacuum and other removal operations have occurred] and the entire top liner will then be powerwashed with water. This nonhazardous washwater will be placed in the WWTU upstream of the API separator.

Revised (May 2008 - July typo fix) Bloomfield Aeration Basins Closure Plan

9

If wastes removed from the ABT units exhibit one or more hazardous characteristics, the wastes will be removed and placed into appropriate RCRA tanks/containers for disposal offsite as hazardous waste. All of the equipment used will then be decontaminated with a high pressure steam cleaner and the rinse waters will be collected and placed in the WWTU upstream of the API separator. In addition, the remaining materials [after vacuum and other removal operations have occurred] and the entire top liner then will be powerwashed with water. The liner/residue washwater will be collected in the WWTU system for handling through the oil/water separator and benzene strippers, followed by aggressive biological treatment in the other ABT unit still in service. This procedure will be followed even if the washwaters do not exhibit a hazardous characteristic.

As required by NMED, the RCRA liners will be inspected for any damage and repaired if necessary. If there is damage to the 100-mil HDPE top liner, then the upper 100-mil liner will be removed from an area of sufficient size to allow for a thorough inspection of the underlying 60-mil HDPE secondary liner. If the 60-mil liner is damaged, then it will be repaired.

There has not been any indication based on an accumulation of fluids in the underlying (non-RCRA) collection system that the RCRA 60-mil HPDE secondary liner has any damage. However, if the 60-mil liner is damaged, then the underlying (i.e., lowermost) 100-mil HDPE liner will be inspected and may also be repaired although this liner is not required. If the lowermost 100-mil liner is damaged, then the underlying environmental media (e.g., 6" layer of soil with 33% bentonite and native soils) will be investigated to determine if mobile non-aqueous phase liquid (NAPL) hydrocarbons are present immediately beneath the ponds. Only if mobile NAPL is present immediately beneath the ponds, which could migrate to ground water, will remediation of the underlying environmental media be conducted to remove the mobile NAPL. Otherwise, any impacts to the underlying media should not present a threat to human health or the environment due to the fact that multiple overlying liners will prevent any direct contact to or leaching of contaminants.

After all repairs are completed, the impacted leachate collection systems will be flushed with clean water. The leachate collection system consists of a geonet, which is designed to collect any leaks passing through the overlying 100-mil HDPE top liner, drain lines to a sump equipped with a 6" observation pipe, and a 60-mil HDPE secondary liner. Samples of the flush water will be analyzed for hazardous characteristics using methods specified in 40 CFR Part 261 C – Characteristics of Hazardous Waste to determine when flushing has adequately cleaned the collection system. The

10

collected flush water will be pumped back to the WWTU system for handling through the API separator and benzene strippers.

All hazardous waste and waste residues will be removed and properly disposed by conducting the modified closure process and there will be no potential for any post-closure escape of such wastes, thus meeting the Enforceable Document modified closure performance standards in §§265.111(a) and (b) as specified by §265.110(d)(2).



### **Construction Details**

The original Schedule of Closure in the 1995 closure plan provided about 13 weeks for the closure of the ABT units. The closure time frame will be doubled for serial closure of the South ABT unit, followed by North ABT-E and North ABT-W, plus any additional time to repair damage to the liners and address impacts to underlying environmental media.

The schedule for closure of the ABT South ABT unit is as follows:

"Description	Duration
Start of closure	
[in this case 60 days after NMED plan approval]	
Aeration of impoundments	2 weeks
Testing of treated waste water	1 week
Removal of treated waste water	1 week
Drying of residual solids	4 weeks
Testing of residual solids	l week



The current cost of ABT unit closure⁵ is presently estimated at \$35,532.00, based on the following:

"Since these impoundments are undergoing continuous treatment in which the waste stream (a D018 waste because of benzene concentration) is being rendered non-hazardous, closure will simply require:

- 1) Stop adding new waste to the treatment stream [in this case the ABT unit];
- 2) Continue treatment until TC characteristic is gone;
- 3) Empty impoundments;
- 4) Analyze sediments for TC characteristics; and
- 5) Remove and dispose of sediments.



⁵The 1995 Closure Plan estimated closure costs at \$20,800 total, based on the same type of analysis used here; but this plan updates those costs to be current for 2007.

### Cost Estimate

Vigorous aeration with diesel pump	
Operator: 168 hours @ \$30/hr	\$ 5,040
Fuel for Pump: 8 gph x \$3.00/gal x 168 hrs	4,032
Testing of treated water	
Benzene: 15 samples @ \$120/sample	1,800
Testing of residual solids	
TCLP: 15 samples @ \$500/sample	7,500
Removal of residual solids	
Labor: 2 workers @ 40 hrs/ea x \$30/hr	2,400
Disposal: 40,000 lbs x \$0.20/lb + \$2,200 freight	10,200
Washing of impoundments	
Mobil wash: 24 hours x \$80/hr	1,920
Flushing of equipment	
Mobil wash: 8 hrs x \$80.hr	640
Final testing and certification	2,000

Total Closure Cost (estimate) \$<u>35,532</u>

-

**NMED Approval with Modifications** 



BILL RICHARDSON Governor

DIANE DENISH Lieutenant Governor

### NEW MEXICO ENVIRONMENT DEPARTMENT

### Hazardous Waste Bureau

2905 Rodeo Park Drive East, Building 1 Santa Fe, New Mexico 87505-6303 Phone (505) 476-6000 Fax (505) 476-6030



RON CURRY Secretary

JON GOLDSTEIN Deputy Secretary

#### **CERTIFIED MAIL - RETURN RECEIPT REQUESTED**

August 7, 2008

James R. Schmaltz Environmental Manager Western Refining Southwest, Inc Bloomfield Refinery. P.O. Box 159 Bloomfield, NM 87413

#### RE: APPROVAL WITH MODIFICATIONS CLOSURE PLAN FOR THE AERATION LAGOONS WESTERN REFINING SOUTHWEST, INC., BLOOMFIELD REFINERY EPA ID # NMID089416416 HWB-GRCB-07-006

Dear Mr. Schmaltz:

The New Mexico Environment Department (NMED) has reviewed Western Refining Southwest, Inc., Bloomfield Refinery's (Western) *Closure Plan North and South Aeration Lagoons* (Closure Plan), dated May 2008. NMED hereby approves this Closure Plan with modifications. Western must adhere to all requirements established below.

#### Comment 1

A typographical error was found in Section 4.2 (Closure Procedures), page 10, paragraph three which states "[f]ollowing appropriate aggressive biological treatment conducted in accordance with 40 CFR §261.3(b)(2)(i), the treated wastewaters will then be routed..."

The correct citation is 40 CFR §261.31(b)(2)(i), which was stated accurately in your response to comments, Comment 6. Western must submit a replacement page with this revision.

Mr. Schmaltz August 7, 2008 Page 2 of 3

#### Comment 2

Western stated in their May 28, 2008 letter *Response to March 31, 2008 Notice of Disapproval*, Comment 14 that "[i]n NMED's letter dated June 1, 2007, which instructed Western in the preparation of the closure plan and noted the requirement for a post-closure care permit, the NMED stated that, "[t]he permit will require to conduct short-term and long-term monitoring of soil and groundwater in the vicinity of the surface impoundment." Western requests that installation of the ground water monitoring system be implemented pursuant to the post-closure care permit instead of the closure plan for the aeration lagoons."

As part of the closure process, Western must install the four monitoring wells at the locations identified in the attachment Figure 1 (refer to 40 CFR 264.112(b)(5)) and must adhere to the following:

- a. Western must follow the Monitoring Well Construction Requirements located in Section IX of the Order.
- b. The well screen must be approximately 15 feet in length, intersect the water table with approximately five feet of screen above the water table and approximately 10 feet of screen below the water table.
- c. Western must collect soil samples at five foot intervals. The most contaminated sample(s), sample(s) obtained from just above the water table, and any other sample collected at the discretion of Western must be submitted to a certified analytical laboratory for chemical analysis. The chemical analysis must include diesel range organics (DRO) extended, gasoline range organics (GRO), volatile organic compounds (VOCs), and RCRA metals. If DRO exceeds 2,000 part per million, then the samples must be analyzed for semi-volatile organic compounds (SVOCs).
- d. Groundwater samples must be collected from the four newly installed monitoring wells no later than five days after the completion of the well development. The samples must be analyzed for DRO extended, GRO, VOCs, SVOCs, RCRA metals, and general chemistry to include major cations and anions. When sampling, Western must follow the applicable requirements found in Section VIII.B (Groundwater and Surface Water Monitoring) of the July 27, 2007 Order.
- e. Since closure of the Aeration Lagoons will begin 60 days after NMED's approval, the monitoring wells must be installed 180 days from the day closure begins. Western must contact NMED one week prior to the start date of closure.



Mr. Schmaltz August 7, 2008 Page 3 of 3

Western must submit a closure report to NMED within 210 days from the day closure is completed and notify NMED in writing of the completion of closure activities at the aeration lagoons within 5 business days of the closure completion date. The closure report must address all closure activities as well as summarize all details of monitoring well installation.

Please contact Hope Monzeglio of my staff at (505) 476-6045, should you have any questions.

Sincerely,

John E. Kieling

Program Manager Permits Management Program Hazardous Waste Bureau

cc: J. Bearzi, NMED HWB

D. Cobrain, NMED HWB

H. Monzeglio, NMED HWB

L. King, EPA Region 6

A. Hains, Western Refining El Paso

File: GRCB 2008 and Reading GRCB 07-006



t .





۱





**Attachment 2** 

Photographs of Closure (Before)



Photo 1 South ABT Unit



Photo 2 Northeast ABT Unit

Western Refining Southwest, Inc. Bloomfield Refinery



Photo 3 Northwest ABT Unit

Western Refining Southwest, Inc. Bloomfield Refinery

# **Attachment 3**

**Photographs of Closure (After)** 



Photo 1 South ABT Unit – Cleaned and Repaired



Photo 2 Pressure Washing Northwest ABT Unit

Western Refining Southwest, Inc. Bloomfield Refinery



Photo 3 Northeast ABT Unit After Cleaning – Before Discharge Piping Replacement and Liner Repair

# **Attachment 4**

# **Liner Repair Photos**



Photo 1 Tear in the Liner Underneath the West Crossover Pipe between #1 ABT Unit and #2 ABT Unit



Photo 2 West Crossover Piping Replacement between #1 ABT Unit and #2 ABT Unit and Liner Repair in Progress



Photo 3 Completed Repair of Liner and West Crossover Piping in the #1 ABT Unit



Photo 4 Completed Repair of Liner and West Crossover Piping From #1 ABT Unit into #2 ABT Unit



Western Refining Southwest, Inc. Bloomfield Refinery



Photo 5 Example of Scrape in the #2 ABT Unit



Photo 6 Example of Plastic Weld Bead to Reinforce the Scrape in #2 ABT Unit



Photo 7 Completed Repair of East Crossover Piping From #1 ABT Unit to #3 ABT Unit



Photo 8 Tear on East Wall of #3 ABT Unit



Western Refining Southwest, Inc. Bloomfield Refinery



Photo 9 Repair of East Tear with a Plastic Weld Bead



Photo 10 Puncture in North Wall of #3 ABT Unit





Photo 11 Plastic Weld Bead Repair on north Wall of #3 ABT Unit

### **Attachment 5**

# **Sludge Sample Locations**

















































#1AL-4	#1AL-15	#1AL-14
#1AL-6	#1AL-8	#1AL-11
#1AL-7	#1AL-9	#1AL-10
#1AL-16	#1AL-12	#1AL-13
	#1AL-4 #1AL-6 #1AL-7 #1AL-16	#1AL-4 #1AL-15 #1AL-6 #1AL-8 #1AL-7 #1AL-9 #1AL-16 #1AL-12

ΓN





Figure 2 – Northwest ABT Unit

#3AL-4	#3AL-5	
#3AL-3		
#3AL-2		
#3AL-7	#3AL-1	
		#3AL-6

Figure 3 - Northeast ABT Unit

### **Attachment 6**

### Chain-of-Custody Forms and CD of Laboratory Analytical Reports

### **Contents**

- Sludge Characterization #1 Aeration Lagoon
- Sludge Characterization #2 Aeration Lagoon
- Sludge Characterization #3 Aeration Lagoon
- #1 Aeration Lagoon Flush Water
- #3 Aeration Lagoon Liner Water
- Soil Analysis from #1 to #2 Crossover Piping
- Soil Analysis from #3 Lagoon Discharge
- Soil Boring Samples
- Monitoring Well Ground Water Sample

### Sludge Characterization #1 Aeration Lagoon

		6								(1)	Y or I	) se	Alr Bubble	<b> </b>												<u></u>			]
			1				200	<b>77</b> 71	र्द्	- 0	স্চম্ব	-	<u>177</u>		X		$\mathbf{\lambda}$		X	,	X		$\boldsymbol{\chi}$	7				X	
5 ^{6°} a			)   1				ý,	দ	W	Q 1		9.	707	X		X		X	<u>``</u> `	X		X		X				5	al repor
	ll S			109	N			<u> </u>			IJ	צ	9-1-27	X		X		$\boldsymbol{\chi}$		X		X		Х	·		Ĉ	200	nahytic
			E	M 87	4107						(AOV	'-in	192) 0728															25	the at
	Ć	₹	tal.co	le, N	-345-	lues					(	∀C	V) 80828								_					6	, d	S.X	ated or
	A F		men	lerqu	505-	Red		s.80	2 b(	808	/ səp	loli	8081 Pes													cp	Š	1 4 h	rfy not
		Ē	viron	nbnq	Fax	ysis	(*C	IS'*(	<u>СЧ''</u>	ON	I' [₽] ON'	'I')	되) snoinA													70	යු	3	be clea
	L L	u y u y	allen	- A		Anal					als.	tel.	A 8 AROR			·						_				Full	۲	1-1	ta will I
	ina Est		- M	ШZ	3975			<u> </u>		{) 	1A9 1		(NG) 0158													0-	P	24	ted da
			I ≩	vkins	345-						709 F						_									928	Z	true	ontrac
			1201	Hav	505-		(ies	<u></u>	585		0100 814 F	nou													$\vdash$	ļ	$\overline{\mathbf{O}}$	de i	/ sub-c
				4901	Tel.		(/)u		(GE	<u>иа</u> Нд																ırks:	26	C,	h. An
				•			()	208	<u>;)</u> s,i	BM.	L + 3	311	N + X3T8										_			sema	QD	Per	ossibīli
-				5	<b></b>					_		1														<u>ur.</u>			This p
			!	1021						5																е (	کر ا		otice o
		2		citta						र्वे																	3		u se sa
		Q/		201						2				-	-	3	N	m	m	7	· <del></del>	6	5	9	9	) date			s serve
		7							$\mathbf{R}$	rdn			Č						·								g_		ŝ.
		tush	#	۱.					ž.	Gu			ative	9									_				꽃		oratorie
	ai.	Å.		8				ز ہے۔		~		- A COMPANY	Type	<u>S</u>											≻	-	A		ed labo
	Time		aí.	3			ger	4	-			YN21	Pre	~															ccredit
	punc	darđ	lame	ž	44		Jana	~	3	عم			ner nd #		1		3		Â		ry)		5	لہ	3	<u> </u>		)	offber a
	n-Arc	Stan	- ect	fa	ect ŧ		ect 1	Ĩ,	3	npler			ontai pe ar	ル	40	1	t o		5		J	2	74	7	5	- Ned	- North	$\sum$	ted to
	Tur		Б Ро Ч	8	0 d		Pro.	$\underline{\mathbb{C}}$	ノ	San			ŏĔ	G	$\mathfrak{S}$	G	5	Ξ	٣	3	Ð	Ð	コ	J	$\underline{\mathfrak{I}}$	Rece	<b>N</b> age		ontrac
							S		(uc				Q																e subc
	J	1	7				Le		lidati	1	1		est													K	2		may b
	000	13	) Ge	10	Z	J	C C		II Val				nbə	17		00		η		2		ヨ	=	9			Ę		mental
	Ř	H.	्री	49,	2	1.6	-2		4 (Fu				ਰ ਸ	1	->	1	ار	لي		الر	1	ن ا	4	1	7	·	Ē.		inviron
	^d	R	с¥		2	7-1	4		evel 4				amp	I ₿		Ā	-	<1 	-		1	$\overline{\triangleleft}$	41	A			-	-	) Hall E
	sto	ù,	D	र्भ	Ae	33	귁		Ϋ́				۵	#		#		╈		⊅∣	ŀ	#	#	Ħ		Xq p			litted to
	Cu	÷ j	R.	Å	く	୬	-ti-						X	de La	2							·				ini he			s subr
	of-	۲ ۲	7	3	9	5	1.10						Mai	3						<b></b> ,		╺╌┥		-	7	Relino	Relinc		ample
	in-	-j	10	ess:	1	ß		ige:			) ()		e	2		10		33		R		V		R	~	୍ୱ ଝ	3		sary, s
	ha	Ves	(1)	Add	1	Ú.	Fax	acka	dard	_	Щ.		<u>ب</u>	ナ	7	ž		E	~	र्		140		<u> </u>	ブ		Time		neces
-	C	entr		ailing		one <del>j</del>	tail of	/oci	Stan	othe	ĘDD		ate	_00	_					Ī						a a	e: e:		=
		ซิ		ž		4 d	en	ð				Į		FO	-	[									$\rightarrow$	Ň	Dat		ار. م

		•											(N	110	۲)	sel	qqn	ir Bi	A			Γ						_					Τ				-
		7 ¥	ז 1 1											-		<u> </u>		~~	<u>_</u>	.7		Ļ	4			+	_	_				$\downarrow$					
	ľ										<u></u>	1	57	nux	m	6	ू १ म	50 (CR	2 1	Ŷ		4	\ 7				+	+			<u> </u>	_					
	Ø L	ב ב ע ע	i I		ഉ				4	ন্থ	<u>ू</u>	w	v (	) r ]	7	10 10	J.	1973		$\overline{\mathcal{N}}$	$\mathbf{x}$			$\checkmark$		+		_	`			╀					
	5 1				871(	107	5			10	, 			( <del>\</del> (	<u>)</u> 201	ime	es)	012	8		~			$\wedge$		+	+	+				+	╀	_			
	ĺ	5 8		2	ΜN Ν	45.4									(\	10/	<u>v) a</u>	097	8				-				- -	┽				+	╈				
	Ĺ	▝▁	lents		uque	05-3				s'8	ЪС	28	08	ļs	əpi	oite	ъЧ	180	8							+				1		+	+				
	8 H U V		ironn	5	ndne	, Z	Ś		(*(	)S'	^{\$} Oc	1' ² (	DN	^{'8} O	N'I	С, न	) sı	Ioin	$\forall$			1				1						1	T				
	L	ūΧ	llenv		Alb .	ш					-			S	slet	θM	8 A	้หวง	н																		1.15
			ed w	21.12	цЧ	975							()	l∀c	)   JO	N AV	19)	015	8			<u> </u>						_				_	$\perp$				
	2 2 2			***	kins	245-3							<u>.</u> ۲) (۲.	¢09	3 p	sthc	୬₩) 	803	┦							<b>_</b>		_				<b>_</b>	_				
					Haw	505-3							(1.	811	b p		∍M)	Hd.	<u>-</u>				_			-	_	$\downarrow$				Ļ	$\downarrow$	_			4
					1901	Tel 1			( əs		1/58	10	1 2 2 2 2 2 2 2 2	-10 -				Hd.	L   				-					+	-			+	+-		rks:		And a
					ч	•					18)	5.9		. T 			+ x 		=  =				-			-	+	+				┢	+		emai		di l'inc
					5							T							- 				+			+	+	╉	$\neg$			┢	╋		<u>x</u>	1	
		tush 4-day		#/// +	1 UNATACTURATION	)						7.	2				ative its and the second s		の語れたシングを見ていていた		L	D	a .	8											Date Time	Date Time	tratoriae This cantas as notice of
	Time:	۲, X	- -		E CON	>	-		ager:			11 Da		XXX	perature		Preserva	edy I																		2	credited lahr
	Tum-Around	□ Standard	Project Nam		Neveran (	Project #:			Project Mana				Sampler:	Optice	Sample lieu		Container	I ype and #		1-1.ter	1- Yoniar	. Att		1-4 min	<b>z_</b>										Keceived by:	Received by	intracted to other a
	nain-of-Custody Record	KETERN REPINING (BINFU)			7430 CK #770	Nh Church NIN QUUTS		1014-020-020	-ax#.525-632-391	ckage:	ard K-Level 4 (Full Validation)			Type)			Time   Matrix   Sample Request ID			405 560 #1 AL- 15		ites 141 11 - 16	W   #140 10												1617 New Automatical Automatica Automatical Automatical Automatical Automatical Automatical Automatical Automatical Automatical Automatical Automatica Automatica Automatica Automatica Automatica Aut	ime: Relinquished ty:	vessoo. samilas cuhmitad ta Hall Environmantal mau ha cubon
-	່ວ	Client: W		Mailing Ac	0	1310	The second	F10016 #:	email or F	QA/QC Pa	□ Standa	Č					Date		istal -	1 88/10/1														F · •	- Control	Date:	- H

		<b>b</b>					(	Y or N	) səldduð 1iA													6			
A		łŔ				1 mn775	val	ana	1-1-1		×		5		~				$\overline{\mathbf{v}}$		5	#			
Ö		Ë Q				10	- OW	<u>a hyb</u>	10-0-0		7	7		$\checkmark$	4	` `	×		$\neg$	<u> </u>		, Yee		2	report
	L.	5	60	) )					KCI	イン		$\mathbf{X}$		Ź		X	<u> </u>	3	.	Ŕ		1 act		3	alytical
	Al BA		m 1871	t107				(AOV	-im92) 0728			4		•		<u>.</u>		-				'n	,	3	the an
	Ç	) 🕷	NN NN	345-2	lest			(	AOV) 80828													R	5	26	ted on
			nenta	505-	Regi	CB's	9 280	)8 / set	8081 Pesticio														R	వ	ly nota
	<b>NE</b> AA		ironn	ax -	/sis	(*0\$'*0	03'80	Ν' ^ε ΟΝ	IO, I) anoinA													z e V	Y	(M)	e dear
	Ų	ũ X	llenv - Alb		Analy			els	FCRA 8 Met													26	Ž	X	a vili b
	101 128	ļŠ	w.ha NE	3975	1			(HA9 1	AN9) 0168												$ \downarrow$		$\mathcal{P}$	_Ľ	ed by
	5		ww kins	345-3			(	1.402 L	EDB (Wethou									-				Cir	~	Z	ontract
			Haw	202-S			)	1.814	TPH (Method										_		-	Pel	Ŧ	2	Sub-oc
			901	Tel. 5		(leseid)	SBD)	80158	TPH Method												_	ې کې	کړ ا	Joy Tro	r⊧ Any
			4				b) Ho										-	-					$\vec{\mathbf{z}}$	မီ	ssibility
Ī				1		(1208		「鬣鬣						-			-					<u>r</u>	<u> </u>		this po
			tila																			ي (	7 7		tice of
		5	Dued				0 9														ŀ		7		as no
		Fer K	లే			(				0	σ	Q	g	7	=	9	9	3	5	ゴ	ゴ	a d	ge C		serves
2 -		7	#				8-6	2 2 2													ľ	° (			This
		lsh_	<u>ک</u>				F   1			1											$\neg$	-	¥		atories
		Ru	g			1	2		ervat	2			ment								≥	Ç	=		1 labor
	ime:	8	5	11		<u>сі</u>		TA I	rese	Ž	-														redited
	T bri	ard	ame:		l I	anag					2		.)		5		Ś		7		2				ter acc
	Arot	tand	of N F	# 5		, ŭ	)   i		e and	11	Hoi	L	101	ار	6	ل	40	ᆜ	6		5	apo of of of of of of of of of of of of of	<- fig	2	d to oth
	- m I	с С	Peroje	roje					Type	5	$\overline{\mathbb{G}}$	(1)	7	2	L L	1	Ē	3	3	$\square$	3	Receil	7§	~	Itracter
1	<u> </u>	<u> </u>				- S	 	<u>/ 188/1928</u>				$\sim$										<u></u> -	1-		subcor
	q	140				じ	latior		st IL							Í									lay be
	Ö	ち				ž	Valic		anb					n		_	·	5		و		1	3		ental n
	Re	40	3		-	ଡ଼ି	(Full		Re			$\sim$		, }		1		1	1	j			3		/ironm
	Ń	, 3	S S	钤	りげ	70 12	ei 4		hple	H	$\rightarrow$	ہے۔	->	¥	$\rightarrow$	Å		đ	-0	4			3		
	ťõ	2	ta f		-2		چ ل		Sar	با م						3			_	#		F	FI-		ed to H
	us	Ń.	7 10	el (	2	1	Ð			よく		-4		#		7		- <b>T</b>	-		-	shed .	<u>والمج</u>	-	ubmitt
	f-C	(Ja)	NY Y	J L	G	1 G			Aatrij	A luda										~	≥ :				ples s
	-	2 0	4 C	49	ß	5								6		_				7		<u> </u>			ry, san
	lair	1 AN	8	P		ex#:	ard	Type	Time	5/2	イ	SIC	->	00		R	-2	¥	7	F	7	I. N			scessa
	່ວ	st: ∠	N g		le #:	XC Pa	tandă thor		Ð	ii C								=	-+	<u> </u>	#	<u>ہ –</u> م	<u>⊐</u> ⊩ 2		ľμ
		Clier	Maili		Phor	ema QA/C	S D	ш	Dal	414	~										≥	<b>S</b>	Date:		
			• •	•						2			,	'		•	•	•	, '	•	•	1	-	•	

	ATORY							( 	N JO /	RAAR M TJT TJT			X									
	e C	c	8710	107	2		mr G	4.9	- <u>R</u>	V-Ime2) 0728	3	X		X			 					
(			MN	15.4				<b>.</b>		(AOV) 80928	3						 			·	-	
ĺ	Ĭ	enta	raue	05-20			PCB's	280	8 / sə	bioitee9 1808	3						 					
k P B		ironn	naue			(*(	05'70c	1' ² 0	N'EON	l,IO, F) enoinA	1						 				7	- - - - -
	ĩΥ	lenv	Alb	LL LL	গ্রন্থ	and the second second			SIE	steM 8 AADF	1											
5	A L	w.hal	- - -	07.F				(	НАЧ∙	10 AN9) 01 68	3											
6 1		WW	tins l	45-3	₿ ₽			()	· 1409	EDB (Wethod	1											
e			Hawk	05.3				()	1.814	PPH (Method							 		_			
			901	ע ע ע	5	( ə:	seiO\se	(G)	99198	portiam Hq1	-						 					<u>is</u>
S.			4	. <b>F</b>		(^)  	no seĐ	) He		8TEX + MTB	 						 	_				smar
		-		Т		(	1508)	-18N	E + 工I 源限級	ATM + X3TE	2				····					-+		<u>×                                    </u>
ne:	WRush 4-day		on #1 ( Innacterimente					Rank		eservative Type	2	15	91	31		2						10 8 00 119 Date Time
Tum-Around Tin	Standard	Project Name:	Aportion ag	Project #:	-	Project Manager		Sample (34/	Option of the second se	Container Pr Type and #	1-Liter	1- you an	1-11th	1-404		120						Received by:
ustody Record	Refining (BIMPId)	~	1 (2 4990	1 M 97473	32-4/6/	633-3911	K Level 4 (Full Validation)			Sample Request ID	#1AL-13	<b>→</b>	#141-14		this plant -	H H						A Nuntadio
-jo	erv		1. H.S.	17	2-1-0-1-0	505-1				Matrix	Studer	'	$\leq$								Pocharine Pocharia	Relinquish
hain	West		Addres	Par A.C.		ır Fax#: ,	Package: Idard	۲	(Type)	Time	1415	.\	1410	$\neg$	•						Tmo-	Time:
0	Client:		Mailing	AR10	Phone	email o	oA/QC D Star	D Othe		Date	1946				-							0100 0100
# Sludge Characterization #2 Aeration Lagoon

			www hallenvironmental com	4901 Hawkins NF - Alhimmin NM 87109		Tel. 505-345-3975 Fax 505-345-4107	Analysis Request	() () () () () () () () () () () () () (	5.8.2 5.8.2 1.7.2 2.01 0.5.1	(68) (68) (72) (72) (72) (72) (72) (72) (72) (72	) Ne: 0005 0005 0005 0005 0005 0005 0005 00	HA (A) (A) (A) (A) (A) (A) (A) (A) (A) (A		+ MT + MT + MT + + MT +	ВТЕХ ВТЕХ ВТЕХ В270 (5 В270 (5 В270 (5 В270 (5 В270 (7 В270 (7)))))))))))))))))))))))))))))))))))					XX X		XXX						Remarks:	+ Knowle have - copy of level I DU.	" does writer" " IT DV.	his possibility. Any sub-contracted data will be dearly notated on the analytical report.
R	Tum-Around Time:	D Standard & Rush H Clard	Project Name:	#2AL Chauderizotra	Project #:			jroject Manager:		Cludy RULADO	sampler. Cively Hind Bach Kolon		Autooliense cause wat a start wat the	Container Preservative B	Type and # Type <b>Type</b>	3) 9/02/8 (1) Jugy 5/02/1-1	2-	~	λ	-S-	9		~ /	6	01-		alute HCI -12	teceived by: Date Time	Left 10:00 Control 10:00 Contr		tracted to other accredited laboratories. This serves as notice of th
	Chain-of-Custody Record	Client Western Retring Sarthwest Inc.	Blennfiel () Dofilen	Mailing Address: SD Pmc 4990		( AUDINATION NM)	Phone #: (505) (53 - 417)	emailer Farth-Ciudy. Hurpedo Cunrco	QA/QC Package:	C Standard     X Level 4 (Full Validation)	□ Other	The DD (Type)			Date Time Matrix Sample Request ID	0/9/8 0930 Suda #2AL-1	1 0935 10 #2AL-2	0940 #2AL-3	6945 #2.AL-4	0950 #24L-S	0255 #21-6	1000 #2AL-7	1000 # 34-6	1010 #2-4L-9	1015 # 3HL-10	V 1030   #2AL-1	V TB	Date: Time: Relinquished by.	10106120 001 Curvey - DWMWVV		If necessary, samples submitted to Half Environmental may be subco

.

•

*;***

									(	N J	o ۲)	selddug 1 A	/														٦
	H	۲ ۲	• /											Ļ											2	~	
		۲Ö	;			-						1251		X	イ	X	X	$\boldsymbol{\lambda}$		X					2		Įŧ
<b>N</b>				~		1. 2 A.	8	<del>627</del>	7 5	inst-	21	1-3724	$ \mathcal{K} $	$\mathbf{x}$	ス	X	$\boldsymbol{\lambda}$	$\boldsymbol{\lambda}$	$\lambda$	$\lambda$	Ķ					,	a F
	ž			7109	. 5	م المراجع المراجع محمد الم	high	200	g	20	RE	3-9127	X	ス	$ \lambda $	X	$\boldsymbol{X}$	X	X	$\boldsymbol{\lambda}$	X		<u> </u>		3	ð	in Acre
			Ę	MN 8	ž	્ર				(∀	νοΛ·	-im92) 0728											<u> </u>		l 3	7	Į.
	č	2 4 2	ntal.c	ne Le	5-345	dues					()	40V) 80828										Ň			Å.	g.	trated (
	ļ	, ,		nerg	205	s Re			1280	1 80	səh	8081 Pestici												<b> </b>	(	<u>ج</u>	arty no
				bnq	Fax	Iysi	('C	S''O	3.00	N ^{-•} (		O.7) anoinA				<u></u>								<b> </b>	<u>P</u>	$\frac{1}{2}$	P P P
	L T		aller -	<	ы Б	Ana				(แพ		ANT VIGO													5	2	ta will
		A P	, Å	ШZ	-397:	44 44				(HV			-											Ľ.	පි	d d	1 Pot
			ŝ	vkîns	945	÷.,			(	1.81	.tr D				2			_						$\left  - \right $	5		1 test
	-			I Hav	505		(រទទ	əi(l'k	ະ ອີອງ)		08 1														4	10	N SI P
				4901	Tel.		(Áju		)) Ha	-11 +	- 38		-		-									$\left  - \right $	r, arks:		4
	ł		3				()	1208)	s'8N	 \+	- 38	TM + XJT8													Lo cem	Ξ	Iliyisa
					[	(5 4) 			Т																<u>н</u>		- 4 4
		2		2	ļ				2				$\tilde{-}$	-14	1	16	5	3	19	0	4	27			he 'مر أه '	e	1
		d l	-	94					4				240		1	I	Ũ	1	`	ĩ	ĩ	l			ې ¢ _	Ē	000
-		7		Ŝ			i		Ŕ		酸酸		810												Date 0 / (V	Date	
				Ę				0	4		い物		0												· ~		Ĕ
-		, ush		ちろ				g	1		ß	tive										4			\		- attraine
	·	Ľ		1g				ず	1			serva	1 A								$\rightarrow$	건					49
	Lime			3			der.	Ц Ц	-		福	Pres	Ž												X		ofitee
	pun	lard	ame	1		;	lana	7	Ŕ	5		ler d#	\$									R			A	5	herac
	Aro	Stanc	N SCI SCI SCI SCI SCI SCI SCI SCI SCI SCI	4	# 50		≊ct⊵	Я				ntair e an	2 la	<del>.</del>							→	202			ş II	d bay	5
	En l		Proj	#	Proj		₽ ¹ 0į	Ũ	Les C		關	δŢ	1									R			Recei	Recei	1
	Ľ	ų.					ğ	 î	2		LAPUR	0				-											1
	Б	1					YY Y	(i+c) (i+c)		1	1	est II				١			_								of yes
	õ	3	7	16	Σ		3	viali.				eque	2	3	기	り	2	Ę	6 G	0	ې م				_		- Interest
	Re	R.	701	5	2		9	,C, II	5			e Re	ī	7	)	7	A	7	3	)	-/]						
	<u>}</u>	3	Ĩ	<u> </u>		51	40	N N	5			nple		Å	A	$\overline{\mathbf{A}}$	4	Ł	4	4	포	ഫ	1			•  -	
	ğ	کر ا	82	2	eld	נ ו	Ţ					Sai	ন্থি	5	С 43	4	Ý	Jr L	Z	Å	A	1-	ć		÷-€		
	sn	J.F.			ΨY	32	•		1				T X			*1	#		=#	<del>کر</del> .	#	2		]	shed 1	shed	1
	Ļ	2	lel.	B	400	مار	- DY	•				Aatrij	1			_		_			7	-F				linqui	100
	9	5	J.Y	:2	N	20	Ü	*				2	Ň				1	_			-	2	$\dashv$		<u>8</u>		
	air	Jer 1	<u> </u>	Idres		Ň	爤	okag∉	j	ype)		Time	ہم ا	3	235	칫	S	S	Z	8	50	1				:e:	
2 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	<b>5</b>	Ne	A	Jg Ac		е #	Б	C Pac	her	<u>ו</u> מ	,	<i></i>	2	_	2	*	븨	읟	=	_	=	<del>`+ </del>	$\dashv$		<u>ř Š</u>	作	
	•	Clien		Aailin		hon	mail	AQ -	δĒ	泾		Date	3/8						$\neg$			$\rightarrow$				ate:	
	ļ		)	<u> </u> 2	1	_	۵ļ	υL		• 🗠	I	-		. 1	I	I	J	1	Į	I	I	1	ł	ļ	0 6	lo	I

Sludge Characterization #3 Aeration Lagoon

3

		www.hallenvironmental.com	4901 Hawkins NE - Albuquerque, NM 87109	Tel. 505-345-3975 Fax 505-345-4107	A Received Analysis Request and the second	(les	9 5,800 9 9 9 9 9 0 9 8 9 0 9 9 0 9 9 9 9 9	) ( 0 ² , Pd ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) )	11 + 2 1 + 1 1	181 1900 1000 1001 100 100 100 100 100 100	М + ХЭТ8 Новон нат Наем) нат Наем) нат Наем) нат Вазпо (Ма Ма Азля Вазпо (Б, Су) воза Вабов (Уо Вабов (Уо Вабов (Уо Вабов (Уо Су) воза Вабов (Уо Су Су Су Су Су Су Су Су Су Су Су Су Су	XXXXXXX									arts: antact Ciudy Hurtacto with questions	The subcontracted data will be clearly notated on the analytical report
Tum-Around Time: Results by	□ Standard E Rush /0-3/-08	Project Name:	#3AL Characterization	Project #:	● #子耳	Project Manager:	Cindy Hurtado	Sampler (44 / Rock			Container Preservative Type and # Type	0 8°22 None -1	(1) 4₀₹ (1) & ⊶₹	Ŋ	7	Ś	× × -6	L- > >			Received by: Received by: Received by: Date Time	Intracted to other accredited laboratories. This serves as notice of this possib
Chain-of-Custody Record	Client Western Refining (Binfld)		Mailing Address: 井らの CR 4990	Bloomfield, NM 874/3	Phone #: 505-632-4/6/	email or Fax#: 505- 633- 39//	QA/QC Package:	D Other	D EDD (Type)		Date Time Matrix Sample Request ID	10/21/08/1300 Sudge #3 AL-1	1305 0 #3AL-2	1310 # 3AL-3	1315 # 3 AL-4	1320 # 3BL-S	V 1325 V #3 AL-Lo	N 1330 J # 3 AL -7			- 10/21/00 215 [Linden NWT adds	If necessary, samples submitted to Hall Environmental may be subco

•

**#1 Aeration Lagoon Flush Water** 

	_	
•	,	

HALL ENVIRONMENTAL ANALYSIS LABORATORY 4901 Hawkins NE, Suite D Albuquerque, New Mexico 87109 Tel. 505.345.3975 Fax 505.345.4107 www.hallenvironmental.com		8 (8021) (10 endioe Onioe Onioe Onioe Sev (10 escoline Onioe Sev (10 escoline Sev (10 escol	9985999 90757 1, PCB 1, NO ⁵ , 1, NO	+ 381 108 bd 108 bd 08 bd 08 bd 08 bd 100 k 100	8155 + M 8155 + M 8155 + M 8162 + M 810 (Meth 810 (Meth 8270 (Sen 82508 (Y 8081 Pest 8250 (Y 8081 Pest 8250 (Y 8081 Pest 8250 (Y 8081 Pest 8250 (Y 8081 Pest 8250 (Sen 7 2 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	X			×	×					Remarks:	
104/00 Package: Std D Level 4 2 Other: Project Name:	Project #:	Project Manager:	Sampler. C. L. & Wich	Sample Temperaturke: 3 -	Number/Volume H9Cl2 HN03 OK1321	3-10A Hel 1	Iliter Amber	1 500 ml	1 500ml Red H	1 scoul NA i	1 500ml X 1				Received By: (Signature) 11.05	Receiver (Signature)
CUSTODY RECORD	CR 4990		19/1-029	633-39/1	Matrix Sample I.D. No.	H20 #1 AL find weter	-								retinushed By: (Bigneture)	Relinquished By: (Signature)
CHAIN-OF-(	Bloomfr		Phone #: 505-	Fax#: 505- 4	Date	11-19-08 8:40			·		1	** *			Date: Time: H ij-19-08 9:/0	Date: Time: I

**#3 Aeration Lagoon Liner Water** 



			l f							(	N 10	(۲ i	səldduð	iΑ														]
9.79 7.8 7.8					-																			 				eport.
		SA E		60									RCL			<u>.</u>	$\prec$	×	X									alytical r
		ĮÖ	Ę	A 87	4107						(¥	0	-ime2) 07	82		X									<u>+</u>		• * •	the an
	Ç		al.0	e, Z	345	uest						(\	10V) 808	787	×												•	ted on
		L L L	ment	erqu	505-	Req		s'8C	) d (	280	8/8	sepļ	olise9 18	08														ty nota
			viran	nbng	Fax	lysis	(*C	S'*	Эd'	105	۷' ^ε С	)NʻI	<mark>Э,</mark> Я) enoir	١A							•			 ļ		•		e deal
	Ų	u yu L	allen	- A	10	Anal						sist:	eM 8 AAC	)ਮ 20						$\times$				 				t liw et
	4	ĮŞ	h.w	» NE	3976					() 		<u>a no</u>		-8 12					·	;				 				cted da
			8	wkins	-345	ç	·			(1	.81	5 P		1 ] ]	•									 				contrac
		-1 (-	M	1 Hay	. 505		(les	e)(]	se6	5) E	1910	 1 8 C	H Wethoo	ŧΤ														ny sub-
				490	Tel		(Vln	0 51	e)	Нд	1+	38	TM + X31	.8												larks		atty. A
							(	208	3) s,	ЯM	1+	38	TEX + MT	.8												Ren		possib
			l	0-09		_										1	T	ī	/ -	۲		イン				SUS SUS	Time	is notice of this
				て 1-3																						ateo Vo	Date	This serves a
	ime:	Rush_		iver wate			er.			184			⁵ reservative Type	-	12	ANDEr	Nadil	PLACE THE -	NA .	AU02	<b>)</b>			 		101		redited laboratories.
	Turn-Around	C Standard	Project Name:	±3 A6 L	Project #:		Project Manaç			Sampler: 🧭		Semple Methods	Container Type and #	•	3- WA	- 11ter	1- Seo ml	1-Seem	1-500ml	1-500 ml						Received by:	Received by	ntracted to other acc
	ustody Record	Cining (Blm Pld)		0 CR 4990	, NM B7413	32- 4161	162-52,		ErLevel 4 (Full Validation)				Sample Request ID		#3AL live with	<b>P</b>						troblent	-			and by	led by:	mitted to Hall Environmental may be subo
	-of-Cl	ru R.		5.# SC	fre d	5-6	525-6						Matrix	~	51 H2O		 			-				 		Relinq	Relinquish	y, samples sub
	Chair	tt West		ng Addres	Bloom	e# B	l or Fax#	C Package	andard	ther	DD (Type)		e Time		\$ 33									 			Time:	If necessar
		Clien		Mailir	V	Phon	emai	QAQ	び ロ	б П			Dati		2-											Date:	Date:	

## Soil Analysis from #1 to #2 Crossover Piping



		22				(N	10 Y) I	aoedsp	вөН то	) səldı	tu8 niA	' . 			-		 		he tot	
	NMENTAL 30RATORY Vite D	exico 87109 Fax 505.345.410	al.com	JEST	5 Ro	nio Nio	<b>3</b> W	8-1 273	(۵۵۷- عدلاً) محلاً	(VO) ( - <b>9</b> -	1072 1577 8520						· · · · ·		les l	
	ALL ENVIRO NALYSIS LAU 101 Hawkins NE: S	ouquerque, New M 505,345,3975	ww.hallenvironment	ALYSIS REDI		5) 0 [*] )	808) S ' [*] O	bcb,8 No ^{s,} E :	or PA) , NO ₃ , sides /	AN9) B Mei (F, C) Pestio	Of CB AACA anoinA F808							· ·	 a only	68 40 r
			AW .	AN		( 9	zeiO\z	1) 1'1) 2B (C ^g	108 P 109 P 109 P 109 P 109 P 109 P	Netho Metho Metho	EDC (1 EDB (1 15H (1				-		 		 Rutz	11 fra
					<u>ئ</u> ا ر	100 81 (Li	ullose 208) :	199 (G 199 (G	+ 38, + 38,	1W +	BTEX- BTEX- STEX-	2	-						 Remarks:	
	<u>ل</u> مًا ک	and a second	SSAUPT								HEAL No.								 SO HIM	•
QA/QC Packag	Std 🔲 🛛 Level		246 (							Preservative	HgCl ₂ HND ₃						 	.	By (Signature)	By: (Signature)
-	Other:	Project: Name:	+-77-#	Project #:	1	Project Manager:	-t	Sampler:	Sample Temperat		Number/Volume	BON JAR	-						Reperved	Received
	ODY RECORD	Givius (RIMFIR)		4990	NM 874/3			41(2)	371/		Sample I.D. No.	# 1ALE # 2AL ( Easoner		-					ad By: (Signature)	ed By <b>t</b> (Signature)
	I-OF-CUST	Stern Re		50 CR	omfield,	1		os -635-	-529 -50.		lime Matrix	pm Soil				 			 ne: Relinquiste	ne: Relinquishe
	CHAIN	Client: W~		Address: H	Bla			Phone #: A	反#: 0		Date	11/13/08	-						13 08 Z	¹ Date:

## Soil Analysis From #3 Lagoon Discharge

**6**.

Martinesser     Antone Kristivity     Alteraction     Clean field       Illing Address     Antone Kristivity     Alteraction     Clean field     Alter Alterss       Illing Address     Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss       Illing Address     Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss       Illing Address     Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss       Illing Fraction     Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss       Illing Fraction     Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss       Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss       Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss       Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss       Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss       Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss       Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss     Alter Alterss       Alter Alterss     Alter Alter Alters     Alter Alter Alterss     Alter Al							;	I	ALI	للل ا		IRC	LNC LNC	E S	E		
Project Manne:         Project Manne:         Project Manne:           Inite Actives:         A:3 AL         Discrete:         A:3 AL         Discrete:         A:3 AL         Discrete:         A:3 AL         Discrete:         Miler         A:4 AL         Discrete:         Miler         A:4 AL         Discrete:         Miler         A:4 AL         Discrete:         D	Here Nestern Retining - Blar	mille	Standard	C Rush					Z	צי		5		2		2	•
Mile Address         Mile Addres         Mile Address         Mile Address </td <td></td> <td></td> <td>Project Name</td> <td></td> <td>-</td> <td></td> <td></td> <td>-</td> <td>d.wwv</td> <td>allenv</td> <td>ironm</td> <td>ental.</td> <td>E C</td> <td></td> <td></td> <td>   </td> <td></td>			Project Name		-			-	d.wwv	allenv	ironm	ental.	E C			 	
Bill of Experiment         Bill of Experiment         Productive in the second s	ailing Address: # 50 Rd 4990		#3.46	Discharge	- So, 1	4	1901 F	ławkir	is NE	- Alt	nque	due. l	NM 87	109			
	BI DOMPERAL NON	etter	Project #	•			Tel. 5	35-34	5-3975		ax 5	5-34	5410	~			
Inition         Rest         Art Bubbles (Fort)         Prove and standard           Charadose:         Charadose	one #: 505-632- 4161								2 - 1 - 1 1	Anal	sis R	edue	St	1. 1. 1. 1. 1. 1. 1. 1.			
Of Package: Carry Active Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standard Standa	nail or Fax#: 505-632 - 391/		Project Mana	ger.		(	(jəs (Au				(*(						
Standard         All Bubbles (Y GN)           Standard         Level 4 (Tull Validation)           Other         Sample (Tull Validation)           Sample (Tull Validation)         Sample (Tull Validation)           Tabel (Tull Validat	/QC Package:					120	Die Die				)S'*	s,g;			SIX		
Contract         Mathematic         Mathemati	Standard 🔰 Level 4 (	Full Validation)	11-4			8) s'E	/seg)				04'2	2 bC		hiv	21 W		
Image: Second	EDD (Type)		Sampler 14	I'm F		WL +	1991	(1.81	(1.40 AH)		ON' ^e	1808	(∀	0 21	81		(N 1
Image: Solution of the soluti			Compare and Compare	operation of the		. 38 . 38	4 80	t p	9 pr	sle)	DN'I	n səpi	ΌΛ·	172	もい		5 Y)
Bold     Handler     Handler     Handler       Bold     43a.     Soil     #3AL     Discheng - Soil     #3AL       Bold     43a.     Alt     Discheng - Soil     Alt     BS       Reconcel     1     Discheng - Soil     Alt     Discheng - Soil     Alt       Reconcel     1     Discheng - Soil     Alt     Discheng - Soil     Alt       Reconcel     1     Discheng - Soil     Alt     Discheng - Soil     Alt	bate Time Matrix Sample	Request ID	Container Type and #	Preservative		TM + X3		ortieM) H	ontem) 8	eM 8 AA	O,F) ano	011294 18	-lmeč) 07	ng gen	N J7.		seldduB
20 TOP - 201 #2AL "Schede - 10, A" TOP W POR - 201 #2AL "Schede - 10, A" TOP W POR - 10 + 10 + 10 + 10 + 10 + 10 + 10 + 10	1 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1		· 71		<u> SKROKSKA</u>	18	un l	9T	28 ]]	ря С	uA	08	.78	21:			-NA
e. Time: Relargiable by: B. M. H. M.	5-07 430A- 501 75AL V	10(- xb34751	A-109,30		12	-+	_		+	_[		+	$\square$	オ	≯	$\rightarrow$	$ \rightarrow$
e: Time: Relativitated by: Beneficial of the time Remarks: Time: Relativitated by: Time: Relativitated by: Relativitated by			-						$\neg$					-			
Image: Final Science by: Final Scie																	ļ
e: Time: Reinquished by: Received by: Date Time Remarks: Fine: Reinquished by: Received by: Date Time Remarks: Fine: Reinquished by: Received by: Date Time Remarks:																	
e: Time: Reinquished by: Received by: Rece																	
e: Time: Relativished by: B-9/1/05 Vorter Marken by: Received by: Received by: Received by: Received by: Received by: Received by: Date Time Remarks: Remarks:																_	
e: Time: Relinquished by: Relinquished by: Renarks: e: Time: Relinquished by: Renarks: E: Time: Relinquished by: Received							_			· [							
e: Time: Relinquished by: B-g [] to S Work from Received by: B-g [] to S Received by: Received by: Remarks: Remarks: Received by: Received by: Recei																	
e: Time: Relinquished by: B-97/1/OS Worker Marker Date Time Remarks: e: Time: Relinquished by: B-97/1/OS Worker Marker Date Time Remarks:				_			_										
e: Time: Relaquished by: 8-97/11/05 Verter Trader March 12 Jack Time Remarks: Received by: Received by: Jack Ast Time Remarks: Received by: Received by: Date Time																	
e: Time: Reliaquished by: 8-91/1:05 Vorter Trader Q 33 e: Time: Relinquished by: Received by: 120/02 935 e: Time: Relinquished by:							_					-					$\square$
e. Time: Reinquished by: Received by: 120100 955 Time Remarks:	e- Time: Dolinovichad hur		Doctord Lot		i		_		-	$\neg$		_			_	_	
le: Time: Refinquished by: Received by Date Time	8-0911:05 Vobert	rabor	Kecerved by:	rladlo	Date Time	Remar	š						•				[
-	le: Time: Reinquished by:		Received		Date Time	·······						Ξ.					

Soil Boring Samples

	. >						(N	, ol	() səidduB (	ΊÅ														Addo		
_	<b>IA</b>			5.2 - 1 - 5.2 - 1 - 5.2 - 1				·											•				4	ы С		ť
		, 1 1	თ	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		<u>ANII</u>		<u></u>	1111															372		ical repo
	MO	_	8710	107		¥0 ··· `		V U VO		78	X V			X X			X						142	3		e analyt
		l.com	NN NN	est	<u>.</u>	<del>,</del>	*		( <del>∀OA</del> ) 809	28	<u>&gt;</u>	X	X		×	×	X		$\Delta$		X	×	+uc	. 2		d on the
~	Ř J	ienta	rque,	s-cn		PCB's	2808	/ sə	81 Pesticid	08			~			$\sim$			{			-	4	e Vire	0	notateo
Z		ronm		ax o sis F	(*C	PO4,5C	' ^z ON'	103	l,IO,7) enoi	u∀													ples	3		clearly
~	ΞX	llenvi	Albi	naly				sli	stem 8 AAC	วช	$\times$			$\boldsymbol{\lambda}$				X		×			J	3		will be
Q	Į	w.ha	NE -				(H	Чd	io ANG) 01	83													So	p	ĺ	d data
Ŭ	V Z 1 V	Ň	kins   kins				(L'I	709	bonteW) B(	EC													J Q	£1 2		ntracte
			Hawi		(100		(1')	314	Pothem H														S C	ž		sub-co
			1901 Tal	<u>a</u>	(ləs		5) 85	108	potteM H											_			DrD	20		, Any
			ম		(/, 0	LZ08) 8		+ 7	91M + X3										-				emai T	Δ		ssibility
											i			~	~	2	3	~	~				<u>~ *</u>	Γ	6	this po
- I .			3								1	)	)	1	)	)	)		(* ) 	(° )	-4	Ŋ	an	ne	ž	otice of
			Ϋ́Ι.			2				9										Ì			Ë.	Ē	0	is as n
																							Date	Date	1/2	is serve
		E E		1		à	2			Š.										ह					5	es. Th
	Rush	3				23	ज		ative e		ې			d		~	59	~~	ا م	3					ł	oratori
e l			ŝ			-B	5		Typ		2 2 2	-	0	ろう	Q	23	¥1 33]	Z	<u>ک</u>	A	e01	J		$\langle$	Y	ited lat
d Tin	q	ie:	3		lager	¥		X	+ <u>P</u>			~	Ž	_	2	Z	59	-	-	*	Ź	싀	<i>.</i> <del>x</del>		al-	accred
roun	ndar	P Nan	#		t Mar	Z			ainer and #		S S	Cor	ab	a	5	dy Q	Z	ź	37-Fe	2	귀	æ		i py:	J.	o other
-ur	Ste	oject CR	ojeci		ojeçi	X	ampl		Cont		97	175	5	1 I	Š	2010	No	R	Æ	8	2	2	in the second se	Ceive	<b>1</b>	acted to
ц	2				Id In	$-\tilde{\mathcal{F}}$	ဟို	<u>97</u> 2		_		<u>@</u>	<u>ר</u>	2	<u>_</u>	শ্র	ු	3	<u> </u>	<u> </u>	Í	0	<u>~</u>	<u>к</u>	$\neg$	ibcontr
σ	24 11	+	1		ğ	ation			at D		μ			_	,							3106				ry be su
LOC LOC	ALLE	Ę	2		L Y Y	Valid			dues		is			و لـ			5				T L	-03	$\overline{\mathbf{x}}$			ntal ma
Sec	T	UT S	튑Z	29	5	Full			Re		3		$\rightarrow$	<u>ال</u> ر	_		331		ŀ		27	Ž	3	1		innme
- 	N	2	ברל		ŝ	ه 4			nple		4			4			Ŭ I			フ	H	BLA	3			all Env
tõ	Ţ,		80		ġ	Ž			Sar		Σ L			Σ			23				<u>√</u> 0	TR.	$\mathcal{P}_{\underline{s}}$	ы. А		ed to H
,us	ې ب	م تخ	¥ s	33	Ą	- -	r d							Π			<u>щ</u> т		$\dashv$		~		E Steel	shed		nbmitt
-fo	0	8			ellu				Matr		Š		-7	Ĩ	-	->	ton		<del>.</del>	⇒	A col	3		elinqu		mples :
	1	A is		Â	¥ V	ge:	_	1		+	h		_	5	-		0	-+		$\dashv$	1		<u>ه ک</u>	<u>~</u>		ary, sa
	Jes	Addre		₩. (	r Fax	Packa dard	tatior AD				١٩		$\rightarrow$	3		$\rightarrow$	<u>1</u>			7		1	Time: 1 <b>2 oC</b>	Time:		necess
U U	ient:	guilit	<b>)</b>	tone #	nail o	VQC   Stan	NEI		)ate		5		$\rightarrow$	6		$\sum$	e	$\square$		$\mathbf{F}$	5	60/	Dd tei	ie:		Ħ
	0	ž		[]	<u>e</u>	9 🗆	₹ ¤	<u>אן</u> נ		l	~~~(	ļ	ŀ	32	ĺ	7	Jer Jer			ŀ	ie c	3/31		Ď		

						1			(N	Y or	i səldduð 1	∀												[		
		ĮŘ	,																<u> </u>	-	·		⊢	tro la	<u>'</u> 2,¢	
												+							·				┝╌┥	g		report.
	Ī	2		109		1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		2	w'o	স	29(1-HI	大								.  .		$\left  \right $		オ		alyticat
			E C	M 87	4107				₩(	AOV	-imə2) 072	8		Х										M	de la	thean
	C	?₹	tal.co	le, N	-345	luest				(`	40V) 808S	8 X													43	ated on
		E S	nen	nerqu	505	Rec		PCB's	2808	/ səp	081 Pestici	8												ે ઠે.	ilu,	
~	Ž	SI S	NITOL	Ibndl	Fax	lysis	(*C	PO4,5(	' ^z ON'	ON'	D, H) anoin													J	Le Le	
5	u -	בי מ	aller	۲ ۱	5 2	Ana		<b>i</b>	 	- SISI	0 ANY) 01 6	9  	$\mathbf{X}$		Х							<u> </u>		5	F 2	ta will I
્તુ		Z	1.ww	s NE.	-397	t tas			(1.1	09 P	offem) 80:							•			~			<u>ک</u>	24	cted de
Ŕ	2		3	wkin	5-345				(1.8	17 P	odieM) HG			⁻										4	31	contra
U		] []		01 Ha	i. 50!		(Jəs	as/Die	5) 85	1081	odiaM Hq	L											•	te l	- %°	duy sub
	•			490	Те	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	(ʎju	io seĐ)	НЧТ	+ 38	ITM + X3T	3												narks N.a.l.	ZI	sility. A
			<b></b>		, —	1	()	1208) s	.8MT	+ ∃E	ITM + XIT					:								Ren A	H	s possit
									1000			10	2	4	4						$\gamma$				2	e of this
				4	]								Ň	1	Ì.					/				Time	Lime X	is notic
				Ğ				J							ŀ	•								, te	1 (1 G	erves a
				1 2				R							-	.	ł							Da	10 Da	This s
		hst		о Ту				j h à			ive ive	N N			Z						_					tories.
		<u>В</u>		ج ج	Þ,		_	- VQ	È.		ype	14C	102	الأ (	A									· · · (		labora
	Lime:			ves.			ger:∧	X	Se		Pres	3-1	H	b Z	田										$ \geq $	credited
	pun	dard	lame	3			lanag	3	F		ler d #	P		ð	1									×4	3	ther ac
	א-ר ר	Stano	ect ⊳	RA	ect #		ect _A	X	npler.	ple	ontair oe an	NS N	Poly	Å	Pol										paved b	ed to o
	<u>ام</u>		Proj	×	Proj		Proj	$\bigcirc$	San	S	ŏĘ	ک	3	E	I										Rece	Suttered
		Z						ion)	1		₽															be subc
	pro	到		٥	5		Í	alidat			uest								Ì					X		al may t
	С С		42	99	2		٤	ull <			Req	10			$\mathbf{x}$									M		nmenta
	N N N	S	ہ۔ ف	רן	Ed.	و . او	MSO	14 (F			ple	03.2					- [			}			$\mathbf{v}$	26		Enviro
	bo	ن لې	Ď	Rad	ige i	于	ia	Leve			Sam	n n			}			. \								to Hall
	ust	<b>Mill</b>	-0	Ň	Qu	Ŕ	إلم	<b>P</b>	er L	12				_		+							$\downarrow$	Led by	hed by	bmitted
	<b>S</b>	à	E. F.		A A	e.	11	-		EXC	Aatrix	3	-		->			1						inquis	inquis	ples su
	9	Ę	Ę	SS:		J.S.	×	ö			~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~				_	++	_		$\left\{ -\right\}$	$\left\{ + \right\}$			-		<u>К</u>	y. sam
		Les 1	ቚ	ddrei		9	Hax#	ard	P ation	Type	Time	131	-+		$\rightarrow$						-			ime: '2 oč	ime:	ecessar
	ប	<del>ک</del> ر ان:		ling ⊿		ne #:	ail or	QC P: Stand	iredit: VELA		ste	6			$\overline{\}$	$\uparrow$				カ						Ē
		Clie		Mai	4	Pho	em	Ň N N N N	A C	P	D	12			7				•					$\int_{\ell}^{\text{Date}}$	Date	
	ι	۶,										М)												らく		

Air Bubbles (Y or N) ANALYSIS LABORATORY HALL ENVIRONMENTAL ভ্রম্পত্থিন If necessary, samples submitted to Hall Environmental may be subcontracted to other accredited laboratories. This serves as notice of this possibility. Any sub-contracted data will be clearly notated on the analytical report. চিচ 4901 Hawkins NE - Albuquerque, NM 87109  $\overline{\mathbf{i}}$ Fax 505-345-4107 Analysis Request (AOV-im92) 0728 ×. www.hallenvironmental.com (AOV) 80928 X 8081 Pesticides / 8082 PCB's See Officer Cay Anions (F,CI,NO₃,NO₂,PO₄,SO₄) Coc 1 923 RCRA 8 Metals Kernautz Tel. 505-345-3975 (HA9 to AN9) 0168 (1.402 bodieM) 803 (1.814 bodteM) H91 (IPH Method 8015B (Gas/Diesel) Remarks: (Vino 262) H9T **HTBE +** + X3T8 (1208) 2'8MT + 38TM + X3T8 RORA INVESTIGATION - GRUP 1 N N 2 Μ H[3/19 0955  $\sim$ Ì Time ١ 1 Time ١ ì } 1 ) ١ Date Date Kelly Rebiusar Rune Preservative C Rush HW03 POO7 Sale 2 A A A Sov HNQ, えらく Type あろろ せつ Irau Turn-Around Time: . <del>Robinson (ປັນທີ່ເລີຍ</del>Project Manager のないのである Project Name: teed Sofficiently a standard () Himber Type and # Container S) Vats (C)Andbig ゆってう Samplehie Aan(i) (1)R)1.1 s)Versi i) Bild N Prily <u>کم</u> کر (3)/24) Project #: Sampler: 1) Ru Received by Receiv Mevel 4 (Full Validation) Sample Request ID Blownfield Refinen Charl-of-Custody Record 52 leys web EBS-eyciog FB-040109 So Read 4990 Trio Black <u>0911-829</u> Blookhfield Q Client: Westmy Keriniu Refinquished by: Other Excel ふく Relinguisher Matrix 100 Water Wata email or Fax#: Kelly J A Mailing Address: QA/QC Package: Time 5 EDD (Type) Accreditation Time: Time: Standard **D** NELAP Phone #: 4119 Date 62 Date

Cac       A       A         Hall ENVIRONMENTAL       HALL ENVIRONMENTAL         HALL ENVIRONMENTAL       ANALYSIS LABORATORY         Www.hallenvironimental.com       4901 Hawkins NE - Albuquerque, NM 87109         Tel. 505-345-3975       Fax 505-345-4107         Tel. 505-345-3975       Fax 505-345-4107	BTEX + MTBE + TMB's (8021)         BTEX + MTBE + TMB's (8021)         BTEX + MTBE + TPH (Gas only)         TPH Method 8015B (Gas/Diesel)         TPH (Method 504.1)         EDB (Method 504.1)         R310 (PNA or PAH)         8310 (PNA or PAH)         8310 (PNA or PAH)         8310 (PNA or PAH)         8260B (VOA).         8260B (VOA).         8260B (VOA).         8260B (VOA).         82500 (YOA)         82500 (YOA)         82500 (YOA)				ime Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Remarks: Re
Clent User Project Nursch In Project Name: Client User Project Name: Discurfield Reflect Project Name: Discurfield NM Project Name: Disconfield NM Project #:	email or Fax#: Kelly, KobiwSou Such, Coultroject Manager. avac Package: avac Package: atandard intervel 4 (Full Validation) Kelly Mu Wu Accreditation a Sampler: Tracy Ruyu Della Dithe Matrix Sample Request ID Date Time Matrix Sample Request ID Type and # Type	4/1/9 1630 Soil IM 2-2(16-18')(2) Vials MeOH - 1	VI 19 1615 Sal IM 1-2 (4-6') (2) VIALS MORE	4/2/7 [045 Soil) IMECH BLANNK [Z]VIAL NEDH	Date:     Time:     Relinduished br       Parte:     Time:     Relinduished br       Philp     P200     Parte:       Date:     Time:     Relinduished br       Date:     Time:     Recorded br:       Date:     Time:     Recorded br:       Date:     Time:     Recorded br:       If necessary, samples submitted to Hall Environmental may be subcontracted to other accredited laboratories. This serves as notice

•

Coc 3 J HALL ENVIRONMENTAL HALL ENVIRONMENTAL ANALYSIS LABORATORY www.hallenvironmental.com 4901 Hawkins NE - Albuquerque, NM 87109 Tel. 505-345-3975 Fax 505-345-4107 Tel. 505-345-3975 Fax 505-345-4107	BTEX + MTBE + TMB's (8021) BTEX + MTBE + TMB's (8021) BTEX + MTBE + TPH (Gas only) TPH Method 8015B (Gas/Diesel) TPH (Method 504.1) B210 (PUA or PAH) RCRA 8 Metals Anions (F,CI,NO ₃ ,NO ₂ ,PO ₄ ,SO ₄ ) 8310 (PUA or PAH) Anions (F,CI,NO ₃ ,NO ₂ ,PO ₄ ,SO ₄ ) 8081 Pesticides / 8082 PCB's 8250B <del>(VOA)</del> 8270 (Semi-VOA) <del>X'</del> TPH - G20 DA: AQ						Remarks: Repurbo Work Plan	# Auchyr SVOC If DRO concernation is above 2,000 pire.
Chenticular Custody Record Turn-Around Time: Clienticules fear Lepines Scathwey In Estandard Rush Clienticules fear Scathwey In Project Name: Mailing Address: So Read UPPIO Bisonuffield NM Project #: Phone #: (Sor) 632-41166	email or Fax#:     Kellly. Robivisorice Lunr.com     Project Manager:       QA/QC Package:	4/2/9/1100 Soil IIM1-4(18-20) (2) VIALS	$-\frac{1}{3}$ $-\frac{1}{3}$ $-\frac{1}{3}$	4249 1115 Sail IM 2-1(20-22)(2) VIAUS -9	UUUU (Bilaw)		Pate: Time: Relinguished by Received by Bate Time Vila 170 Aulula Luunu Fedag. Date Time	Date: Time: Belinquished by: Received by: Date Time Hard of the submitted to Hall Environmental may be subcontracted to other accredited laboratories. This serves as notice

HALL ENVIRONMENTAL ANALYSIS LABORATORY www.hallenvironmental.com wkins NE - Albuquerque, NM 87109 -345-3975 Fax 505-345-4107	od 418.1) od 504.1) etals crides / 8082 PCB's Aby (Y or N) (Y or N) C40 Aby Aby Aby Aby Aby Aby Aby Aby Aby Aby	Mietho HTPH (Metho BCTRA 8 Md 8310 (PUA 8310 (PUA 82608 (VO 82608 (VO 82500 (PO 82500 (VO 82500 (VO) 82500 (VO) 8250			× ×						Ammon and hard thomas
Turn-Around Time:	Dd 8015B (Gas/Diesel)	TPH Method Type and # Type	(2) Gross Nove -1 2) VIALS MEDH -1	3) Jan None -1	Z) LIALL MCDH - 2	3) Jan Noue - 2	CIVOR NONE -3	(1)Poly HNO3 -3 (1)Auber Nove -3	S)VOM Hich -4	Received by: Date Time Remarks:	Received by: Date Time Over Y Under (19/09 9:30 cm
Chent: Western Reptinic Southwest, Inc Client: Western Reptinic Southwest, Inc Bloom held O Reptineury Mailing Address: So Read 4940 Eleonofield, NM 87413	Phone #: 100 J 0 32 - 41 66 email or Fax#: Kelly, Robinson Cumr.com QAQC Package: D Standard It Level 4 (Full Validation) D Other I Other	Date Time Matrix Sample Request ID	(14-17-17-17-17-17-17-17-17-17-17-17-17-17-	× ×	1230 2001 IM 2-4 (18-13.51)	V V W V			4/2/9/1300 CM EBS-040309 (	1319 FDO The Relinquished by	Date: Time: Relinquished by:

HALL ENVIRONMENAL ANALYSIS LABORATORY www.hallenvironmental.com 4901 Hawkins NE - Albuquerque, NM 87109 Tel. 505-345-3975 Fax 505-345-4107	+ MTBE + TMB's (8021) + MTBE + TMB's (8021) + MTBE + TPH (Gas only) Method 8015B (Gas/Diesel) Method 504.1) Method 504.1) 5 (F,CI,NO ₃ ,NO ₂ ,PO ₄ ,SO ₄ ) 5 (F,CI,NO ₃ ,NO ₂ ,PO ₄ ,SO ₄ ) 5 (YOA) 5 (YOA) ★ (VOA) 5 (YOA) 5 (YOA) 5 (YOA)	BTE       BTE   <				Remarks: # = HOLD SVOC Sounder. Anoly & if DRD 20014 our preat Huar 2,000 ppm. Call Client PM W/ Revul.
Turn-Around Time: Turn-Around Time: Troject Name: RCRA   NVESTIGATION · Comp Project #:	Moject Manager. A.U.L.C. Marthull Sampler: TVCU Parylul Oritica Sample Employation Container Preservative	1 ype and # 1 ype 29 29 -5	(1) VOAS HCI -5 (1) VOAF NONE -5	(1) Poly HN03 -16 (1) Addred Konne -6 (5) VOA3 HCI -6 (1) VOA NONE -6	(J)VIANS MEDH -7 (J)VIANDON HC) -8 (J)Anuber Nond -9 (J)Auber Nond -9	Received by: Date Time Received by: Date Time Received by: Date Time $\eta_1' \eta_1' \eta_2 \eta_2' \eta_3 \eta_3 \eta_3 \eta_3 \eta_3 \eta_3 \eta_3 \eta_3 \eta_3 \eta_3$
Chent: Western Refinity Southworth Client: Western Refinity Southworth Blownfield Refinent Mailing Address: 30 Road 4990 Phone # (355 1.337-41101)	email or Fax#: Kelly, Robinsay (Pumr.U QAQC Package: Standard E Level 4 (Full Validation) ( Other Data Time Matrix Samolo Docurot ID	4/2/9 1630 GW FB-040209		14/1 1673 GW EBS-040207	1/219 1300 GW EBS-040309	Date:     Time:     Relinquished by:       U3/9     LDO     U.S.O       Date:     Time:     Relinquishedby:       Index:     Relinquishedby:     Index:       If necessary, samples submitted to Hall Environmental may be subcompleted

analyti u ∧ ž .

Monitoring Well Ground Water Sample

.



	www.hallenvironmental.com	2   4901 Hawkins NE - Albuquerque, NM 87109	Tel. 505-345-3975 Fax 505-345-4107	Analysis Request		s (8021 (Gas ol	- N) -	λ οι ΛΟγ ) ΛΟγ ) γ ν ν ν ν ν ν ν ν ν ν ν ν ν ν ν ν ν	BTEX + MTE BTEX + MTE BTEX + MTE TPH Method TPH Method B310 (PUA d B310 (PUA d B310 (PUA d B310 (PUA d B310 (PUA d B310 (PUA d CRA B Method B310 (PUA d B310 (PUA d CMA d C CMA d C C C CMA d C C C CMA d C C C C C C C C C C C C C C C C C C C					×			X					* Refer to attached available light.	
Turn-Around Time:	Project Name:	RCRA Imatigution - Gray	Project #: ()		Project Manager:	Keller Lobieson	Sampler Heller Kolin wor		Container Preservative Type	11100 Nove -1	STYDAN HCI -1	UPOLA HNOS -1	() Burber Noul -1	(1) Blig None -1	INVORT Nove -2	STVOA, HCI -2	1) Poly HNU3 -2	(1) Andrei Novie -2	1) Poly None -2	-	Acreived hv. Data Time	UN Shulpan gae	Received by: K I I Data Time
Chent: Western Refine Southwest ly	Elcomfield Refilms	Mailing Address: SD Read 498	Bloomfield NM 87413	Phone #: (SDS) 652~4160	email or Fax# the thy Robinson Curr. Carl	QAVQC Package:	Accreditation	WEDD (Type) Exc	Date Time Matrix Sample Request ID	5/5/9/300 Ag EBW-050509 (	-				75/9 1100 Ag BEB-05059 (						Date Time: Relinningshed Av	3819 1309 HellerKolenna	Date: Time: Relinquished by:

ALL ENVIRONMENTAL VALYSIS LABORATORY ww.hallenvironmental.com .NE - Albuquerque, NM 87109 3975 Fax 505-345-4107 Analysis Request	8310 (PMA or PAH) 8310 (PMA or PAH) Anions (F,CI,NO ₃ ,NO ₂ ,PO ₄ ,SO ₄ ) 8081 Pesticides / 8082 PCB's 8270 (Semi-VOA) 8270 (Semi-VOA) Сел СМеміътът *						Rist of curron - catron.
4901 Haw Tel. 505-3	BTEX + MTBE + TPH (Gas only) TPH Method 8015B (Gas/Diesel) TPH (Mothod 8015B (Gas/Diesel)						H See alter
N - Geoup 1	) ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) )	w, r,	Ŵ ĸ	J. W. 7		3-12	Date Time Re OrDO >>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>>
Time:	ger: Loleldo Dateservative Type	NONE	HNO3 Ninu (	Nowe	HUG Nave	Neur	5 1 0 00
Turn-Around <u>Standard</u> <u>Project Name</u> <u>RCRA</u> INI Project #:	Project Mana ALL Sampler: A Sampler: A Container Type and #	(I)VOAS	(1) Poly	(i) Poly	(1) Poly (1) Annber	() Auy G) VOUN	Received M Received M
stody Record win Suttwent, Inc Refinent and 1990 NM B7413 Iob	Sample Request ID	MW- 58		MW-57		TPLP BLANK	Iby: Rolouruu
h-of-Cu: Hein Refi Loomfield Loomfield coonfield	# Kelly, Ro ge: [ e) Excer	Ag			=		Relinquished Relinquished Relinquished
Client: Wess Client: Wess Mailing Addre	email or Faxt QA/QC Packaç Date Tim Date Tim	5/5/9 1000		5/399/11:36			Date: Time: JSJ9 130 Date: Time: If necessa

	ANAL ENVIRONMENTAL	www.hallenvironmental.com	4901 Hawkins NE - Albuquerque, NM 87109	Tel. 505-345-3975 Fax 505-345-4107	Analysis Request	() () () () () () () () () () () () () (	208) 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 2100 210 21	л л л л л л л л л л л л л л л л л л л	11 - 11 - 11 - 11 - 11 - 11 - 11 - 11 -	() () () () () () () () () () () () () (	Alt Bubbles ( Alt Bu							×					emarks:	
Turn-Around Time:	L Defandard D Rush	Project Name:	RCRA INVESTIGATION - GROND 1	Project #:		Project Manager:	X Il Cleuson	Sampar X. 0 . Parlov X.		Sampti Stronget auto (fair auto auto auto)	Container Preservative Type and # Type	1 (1) VOA NOULE - L	ETVOA HCI -6	(1) Poly HNOZ - 6	U)Amber Nove -6	Oppin None -4	(i) Au H250u -1	U. Bul 11 1	(1) Po (4)	(1) Buy -2	$0.8 k_{1} - 3$		Received by: Date Time	Received by:
Cham-of-Custody Record	Client: Western Refinitur Sathingth In	Blonnfyl) Rhuers	Mailing Address: SD Road 4990	Bloonfield, NM 87413	Phone #: (SCS) 632-411010	email or Fax#: Kelly, Rebiuson Cunn. Cor	QA/QC Package:	Accreditation	C NELAP Cother	eredd (Type) EXEL	Date Time Matrix Sample Request ID	5/5/9 1130 Ag MW - 57 (DUP)				→ → →	5/5/9 1300 Ap EBW 050509	1 1130 AN MW-57	130 Ag MW-57 (DUP)	Not Api FB-050509	1000 Bg MW-58		759 1000 Dell'Aller Della	Date: Time: Relinquished by:

#### METALS ANALYSES

Analyte	Analytical Method
Antimony	SW-846 method 6010/6020
Arsenic	SW-846 method 6010/6020
Barium	SW-846 method 6010/6020
Beryllium	SW-846 method 6010/6020
Cadmium	SW-846 method 6010/6020
Chromium	SW-846 method 6010/6020
Cobait	SW-846 method 6010/6020
Cyanide	SW-846 method 335.3/335.2 mod
Lead	SW-846 method 6010/6020
Mercury	SW-846 method 7470/7471
Nickel	SW-846 method 6010/6020
Selenium	SW-846 method 6010/6020
Silver	SW-846 method 6010/6020
Vanadium	SW-846 method 6010/6020
Zinc	SW-846 method 6010/6020

#### GENERAL CHEMISTRY ANALYSES

Analyte	Analytical Method
Total Dissolved Solids	SM-2540C
Bicarbonate	SW-846 method 310.1
Chloride	EPA method 300.0
Sulfate	EPA method 300.0
Calcium	SW-846 method 7140
Magnesium	SW-846 method 7450
Sodium	SW-846 method 7770
Potassíum	SW-846 method 7610
Manganese	SW-846 method 6010/6020
Nitrate/nitrite	EPA method 300.0
Ferric/ferrous Iron	SW-846 method 6010/6020 & SM 3500F e2+

Total Fe J Dissolved

all cation Anim balance per telly 5/6 of

No Cymide of 5/6

		www.hallenvironmental.com	4901 Hawkins NE - Albuquerque, NM 87109	Tel. 505-345-3975 Fax 505-345-4107	Analysis Request	یا یا یا یا یا یا یا یا یا	0 208) ج الالعام مح مر مر مر مر مر مر مر مر مر مر مر مر مر	// 9085 9085 1) 1) 1) 1) 1) 1) 1) 1) 1) 1) 1) 1) 1)	BE BE BE BE BE BE BE BE BE BE	TM + X3TB BTEX + MT TPH Method TPH Method TPH Method TPH (Method B10 (PUA 8310 (PUA 8310 (PUA 8081 Pestic Anions (F,C 8081 Pestic CCA (Au CVO) 8260B (VOV 8250 (Semi- CCA (Au CA CA CA CA CA CA CA CA CA CA CA CA CA							X	×					Remarks: * See attectual first for	aurons/cation auchyfe by	this rossibility. Any sub-contracted data will be deady acted a stated and the second data will be deady acted as the second as the second data will be deady acted as the second data will be deady acted as the second as the second data will be deady acted as the second as the second as the
urn-Around Time:	Standard CRush	roject Name:	RCRA INVESTIGATION - GROUP	roject #:		roject Manager:	Kulleformon	amplenckelle Colonner	auto trout to the second s	Container Preservative Type and # Type	BUNCA NONE	SIVONS HCI -1	1 Poly HNDS -1	1) Amber Nove -1	1) Poin None -1	NOA Neve -2	ET VOAS HCI - 2	1) Poly HND, -2	UAmber Nove -2	1)Poly Nove -2	3) VONI HCL 3	aikit Huser 1	eceived by Date Time	eceived by: Contract Date Time	acted to other accredited laboratories. This serves as notice of
Chan-of-Custody Record	Client Western Refiniter Sathured, he	Bloomfield Refined	Mailing Address: So Road 4990	Bloanfield NM 87413 F	Phone #: (505) 1032-4166	email or Fax#: Kelly, Robinson Qum cour F	QA/QC Package:	Accreditation	IZ/EDD (Type) Excel	Date Time Matrix Sample Request ID	1999/500 Rey MW - 54 3				$\rightarrow \qquad \rightarrow \qquad$	\$15/9 (428) Ag MW - 55 (					3359 LTD AG NELLANK (	8/3/1 was ha 1 nu - 3	Ju/9 I To Kellindvishedov: Kaluinu R	Date: Time: Relinquished by:	If necessary, samples submitted to Hall Environmental may be subconti

#### METALS ANALYSES

Analyte	Analytical Method
Antimony	SW-846 method 6010/6020
Arsenic	SW-846 method 6010/6020
Barium	SW-846 method 6010/6020
Beryllium	SW-846 method 6010/6020
Cadmium	SW-846 method 6010/6020
Chromium	SW-846 method 6010/6020
Cobalt	SW-846 method 6010/6020
Cyanide	SW-846 method 335.3/335.2 mod
Lead	SW-846 method 6010/6020
Mercury	SW-846 method 7470/7471
Nickel	SW-846 method 6010/6020
Selenium	SW-846 method 6010/6020
Silver	SW-846 method 6010/6020
Vanadium	SW-846 method 6010/6020
Zinc	SW-846 method 6010/6020

#### GENERAL CHEMISTRY ANALYSES

Analyte	Analytical Method
Total Dissolved Solids	SM-2540C
Bicarbonate	SW-846 method 310.1
Chloride	EPA method 300.0
Sulfate	EPA method 300.0
Calcium	SW-846 method 7140
Magnesium	SW-846 method 7450
Sodium	SW-846 method 7770
Potassium'	SW-846 method 7610
Manganese	SW-846 method 6010/6020
Nitrate/nitrite	EPA method 300.0
Ferric/ferrous Iron	SW-846 method 6010/6020 & SM 3500F e2+

Total V Diss R RWAF AT 5/7/09

5/4/09

**Attachment 7** 

Soil Boring and Monitoring Well Location Map







**Attachment 8** 

Soil Boring Logs & Well Construction Diagrams



Client: Western Refining Southwest, Inc. Site: SWMU Group #1, Bloomfield Refinery Job No.: 354 - Bloomfield, NM Geologist: Tracy Payne Driller: Enviro-Drill, Inc. Drilling Rig: CME 75 Drilling Method: Hollow-Stem Auger/ODEX Sampling Method: Split Spoon Comments: N36°41.964 W107°58.552

### WELL CONSTRUCTION

 Total Depth: 27.25' bgl
 State

 Ground Water: Saturated @ 22' bgl
 Fin

 Elev., TOC (ft. msl): 5519.840
 Elev., PAD (ft. msl): 5520.139

 Elev., GL (ft. msl): 5519.938
 Site Coordinates:

 N 36°41'57.80286"
 W 107°58'33.06266"

Well No.: MW-55 (IM1-1) Start Date: 4/2/2009 Finish Date: 4/2/2009





Client: Western Refining Southwest, Inc. Site: SWMU Group #1, Bloomfield Refinery Job No.: 354 - Bloomfield, NM Geologist: Tracy Payne Driller: Enviro-Drill, Inc. Drilling Rig: CME 75 Drilling Method: Hollow-Stem Auger/ODEX Sampling Method: Split Spoon Comments: N36º41.964 W107º58.552 
 Total Depth: 27.25' bgl
 Sta

 Ground Water: Saturated @ 22' bgl
 Fir

 Elev., TOC (ft. msl): 5519.840
 Elev., PAD (ft. msl): 5520.139

 Elev., GL (ft. msl): 5519.938
 Site Coordinates:

 N 36º41'57.80286"
 W 107º58'33.06266"

Well No.: MW-55 (IM1-1) Start Date: 4/2/2009 Finish Date: 4/2/2009









Client: Western Refining Southwest, Inc.Total DeSite: SWMU Group #1, Bloomfield RefineryGroundJob No.: 354 - Bloomfield, NMElev., TeGeologist: Tracy PayneElev., P.Driller: Enviro-Drill, Inc.Elev., GDrilling Rig: CME 75Site CooDrilling Method: Hollow-Stem Auger/ODEXN 36º41'SSampling Method: Split SpoonComments: N36º41.935 W107º58.507; Hydroexcavated to 8'

 Total Depth: 23.75' bgl
 Sta

 Ground Water: Saturated @ 19' bgl
 Fin

 Elev., TOC (ft. msl): 5519.308
 Elev., PAD (ft. msl): 5516.884

 Elev., GL (ft. msl): 5516.737
 Site Coordinates:

 N 36º41'56.12123"
 W 107º58'30.28358"

Well No.: MW-56 (IM1-2) Start Date: 4/1/2009 13:40 Finish Date: 4/1/2009



RPS			Well No.: MW-56 (IM1-2)
Client: Western Refining Southwest, Inc.	Total Depth: 23.75	bgl	Start Date: 4/1/2009 13:40
Site: SWMU Group #1, Bloomfield Refinery	Ground Water: Satu	urated @ 19' bgl	Finish Date: 4/1/2009
Job No.: 354 - Bloomfield, NM	Elev., TOC (ft. msl)	5519.308	
Geologist: Tracy Payne	Elev., PAD (ft. msl)	5516.884	
Driller: Enviro-Drill, Inc.	Elev., GL (ft. msl):	5516.737	
Drilling Rig: CME 75	Site Coordinates:		
Drilling Method: Hollow-Stem Auger/ODEX	N 36⁰41'56.12123"	<b>W</b> 107⁰58'30.28	8358"
Sampling Method: Split Spoon			
Comments: N36º41.935 W107º58.507; Hydroex	cavated to 8'		





Client: Western Refining Southwest, Inc. Site: SWMU Group #1, Bloomfield Refinery Job No.: 354 - Bloomfield, NM Geologist: Tracy Payne Driller: Enviro-Drill, Inc. Drilling Rig: CME 75 Drilling Method: Hollow-Stem Auger/ODEX Sampling Method: Split Spoon

Total Depth: 24.25' bgl Ground Water: Saturated @ 19' bgl Elev., TOC (ft. msl): 5521.174 Elev., PAD (ft. msl): 5518.538 Elev., GL (ft. msl): 5518.456 W 107º58'30.93918"

Well No.: MW-57 (IM1-3) Start Date: 3/31/2009 Finish Date: 3/31/2009

Site Coordinates: N 36º41'55.48996" Comments: N36º41.925 W107º58.516; Hydroexcavated to 7'




# WELL CONSTRUCTION

Client: Western Refining Southwest, Inc. Site: SWMU Group #1, Bloomfield Refinery Job No.: 354 - Bloomfield, NM Geologist: Tracy Payne Driller: Enviro-Drill, Inc. Drilling Rig: CME 75 Drilling Method: Hollow-Stem Auger/ODEX Sampling Method: Split Spoon 
 Total Depth: 24.25' bgl
 Sta

 Ground Water: Saturated @ 19' bgl
 Fir

 Elev., TOC (ft. msl): 5521.174
 Elev., PAD (ft. msl): 5518.538

 Elev., GL (ft. msl): 5518.456
 Site Coordinates:

 N 36°41'55.48996"
 W 107°58'30.93918"

Well No.: MW-57 (IM1-3) Start Date: 3/31/2009 Finish Date: 3/31/2009

Comments: N36º41.925 W107º58.516; Hydroexcavated to 7' Sampling Sample Type/Container/No Organic Vapor (ppm) Sample Depth Sample Description **Completion Results** Recovery (%) **USCS Class** Saturation Depth (ft.) bg Time <u>1</u> 0 Saturated 8 Gravelly Sand (SW) ، دو من من 18 Fine to coarse grain, loose, moist to ð saturated, dark gray, strong hydrocarbon ÷ U odor Ţ . ,0 Ó Gravelly Sand (SW) Sch. 40 PVC Slotted 0.01" Screen w/Threaded Joints 20 90 Similar to above, saturated, strong 0.0 hydrocarbon odor Gravelly Sand (SW) Ó ß 0/20 Sieve Sand Filter Pack ە مى 22-Similar to above 100 ß -23' Clay (CH) Flush Threaded Sch. 40 PVC Cap 100 High plasticity, very stiff, damp, gray and 24 light yellowish orange 24.5' 6" Sand Bed Total Depth = 24.25' BGL _ 26 28 -4 4 30-1 -32 111111 1111111111 34 36 RPS Sheet: 2 of 2 512/347-7588 404 Camp Craft Road 512/347-8243 fax Austin, Texas 78746



# Client: Western Refining Southwest, Inc. Site: SWMU Group #1, Bloomfield Refinery Job No.: 354 - Bloomfield, NM Geologist: Tracy Payne Driller: Enviro-Drill, Inc. Drilling Rig: CME 75 Drilling Method: Hollow-Stem Auger/ODEX Sampling Method: Split Spoon

Comments: N36º41.930 W107º58.548

# WELL CONSTRUCTION

 Total Depth: 27' bgl
 Sta

 Ground Water: Saturated @ 19.5' bgl
 Fir

 Elev., TOC (ft. msl): 5520.289
 Elev., PAD (ft. msl): 5520.554

 Elev., GL (ft. msl): 5520.466
 Site Coordinates:

 N 36°41'55.88264"
 W 107°58'32.76780"

Well No.: MW-58 (IM1-4) Start Date: 4/3/2009 Finish Date: 4/3/2009





# WELL CONSTRUCTION

Client: Western Refining Southwest, Inc. Site: SWMU Group #1, Bloomfield Refinery Job No.: 354 - Bloomfield, NM Geologist: Tracy Payne Driller: Enviro-Drill, Inc. Drilling Rig: CME 75 Drilling Method: Hollow-Stem Auger/ODEX Sampling Method: Split Spoon Comments: N36°41.930 W107°58.548 
 Total Depth: 27' bgl
 Sta

 Ground Water: Saturated @ 19.5' bgl
 Fir

 Elev., TOC (ft. msl): 5520.289
 Elev., PAD (ft. msl): 5520.554

 Elev., GL (ft. msl): 5520.466
 Site Coordinates:

 N 36°41'55.88264"
 W 107°58'32.76780"

Well No.: MW-58 (IM1-4) Start Date: 4/3/2009 Finish Date: 4/3/2009



**Attachment 9** 

Field Monitoring Parameter Results

#### ATTACHMENT 9 Soil Sample Vapor Screening Results

## North and South Lagoon Closure Report Western Refining - Bloomfield Refinery

Sample Interval Depth	IM-1-1 (MW-55)	IM-1-2 (MW-56)	IM-1-4 (MW-58)	Sample Interval Depth	IM-1-3 (MW-57)
0 – 2'	0	1.1	1.1	0 – 2'	1.1
2 – 4'	0.2	1.8	395	2 – 4'	237
4 – 6'	434	116	380	4 - 6'	567
6 – 8'	623	40	58	6 – 8'	
8 –10'	891	0.4	80	7 – 9'	415
10 – 12'	630	1.3	43	9 – 11'	35.9
12 – 14'	518	10.8	29	11 – 13'	107
14 – 16'	806	13.2	17	13 – 15'	14.7
16 – 18'	432	50		15 – 17'	10.8
18 – 20'	1085		312	17 – 19'	856
20 – 22'	1047				

Notes:

Units - ppm

#### ATTACHMENT 9 Ground Water Level Field Data

#### North and South Lagoon Closure Report Western Refining - Bloomfield Refinery

Well	Date	Top of Casing (ft msl)	Depth to Bottom (ft bgs)	Depth to Product (ft bgs)	Depth to Water (ft bgs)	Groundwater Elevation (ft msl)
MW-55	5/5/2009	5519.84	26.29	NPP	21.7	5498.14
MW-56	5/5/2009	5519.308	25.36	19.88	20.12	5499.188
MW-57	5/5/2009	5518.538	26.00	21.53	21.55	5496.988
MW-58	5/5/2009	5520.289	27.42	NPP	20.87	5499.419

#### Notes:

ft msl = feet above mean sea level.

ft bgs = feet below grade surface

MW = Monitoring well

NPP = No product present

Ground Water Sampling Field Parameter Data ATTACHMENT 9

# North and South Lagoon Closure Report Western Refining - Bloomfield Refinery

Well	Date	Well Volume	Temp (degrees C)	Specific Conductivity (mS/cm)	Dissolved Oxygen (mg/L)	Dissolved Oxygen (%)	Hq	ORP	(mqq) * Olq	O2 * (% by Vol)	CO2 * (% by Vol)
MW-55	5/5/2009	0	17.77	1.285	2.23	26.1	7.88	-207.2	205	1.3	2.9
		<b>T</b>	17.05	1.221	2.48	24.4	7.61	-182.4		•	I
		2	17.04	1.139	2.43	24.3	7.56	-159	•	1	ı
		З	17.17	1.118	2.65	24.6	7.6	-145.5	,		
MW-56	5/5/2009	0	17.8	2.285	2.37	23.1	7.49	-146.4	218	0.5	3.1
			17.44	2.282	0.75	7.9	7.21	-126.1	•		1
		2	17.2	2.288	1.37	13.8	7.41	-115.8	•	ſ	ſ
		З	16.98	2.277	1.3	12.7	7.37	-114.2	•	•	ł
MW-57	5/5/2009	0	19.61	2.228	4.42	42.7	7.49	-101.8	295	0.0	7.2
			19.23	2.239	1.47	15.2	7.48	-95.4	•	•	ı
		27	19.07	2.233	0.91	9.7	7.38	-91.2	1	t	ı
		ന	19,28	2.239	0.85	8.7	7.37	-83.7	•		ı
		4	19.37	2.245	0.63	6.8	7.25	-73.0	•		ı
MW-58	5/5/2009	0	17.40	1.92	2.44	32.5	7.46	-102.6	260	0.0	4.7
		¥	17.33	1.852	1.04	÷	7.31	-77.4		,	ı
		CV	17.22	1.901	1.24	<u>6</u>	7.26	-62.9	,		ı
		3	17.31	1.918	1.5	18.5	7.19	-82.7	•	ı	1

Notes:

ppm = parts per million

mg/L = milligrams per liter * = Soil gas field monitoring collected prior to ground water purging.

F:NData Validation_Group 1VAttachment 9_Field Monitoring Data Summaries/Attachment 9_Field Monitoring Data

3 of 3

**Attachment 10** 

**Well Survey Information** 

# Fixed width point lat/long/elevation listing

# **Project : WESTERN REFINERY**

User name	hwilleto	Date & Time	2:38:25 PM 7/30/2009
Coordinate System	Projection from data collector	Zone	Zone from data collector
Project Datum	(WGS 84)		
Vertical Datum		Geoid Model	GEOID03
Coordinate Units	US survey feet		
Distance Units	US survey feet		
Height Units	US survey feet		

#### Point listing

N	ame	Latitude	Longitude	Elevation	Feature Code
9	039	36°41'56.11551"N	107°58'30.25904"W	5516.737	MW-56 GRADE
9	040	36°41'56.11791"N	107°58'30.27063"W	5516.884	MW-56 PAD
9	041	36°41'56.12123"N	107°58'30.28358"W	5519.308	MW-56 TOP OF CASING
9	042	36°41'57.82465"N	107°58'33.05812"W	5519.938	MW-55 GRADE
9	043	36°41'57.81222"N	107°58'33.06291"W	5520.139	MW-55 PAD
9	044	36°41'57.80286"N	107°58'33.06266"W	5519.840	MW-55 TOP OF CASING
9	045	36°41'55.95435"N	107°58'32.16664"W	5520.016	RW-43 TOP OF CASING
9	046	36°41'55.51141"N	107°58'30.95023"W	5518.456	MW-57 GRADE
9	047	36°41'55.49527"N	107°58'30.94312"W	5518.538	MW-57 PAD BOLT
9	048	36°41'55.48996"N	107°58'30.93918"W	5521.174	MW-57 TOP OF CASING
9	107	36°41'55.88382"N	107°58'32.74826"W	5520.466	MW-58 GRADE
9	108	36°41'55.88269"N	107°58'32.75829"W	5520.554	MW-58 PAD
9	109	36°41'55.88264"N	107°58'32.76780"W	5520.289	MW-58 TOP OF CASING

#### Back to top

Attachment 11

**Soil and Ground Water Analytical Summaries** 

#### ATTACHMENT 11 Soil Sample Analytical Results Summary

				<u>^</u>		~		<u>^</u>		5.)
		6	20.	.22	(,)	-18'	6	-12	5	<u>e</u> .
	NMED Residential		(18-	(20-	4	(16-	4	03	5	8
	Soil Screening Level 12/	-	-	-1	-7	-5 (	έ	-31	4	4
		Ξ	Σ	M	- w	Ā	Ĩ	ω	Ξ	Ξ
otal Petroleum Hydrocarobans (ma	(ka)		L E I LOMICALI	r = Piulika ≁i s		E Translatio	l Barandea	L ⊒ 5.3etter da	197. SE Ge	ioussia.
Diesel Range Organics (DRO)	520	1400	510	360	< 10	< 10	4000	14	5700	50
Gasoline Range Organics (GRO)	NS	92	< 50	35	< 5.0	< 5.0	110	< 5.0	26	< 5.0
Motor Oil Range Organics (MRO)	440	570	< 50	61	80	< 50	2000	< 50	3700	< 50
olatile Organic Compounds (VOCs	(ug/kg-dry)		nn:	5.4,2 A	see to	-44.56	ಬಳಲ್ಲಾ ಇಲ್ಲಿ	ભારતીને ભાનેષ	i alige dat	a thi tati
1,1,1,2-Tetrachloroethane	43,200	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1.1.1-Trichloroethane	563.000	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1,1,2,2-Tetrachloroethane	5.550	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1,1,2-Trichloroethane	11,900	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1.1-Dichloroethane	1,400.000	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1,1-Dichloroethene	206,000	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1.1-Dichloropropene	NS	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1.2.3-Trichlorobenzene	NS	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1,2,3-Trichloropropane	86	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1,2,4-Trichlorobenzene	69;300	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1,2,4-Trimethylbenzene	58.000,000	6,000	18.0	800	< 0.826	< 0.928	2,400	< 0.928	< 0.780	2.52
1,2-Dibromo-3-chloropropane	1,840	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1.2-Dibromoetnane (EDB)	27 400	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1.2-Dichloroethana (EDC)	57,400	< 0.945	< 0.880	< 0.698	< 0.820	< 0.928	< 0.775	< 0.920	< 0.780	< 0.949
1,2-Dichloropropene	6,000	< 0.945	< 0.000	< 0.098	< 0.820	< 0.928	< 0.775	< 0.920	< 0.780	< 0.949
1.3.5-Trimethylbenzene	24 800	170	< 0.880	31.5	< 0.826	< 0.928	330	< 0.928	< 0.780	< 0.949
1.3.Dichlorobenzene	32,600	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1.3-Dichloropropage	NS	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
1.4-Dichlorobenzene	39,500	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
2.2-Dichloropropane	NS	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
2-Butanone	31,800,000	30.9	23.7	20.7	35.1	< 3.71	11.9	< 3.71	9.06	5.55
2-Chlorotoluene	NS	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
2-Hexanone	NS	< 3.78	< 3.52	< 3.59	< 3.31	< 3.71	< 3.10	< 3.71	< 3.12	< 3.80
4-Chlorotoluene	NS	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
4-Isopropyltoluene	NS	34.1	< 0.880	12.2	< 0.826	< 0.928	7.93	< 0.928	< 0.780	< 0.949
4-Methyl-2-pentanone	NS	< 3.78	< 3.52	< 3.59	14.0	< 3.71	< 3.10	< 3.71	< 3.12	< 3.80
Acetone	28,100,000	<1.5	<0.75	<0.75	< 0.75	10.3	92.1	7.73	52.9	26.6
Benzene	10,300	200	. 69	32.0	< 0.826	< 0.928	22.5	< 0.928	1.76	< 0.949
Bromobenzene	37,000	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Bromodichloromethane	14,400	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Bromoform	NS	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Bromomethane	8,510	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Carbon disulfide	460,000	< 3.78	< 3.52	4.01	< 3.31	< 3./1	< 3.10	< 3./1	< 3.12	< 3.80
Carbon tetrachionide	3,470	< 0.945	< 0.860	< 0.876	< 0.820	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Chlorosthane	63 300	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Chloroform	4 000	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Chloromethane	21.800	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
cis-1.2-DCE	76,500	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
cis-1,3-Dichloropropene	NS	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Dibromochloromethane	14.800	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Dibromomethane	NS	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Dichlorodifluoromethane	161,000	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Ethylbenzene	128.000	3.500	790	480	< 0.826	< 0.928	18.0	< 0.928	2.27	1.55
Hexachlorobutadiene	NS	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Isopropylbenzene	NS	560	370	220	< 0.826	< 0.928	6.23	< 0.928	< 0.780	< 0.949
Methyl tert-butyl ether (MTBE)	NS	35.6	69.8	25.5	76.7	1.29	0.31	11.4	< 0.780	2.63
Methylene chloride	182,000	0.03	3.50	2.88	3.97	4.18	4.40	4.19	1.79	2.03
Naphinaiene	62,100	1 200	2.000	660	< 0.820	< 0.928	16.0	< 0.928	< 0.780	20.7
n-Dutyioenzene	62 100	2 200	1 600	1 200	< 0.826	< 0.928	10.9	< 0.928	0.810	1.43
sec-Butylbenzene	60 600	460	450	310	< 0.820	< 0.928	9.59	< 0.928	< 0.780	1.67
Styrene	100.000	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
tert-Butylbenzene	106,000	5.09	2.32	2.02	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Tetrachloroethene (PCE)	12.500	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Toluene	252,000	2.55	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
trans-1,2-DCE	112.000	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
trans-1.3-Dichloropropene	NS	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Trichloroethene (TCE)	638	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Trichlorofluoromethane	588,000	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Vinyl chloride	4.370	< 0.945	< 0.880	< 0.898	< 0.826	< 0.928	< 0.775	< 0.928	< 0.780	< 0.949
Xylenes, Total	82,000	69.8	3.55	24.7	< 0.826	< 0.928	40.1	< 0.928	< 0.780	< 0.949



#### ATTACHMENT 11 Soil Sample Analytical Results Summary

								_		6
		E S	(,o	52')	<u> </u>	18')	<u> </u>	17')	_	61
	NMED Residential	ž	-8	50-	4	-9	<del>1</del> -6,	5	2-4	×
	Soil Screening Level (2)	<u> </u>	1 (1	10	5	2 (1	3 (2	3 (1	4 (2	- -
	-	-	-	-	<u> </u>	Ц	Ξ.	2	-	-
		Σ_	IM	MI	MI	IM	MI	MI	IM	M
mi-Volatile Organic Compounds	(me/kg) ⁽¹⁾	de la	W. C. Hw	Haler Ader	1.1.54	N AR ART	1946 - 449		CLARM	357.3
1 2 4-Trichlorobenzene	69	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
1.2.4-Inchlorobenzene	37	< 0.20	NA NA	NA.	ΝΔ	NA.	<10	NA	<10	NA
1.2 Dichlorobenzene	23	< 0.20	NA NA	ΝΔ	NΔ	NA NA	<1.0	NA	<1.0	NA
1.3-Dichlorobenzene	10	< 0.20	NA	NIA	NA NA	NA	< 1.0	NA NA	< 1.0	
1,4-Dichlorobenzene	40	< 0.20	NA NA	NA	NA	NA NA	< 1.0	INA	< 1.0	NA NA
2,4,5-Trichlorophenol	6,110	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
2,4,6-Trichlorophenol	6	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
2.4-Dichlorophenol	183	< 0.40	NA	NA	NA	NA	< 2.0	NA	< 2.0	NA
2,4-Dimethylphenol	1,220	< 0.30	NA	NA	NA	NA	< 1.5	NA	< 1.5	NA
2,4-Dinitrophenol	122	< 0.40	NA	NA	NA	NA	< 2.0	NA	< 2.0	NA
2,4-Dinitrotoluene	122	< 0.50	NA	NA	NA	NA	< 2.5	NA	< 2.5	NA
2.6-Dinitrotoluene	NS	< 0.50	NA	NA	NA	NA	< 2.5	NA	< 2.5	NA
2-Chloronaphthalene	NS	< 0.25	NA	NA	NA	NA	< 1.3	NA	< 1.3	NA
2-Chlorophenol	166	< 0.20	NA	NA	NA	NA	<10	NA	< 1.0	NA
2 Methylnanbthalene	310	0.58	NA	NA	NA	NA	13	NA	<13	NA
2 Methylaboaol	NC	< 0.50	NA	NA	NA	NA	225	NA	<25	NA
2-Methyphenor	NIC	< 0.00		NA	NA NA	NA	< 1.0	NA	<10	NA
2-initroaniline	NO	< 0.20			NA NA		< 1.0		<u>&lt; 1.0</u>	NA NA
2-Nitrophenol	NS	< 0.20					< 1.0		< 1.0	
3,3 -Dichlorobenzidine	11	< 0.25	NA	NA	NA	NA	< 1.3	NA	< 1.3	NA
3+4-Methylphenol	NS	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
3-Nitroaniline	NS	< 0.20	NA	NA	ŇA	NA	< 1.0	NA	< 1.0	NA
4.6-Dinitro-2-methylphenol	NS	< 0.50	NA	NA	NA	NA	< 2.5	NA	< 2.5	NA
4-Bromophenyl phenyl ether	NS	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
4-Chloro-3-methylphenol	NS	< 0.50	NA	NA	NA	NA	< 2.5	NA	< 2.5	NA
4-Chloroaniline	NS	< 0.50	NA	NA	NA	NA	< 2.5	NA	< 2.5	NA
A Chlorophenyl phenyl ether	NS	< 0.20	NA	NA	NA	NA	<10	NA	<10	NA
4 Nitroaniline	NS	< 0.25	NA	NA	NA	NA	<13	NA	<1.0	NA NA
4 Number al	NE	< 0.25	NA		NA	NA.	<1.5	NA	<1.0	NA
4-Nitrophenol	JN3	< 0.20	NA NA	NA NA	NA	NA NA	< 1.0	NA	< 1.0	
Acenaphthene	3,730	< 0.20	NA .	NA	NA	NA	< 1.0	NA	< 1.0	NA
Acenaphthylene	NS	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Aniline	NS	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Anthracene	22,000	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Azobenzene	NS	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Benz(a)anthracene	NS	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Benzo(a)pyrene	1	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Benzo(b)fluoranthene	6	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Benzo(g,h,i)pervlene	NS	< 0.50	NA	NA	NA	NA	< 2.5	NA	< 2.5	NA
Benzo(k)fluoranthene	62	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Benzoic acid	NS	< 0.50	NA	NA	NA	NA	225	NA	225	NA
Benzul alashol	NIC	< 0.20	NA	NA NA	NA	NA NA	<10	NA	<10	- NA
Benzyl alconol	NJ NIC	< 0.20			NA NA	NA	<1.0	NA NA		NA
Bis(2-chloroethoxy)methane	INS	< 0.20			NA NA	INA NA	< 1.0		< 1.0	
Bis(2-chloroethyl)ether	Marcussille, Zhimmer, and	< 0.20	NA	NA.	NA	NA	< 1.0	NA	< 1.0	NA
Bis(2-chloroisopropyl)ether	39	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Bis(2-ethylhexyl)phthalate	347	< 0.50	NA	NA	NA	NA	< 2.5	NA	< 2.5	NA
Butyl benzyl phthalate	NS	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Carbazole	NS NS	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Chrysene	615	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Dibenz(a,h)anthracene	i i i i i i i i i i i i i i i i i i i	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Dibenzofuran	142	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Diethyl phthalate	48,900	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Dimethyl phthalate	100.000	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Di-n-butyl phthalate	6 110	< 0.50	NA	NA	NA	NA	< 25	NA	< 25	NA
Di n ootul phthalate	NS	< 0.20	NA	NA	NA	NA	<10	NA NA	<10	NA NA
Di-ii-octyl philalate	2:000	~ 0.20	NA	NA NA	NA NA	NA	<1.0	NA	< 1.0	NA
Fluoranthene	2:290	0.25		NA NA			1.5		1.5	
Fluorene	2,000	< 0.50	NA NA	NA NA	NA NA	NA NA	< 2.5	NA	< 2.5	NA
Hexachlorobenzene	1	< 0.20	NA	NA NA		NA	< 1.0	NA	< 1.0	
Hexachlorobutadiene	NS	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Hexachlorocyclopentadiene	366	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Hexachloroethane	61	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Indeno(1.2,3-cd)pyrene	6	< 0.25	NA	NA	NA	NA	< 1.3	NA	< 1.3	NA
Isophorone	5,120	< 0.50	NA	NA	NA	NA	< 2.5	NA	< 2.5	NA
Naphthalene	80	0.25	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Nitrobenzene	23	< 0.50	NA	NA	NA	NA	< 2.5	NA	< 2.5	NA
N-Nitrosodi-n-propulamine	NS	< 0.20	NA	NA	NA	NA	<10	NA	<10	NA
N Nitrosodinbervismine	002	20.20	NA NA	NA NA		NA NA	210	NA NA	< 1.0	NA NA
IN-INITOSOCIPTIEN Viamine	20	0.20	NA NA		I NA	NA NA	< 1.0		<u>&lt;1.0</u>	
Pentachlorophenol	30	< 0.40		NA	NA	NA NA	< 2.0	NA	< 2.0	NA
Phenanthrene	1;830	< 0.20	NA	NA NA	NA	NA	1 1.1	NA	< 1.0	NA
Phenol	18,300	< 0.20	NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Ругепе	2,290	< 0.20	NA NA	NA	NA	NA	< 1.0	NA	< 1.0	NA
Pyridine	NS	< 0.50	NA	NA	NA	NA	< 2.5	NA	< 2.5	NA

#### ATTACHMENT 11 Soil Sample Analytical Results Summary

#### North and South Aeration Lagoon Closure Report Western Refining Southwest, Inc. - Bloomfield Refinery

		NMFD Residential Soil Screening Level ⁽²⁾	IM I-1 (8-10)	IM 1-1 (18-20')	IM 1-1 (20-22')	IM 1-2 (4-6')	IM 1-2 (16-18')	IM 1-3 (4-6')	IM 1-3 (15-17')	(M 1-4 (2-4')	[M 1-4 (18-19.5')
Tot	al Metals (mg/kg)		Section 200	15 ANZA (;	e etable	1742 h. NA	-164 S. Shi	vilsad (d), «	슬날날씨를	气动动动	t of the late
	Arsenic	4	< 2.5	< 2.5	< 2.5	< 2.5	< 2.5	2.5	< 2.5	2.8	< 2.5
Ē	Barium	15.600	82	170	120	160	280	200	110	200	360
- 6	Cadmium	39	< 0.10	< 0.10	< 0.10	< 0.10	< 0.10	< 0.10	< 0.10	< 0.10	< 0.10
. Į	Chromium	100,000	2.0	2.9	15	8.4	29	6.9	8.6	160	6.4
1	Lead	400	2.3	2.1	2.0	5.1	2.3	5.9	1.1	8.8	6100
- 1	Мегсигу	100,000	< 0.033	< 0.033	< 0.033	< 0.033	< 0.033	< 0.033	< 0.033	< 0.033	< 0.033
1	Selenium	391	< 13	< 13	< 13	< 12	< 12	< 13	< 12	< 13	< 12
	Silver	391	< 0.25	< 0.25	< 0.25	< 0.25	< 0.25	< 0.25	< 0.25	< 0.25	< 0.25

#### Notes:

 (1) = SVOC analysis was only required for samples with DRO results greater than 2,000 mg/kg.
 (2) = NMED Soil Screening Levels. June 2006. Future reporting will be compared to the revised Soil Screening Levels issued by NMED in August 2009. mg/kg = milligrams per kilogram

bgs = below grade surface ug/kg = micrograms per kilogram

	Screening					
	Level ⁽¹⁾	MW-55	MW-56	MW-57	MW-57 (DLP)	MW-58
otal Petroleum Hydrocarbons (TPHs) - (m	ал Соте	14.47 No. 24.05				PERSONCE 2003
Diesel Range Organics (DRO)	172	5.7	5.2	5.0	<b>4</b> .7	3.5
Motor Oil Range Organics (MRO)	134	<u> </u>	< 5.0	< 5.0	< 5.0	< 5.0
Gasoline Range Organics (GRO)	1.0 1	42	16	8.0	73	21
Volatile Organic Compounds (VOCs) - (ug/	T.)			Lanar Salakitis	Codespondent	WE WE WE WE WE
1 1 1 2-Tetrachloroethane	0.52	< 20	< 5.0	< 5.0	< 5.0	< 5.0
1 1 1-Trichloroethane	60	< 20	< 5.0	< 5.0	< 5.0	< 5.0
1 1 2 2 Tetrachloroethane	10	< 40	< 10	< 10	< 10	< 10
1,1,2,2 ⁻ Techloroethane	10	< 20	< 5.0	< 5.0	< 5.0	< 5.0
1 1-Dichloroethane	- 25	< 20	< 5.0	< 5.0	< 5.0	< 5.0
1.1-Dichloroethene		< 20	< 5.0	< 5.0	< 5.0	< 5.0
1 1-Dichloropropene		< 20	< 5.0	< 5.0	< 5.0	< 5.0
1,7 2 3-Trichlorobenzene	70	< 20	< 5.0	< 5.0	< 5.0	< 5.0
1,2,3-Trichloropropage	0.0096	< 40	< 10	< 10	< 10	< 10
1.2.4-Trichlorobenzene	- 0.0020	< 20	< 50	< 50	< 5.0	< 5.0
1,2,4-Trimethylbenzene	- 15	510	340	270	280	81
1,2,4-Trimentytoenzene	0.2	< 40	< 10	< 10	< 10	< 10
1.2-Dibromoethane (EDB)	0.2	< 20	< 50	< 5.0	< 5.0	< 5.0
1,2-Dichlorobenzene	600	< 20	< 5.0	< 5.0	< 5.0	< 5.0
1,2-Dichloroethane (EDC)	5	< 20	< 5.0	< 5.0	< 5.0	93
1,2-Dichloropropage	5	< 20	< 5.0	< 5.0	< 5.0	-50
1.3.5 Trimethylbenzene	12	120	61	36	37	47
1,3,5-Himenybenzene	12	<u> </u>	< 5.0	- 50	-50	- 50
1,3-Dichloropropage		< 20	< 5.0	< 5.0	< 5.0	< 5.0
1,3-Dichlorobenzene	75	< 20	< 5.0	< 5.0	< 5.0	< 5.0
1,4-Dichlorobenzene	23	100	210	220	250	58
2.2 Dichloropropage	2.5	< 10	<u></u>	< 10	< 10	< 10
2,2-Dichloropropane	7100	< 200	50	< 10	< 50	< 50
2-Butanone	7100	< 200	50	< 50	< 50	< 50
2-Unitrololuene		< 200	< 50	< 50	< 50	< 50
2-mexanone	150	120	260	250	200	87
2-Methymaphthalene	150	- 20	200	250	250	< 5.0
4-Chiorodoluene	-	< 20	82	9.2	9.0	< 5.0
4-Isopropynomene		< 200	<u> </u>	< 50	- 50	< 5.0
4-Methyl-2-pentanone	22000	< 200	82	< 50	54	00
Renzena	5	10000	2800	1100	910	1000
Bromohenzene	20	< 20	2000	< 5.0	< 5.0	< 5.0
Bromodichloromethane	1 1	< 20	< 5.0	< 5.0	< 5.0	< 5.0
Bromoform	8.5	< 20	< 5.0	< 5.0	< 5.0	< 5.0
Bromomethane	8 7	< 20	< 5.0	< 5:0	< 5.0	< 5.0
Carbon disulfide	1000	< 200	< 50	< 50	< 50	< 50
Carbon Tetrachloride	1000	< 20	< 50	< 50	< 50	< 50
Chlorobenzene	100	< 20	< 5.0	< 5.0	< 5.0	< 5.0
Chloroethane		< 40	< 10	< 10	< 10	< 10
Chloroform	100	< 20	< 50	< 50	< 50	< 5.0
Chloromethane	1.8	< 20	< 5.0	< 5.0	< 50	< 5.0
cise 1.2-DCF	70	< 20	< 5.0	< 5.0	< 5.0	< 5.0
cis-1.3-Dichloropropene		< 20	< 5.0	< 5.0	< 5.0	< 5.0
Dibromochloromethane	0.8	< 20	< 5.0	< 5.0	< 5.0	~ 5.0
Dibromomethane	370	~ 20			< 5.0	<u> </u>
Dichlorodifluoromethene	300	~ 20		< 5.0	< 5.0	<u> </u>
Ethylhonzane	700	1300	310	260	260	< 3.0 210
Elfyldenzene	100		510	200	200	210
	0:00	~ 20	<u> </u>	<u> </u>	< 5.0	< 5.0
Isopropyidenzene	080	1000	42	4/	40	18000
Methyl tert-butyl ether (MIBE)	12	1900	1 2100	1 1900	1 1/00	1 12000



	Screening					
	Level (1)	MW-55	MW-56	MW-57	MW-57 (DUP)	MW-58
Methylene Chloride	5	< 60	< 15	< 15	< 15	< 15
Naphthalene	0.14	330	360	380	430	100
n-Butylbenzene	68	< 20	16	16	16	6.1
n-Propylbenzene	60.8	98	71	78	81	130
sec-Butylbenzene	68	< 20	11	12	13	15
Styrene	100	< 20	< 5.0	< 5.0	< 5.0	< 5.0
tert-Butylbenzene	Sec. 4	< 20	< 5.0	< 5.0	< 5.0	6.0
Tetrachloroethene (PCE)	20	< 20	< 5.0	< 5.0	< 5.0	< 5.0
Toluene	750	760	5.5	< 5.0	< 5.0	< 5.0
trans-1,2-DCE	100	< 20	< 5.0	< 5.0	< 5.0	< 5.0
trans-1,3-Dichloropropene		< 20	< 5.0	< 5.0	< 5.0	< 5.0
Trichloroethene (TCE)	5	< 20	< 5.0	< 5.0	< 5.0	< 5.0
Trichlorofluoromethane	1300	< 20	< 5.0	< 5.0	< 5.0	< 5.0
Vinyl chloride	1	< 20	< 5.0	< 5.0	< 5.0	< 5.0
Xylenes, Total	620	1500	390	78	78	320
Volatile Organic Compounds (ug/I	$oldsymbol{ ho}$ ) is a set of the set of	en andere and	l Archithe State		a and the second se	
1,2,4-Trichlorobenzene	-	< 10	< 10	< 20	< 20	< 20
1,2-Dichlorobenzene	600	< 10	< 10	< 20	< 20	< 20
1,3-Dichlorobenzene	-	< 10	< 10	< 20	< 20	< 20
1,4-Dichlorobenzene	75	< 10	< 10	< 20	< 20	< 20
2,4,5-Trichlorophenol	3700	< 10	< 10	< 20	< 20	< 20
2,4,6-Trichlorophenol	6.1	< 10	< 10	< 20	< 20	< 20
2,4-Dichlorophenol	110	< 20	< 20	< 40	< 40	< 40
2,4-Dimethylphenol	730	< 10	< 10	< 20	< 20	< 20
2,4-Dinitrophenol	73	< 20	< 20	< 40	< 40	< 40
2,4-Dinitrotoluene	73	< 10	< 10	< 20	< 20	< 20
2,6-Dinitrotoluene	37	< 10	< 10	< 20	< 20	< 20
2-Chloronaphthalene	2900	< 10	< 10	< 20	< 20	< 20
2-Chlorophenol	180	< 10	< 10	< 20	< 20	< 20
2-Methylnaphthalene	150	53	85	100	100	38
2-Methylphenol	1800.	< 10	< 10	< 20	< 20	< 20
2-Nitroaniline	=	< 10	< 10	< 20	< 20	< 20
2-Nitrophenol	-	< 10	< 10	< 20	< 20	< 20
3,3 ⁻ Dichlorobenzidine		< 10	< 10	< 20	< 20	< 20
3+4-Methylphenol	180	< 10	< 10	< 20	< 20	< 20
3-Nitroaniline	3.2	< 10	< 10	< 20	< 20	< 20
4,6-Dinitro-2-methylphenol		< 20	< 20	< 40	< 40	< 40
4-Bromophenyl phenyl ether		< 10	< 10	< 20	< 20	< 20
4-Chloro-3-methylphenol	1	< 10	< 10	< 20	< 20	< 20
4-Chloroaniline	+	< 10	< 10	< 20	< 20	< 20
4-Chlorophenyl phenyl ether		< 10	< 10	< 20	< 20	< 20
4-Nitroaniline		< 10	< 10	< 20	< 20	< 20
4-Nitrophenol		< 10	< 10	< 20	< 20	< 20
Acenaphtnene	2200	< 10	< 10	< 20	< 20	< 20
Acenaphthylene	- 10	< 10	< 10	< 20	< 20	< 20
Aniine	12	< 10	< 10	< 20	< 20	< 20
Anuracene	0.12	< 10	< 10	< 20	< 20	< 20
Representation	0.12	< 10	< 10	< 20	< 20	< 20
Denz(a)anthracene	0.029	< 10	< 10	< 20	< 20	< 20
Benzo(a)pyrene	0.2	< 10		< 20	< 20	< 20
Benzo(b)Huoranthene	0.029			< 20	< 20	< 20
Benzo(g,n,1)peryiene	0.20	< 10	< 10	< 20	< 20	< 20
Denzo(k)Huorantnene	0.29	< 10		< 20	< 20	< 20
Benzyl alaobal	10000	< 10	< 20	< 40	< 40	< 40
	10000			<u> </u>	<u> </u>	<u> </u>



	Screening					
	Level (1)	MW-55	MW-56	MW-57	MW-57 (DUP)	MW-58
Bis(2-chloroethoxy)methane	110	< 10	< 10	< 20	< 20	< 20
Bis(2-chloroethyl)ether	0.012	< 10	< 10	< 20	< 20	< 20
Bis(2-chloroisopropyl)ether	-	< 10	< 10	< 20	< 20	< 20
Bis(2-ethylbexyl)phthalate	6	< 10	< 10	< 20	< 20	< 20
Butyl henzyl phthalate	35	< 10	< 10	< 20	< 20	< 20
Carbazole	55	< 10	< 10	< 20	< 20	< 20
Chrysene	20	< 10	< 10	< 20	< 20	< 20
Dihang(a h)anthragana	0.0020	< 10	< 10	< 20	20	< 20
Dibenz(a,ii)antiracene	0.0029	< 10	< 10	< 20	< 20	< 20
Didenzoiuran	-	< 10	< 10	< 20	< 20	< 20
Dietnyl phthalate	29000	< 10	< 10	< 20	< 20	< 20
Dimethyl phthalate	-	< 10	< 10	< 20	< 20	< 20
Di-n-butyl phthalate	-	< 10	< 10	< 20	< 20	< 20
Di-n-octyl phthalate	- 100	< 10	< 10	< 20	< 20	< 20
Fluoranthene	1500	< 10	< 10	< 20	< 20	< 20
Fluorene	1500	< 10	< 10	< 20	< 20	< 20
Hexachlorobenzene	1	< 10	< 10	< 20	< 20	< 20
Hexachlorobutadiene	0.86	< 10	< 10	< 20	< 20	< 20
Hexachlorocyclopentadiene	50	< 10	< 10	< 20	< 20	< 20
Hexachloroethane	4.8	< 10	< 10	< 20	< 20	< 20
Indeno(1,2,3-cd)pyrene	0.029	< 10	< 10	< 20	< 20	< 20
Isophorone	71	< 10	< 10	< 20	< 20	< 20
Naphthalene	0.14	200	120	150	150	45
Nitrobenzene	3.4	< 10	< 10	< 20	< 20	< 20
N-Nitrosodimethylamine	0.00042	< 10	< 10	< 20	< 20	< 20
N-Nitrosodi-n-propylamine	0.0096	< 10	< 10	< 20	< 20	< 20
N-Nitrosodiphenylamine	14	< 10	< 10	< 20	< 20	< 20
Pentachlorophenol	1	< 20	< 20	< 40	< 40	< 40
Phenanthrene		< 10	< 10	< 20	< 20	< 20
Phenol		15	< 10	< 20	< 20	< 20
Prese	1100	15	< 10	< 20	< 20	< 20
Pyrelle	1100	< 10	< 10	< 20	< 20	< 20
	J	< 10	<u> </u>	<u> </u>	I < 20 A Statistication Activ	< 20
Tetal Dissolved Calida		740	1700	1700 A 1940296	1000 1000	1400
Total Dissolved Solids	1000	740	1/00	1/00	1600	1400
Chloride	250	11	270	260	260	230
Fluoride	1.0	0.56	0.56	0.56	0.57	0.38
Nitrogen, Nitrate (As N)	10	0.32	8.1	0.14	0.15	0.14
Nitrogen, Nitrite (As N)	1.0	< 0.10	< 1.0	< 1.0	< 1.0	< 1.0
Phosphorus, Orthophosphate (As P)		< 0.50	< 0.50	< 0.50	< 0.50	< 0.50
Sulfate	600	4.0	19	9.3	.9.3	110
Alkalinity, Total (As CaCO3)		570	1100	1100	1100	800
Bicarbonate	-	570	1100	1100	1100	800
Carbonate		< 2.0	< 2.0	< 2.0	< 2.0	< 2.0
Specific Conductance (umhos/cm)		1200	2100	2100	2100	1900
Calcium	-	35	130	120	120	130
Iron		0.88	0.26	0.32	0.30	0.020
Magnesium		11	36	44	44	33
Potassium		6.0	3.1	4.7	4.7	4.6
Sodium	-	220	420	410	410	340
l Metals - (mg/L)	a sa	a di Baran I.		Strange - T	te se same	S. S. States
Arsenic	0.01	< 0.020	< 0.020	< 0.020	< 0.020	< 0.020
Barium	1.0	0.60	1.2	1.8	1.7	0.53
Cadmium	0.005	< 0.0020	< 0.0020	< 0.0020	< 0.0020	< 0.0020
Calcium		36	130	120	120	180
Chromium	0.05	< 0.0060	< 0.0060	< 0.0060	< 0.0060	0.0001
Iron	1.0	33	20	37	37	20.0091
101	List was to be a second s	ູງເປ	4.7	J. 3./	3.1	2 <del>7</del>

#### North and South Aeration Lagoon Closure Report Western Refining Southwest, Inc. - Bloomfield Refinery

	Screening Level ⁽¹⁾	MW-55	MW-56	MW-57	MW-57 (DUP)	MW-58
Lead	0.015	0.0098	< 0.0050	< 0.0050	< 0.0050	0.13
Magnesium	-	12	42	47	46	44
Manganese	0.2	1.6	3.6	3.5	3.4	5.3
Mercury	0.002	< 0.00020	< 0.00020	< 0.00020	< 0.00020	< 0.00020
Potassium		6.0	3.5	5.0	5.0	5.9
Selenium	0.05	< 0.050	< 0.050	< 0.25	< 0.25	< 0.25
Silver	0.05	< 0.0050	< 0.0050	< 0.0050	< 0.0050	< 0.0050
Sodium	-	220	430	420	420	350

Notes:

(1) = Screening level is most conservative regulatory screening limit based on guidelines outlined in Order.

ug/L = micrograms per liter

mg/L = milligrams per liter

Attachment 12

Quality Assurance / Quality Control Report

# **1.0 DATA VALIDATION INTRODUCTION**

This summary presents data verification results for soil and groundwater samples collected from monitoring wells installed at the Bloomfield Refinery in accordance with the approved North and South Aeration Lagoon Closure Work Plan. The data review was performed in accordance with the procedures specified in the Order issued by NMED (NMED, 2007), USEPA Functional Guidelines for Organic and Inorganic Data Review, and quality assurance and control parameters set by the project laboratory Hall Environmental Analysis Laboratory, Inc.

A total of 9 soil samples and 5 groundwater samples were collected between March 2009 and May 2009 in accordance with the North and South Aeration Lagoons Closure Work Plan. Soil and groundwater samples were submitted to Hall Environmental Analysis Laboratory for the following parameters:

- volatile organic compounds (VOCs) by USEPA Method 8260B;
- semi-volatile organic compounds (SVOCs) by USEPA Method 8270;
- Gasoline and diesel range organics by SW-846 Method 8015B;
- Total recoverable metals (arsenic, barium, cadmium, chromium, lead, selenium, and silver) by SW846 Method 6010/6020;
- Mercury by EPA Method 7470.

Groundwater samples submitted to Hall Environmental Analysis Laboratory were also analyzed for the following additional analytes:

- Anions (chloride, Nitrate+Nitrite, sulfate, fluoride, ) by USEPA Method 300.0;
- Alkalinity (total alkalinity, carbonate, and bicarbonate) by SM 2320B;
- Dissolved metals (iron, calcium, magnesium, manganese, potassium, and sodium) by USEPA Method 6010B;
- Specific conductance by USEPA Method 120.1; and
- Total dissolved solids (TDS) by Method SM2320B

Additionally, 10 quality assurance samples consisting of trip blanks, field blanks, equipment rinsate blanks, and field duplicates were collected and analyzed as part of the investigation activities. Table A presents a summary of the sample identifications, laboratory sample identifications, and requested analytical parameters.

# 2.0 QUALITY CONTROL PARAMETERS REVIEWED

Sample results were subject to a Level II data review that includes an evaluation of the following quality control (QC) parameters:

- Chain-of-Custody;
- Sample Preservation and Temperature Upon Laboratory Receipt
- Holding Times;
- Blank Contamination (method blanks, trip blanks, field blanks, and equipment rinsate blanks);
- Surrogate Recovery (for organic parameters);
- Laboratory Control Sample (LCS) Recovery and Relative Percent Difference (RPD);
- Matrix Spike/Matrix Spike Duplicate (MS/MSD) Recovery and RPD;
- Duplicates (field duplicate, laboratory duplicate); and
- Other Applicable QC Parameters.

The data qualifiers used to qualify the analytical results associated with QC parameters outside of the established data quality objectives are defined below:

- J The analyte was positively identified; however, the result should be considered an estimated value.
- UJ The reporting limit is considered an estimated value.
- R Quality control indicates that the data is not usable.

Results qualified as "J" or UJ" are of acceptable data quality and may be used quantitatively to fulfill the objectives of the analytical program, per EPA guidelines.

Results for the performance monitoring events that required qualification based on the data verification are summarized in Table B.

# 2.1 CHAIN-OF-CUSTODY

The chain-of-custody documentation associated with project samples was found to be complete. Chain-of-custodies included sample identifications, date and time of collection, requested parameters, and relinquished/received signatures.

# 2.2 SAMPLE PRESERVATION AND TEMPERATURE UPON LABORATORY RECEIPT

Samples collected were received preserved and intact at the respective project laboratories. The samples were received by the laboratory at the correct temperature (4  $\pm$  2° Celsius) with the following exceptions:

- Samples collected on March 31, 2009 through April 2, 2009, and May 5th, 2009 were received as low as 1.0 degrees Celsius. The temperature outlier did not significantly impact the sample results; therefore, data qualification was not required.
- Samples collected on April 3, 2009 were received as low as 2.0 degrees Celsius. The temperature outlier did not significantly impact the sample results; therefore, data qualification was not required.

# 2.3 HOLDING TIMES

All samples were extracted and analyzed within method-specified holding time limits, except for the following:

• Equipment blank sample EBS-033109 exceeded EPA hold time for SVOC analysis extraction by 0.8 days. Data qualification was not needed because associated field samples were not analyzed for SVOCs.

# 2.4 BLANK CONTAMINATION

# 2.4.1 Method Blank

Method blanks were analyzed at the appropriate frequency. Target compounds were not detected in the method blanks except for the following:

- Methylene chloride was detected in the method blank for analytical batch 18762 at a concentration of 4.140 ug/kg-dry. Methylene chloride was detected above the laboratory reporting limit in associated soil sample IM 1-1 (8-10') at 6.03 ug/kg-dry, IM 1-1 (18-20') at 3.50 ug/kg-dry, IM 1-1 (20-22') at 2.88 ug/kg-dry, IM 1-2 (4-6') at 3.97 ug/kg-dry, IM 1-2 (16-18') at 4.18 ug/kg-dry, IM 1-3 (4-6') at 4.40 ug/kg-dry and IM 1-3 (15-17') at 4.19 ug/kg-dry. Detected concentrations for associated field samples were qualified "J+" to account for a potential high bias.
- Acetone was detected in the method blank for analytical batch R33094 at a concentration of 19.78 ug/L. Data qualification was not needed because all associated field samples were non-detect.

# 2.4.2 Trip Blank

Trip blanks were analyzed at the appropriate frequency as specified in the Order. Target compounds were not detected in the trip blanks.

## 2.4.3 Field Blanks/Equipment Rinsate Blank

Equipment rinsate blanks were performed at the appropriate frequency as specified in the Final Stipulated Order (NMED, 2007). Target compounds were not detected in the field blanks and equipment rinsate blank except for the following:

#### Field Blanks

- Bromodichloromethane (2.1 ug/L) and chloroform (17 ug/L) were detected in the field blank FB-033109 above the respective laboratory detection limits. Data qualification was not needed because all associated field sample results for bromodichloromethane and chloroform were non-detect.
- Bromodichloromethane (8.6 ug/L), chloroform (38 ug/L), and dibromochloromethane (1.0 ug/L) were detected in the field blank FB-040109 above the respective laboratory detection limits. Data qualification was not needed because all associated field sample results for bromodichloromethane, chloroform, and dibromochloromethane were non-detect.
- Bromodichloromethane (2.0 ug/L) and chloroform (17 ug/L) were detected in the field blank FB-040209 above the respective laboratory detection limits. Data qualification was not needed because all associated field sample results for bromodichloromethane and chloroform were non-detect.
- Bromodichloromethane (1.2 ug/L) and chloroform (9.2 ug/L) were detected in the field blank FB-040309 above the respective laboratory detection limits. Data qualification was not needed because all associated field sample results for bromodichloromethane and chloroform were non-detect.
- Bromodichloromethane (2.2 ug/L), chloroform (6.3 ug/L), and dibromochloromethane (2.1 ug/L) were detected in the field blank FB-050509 above the respective laboratory detection limits. Data qualification was not needed because all associated field sample results for bromodichloromethane, chloroform, and dibromochloromethane were non-detect.

#### Equipment Rinsate Blanks

- Bromodichloromethane (1.3 ug/L) and chloroform (7.9 ug/L) were detected in the soil equipment blank EBS-033109 above the respective laboratory detection limits. Data qualification was not needed because all associated field sample results for bromodichloromethane and chloroform were non-detect.
- Bromodichloromethane (8.6 ug/L) and chloroform (43 ug/L) were detected in the soil equipment blank EBS-040109 above the respective laboratory detection limits. Data qualification was not needed because all associated field sample results for bromodichloromethane and chloroform were non-detect.
- Bromodichloromethane (1.8 ug/L) and chloroform (15 ug/L) were detected in the soil equipment blank EBS-040209 above the respective laboratory detection



limits. Data qualification was not needed because all associated field sample results for bromodichloromethane and chloroform were non-detect.

- Bromodichloromethane (1.2 ug/L) and chloroform (9.2 ug/L) were detected in the soil equipment blank EBS-040309 above the respective laboratory detection limits. Data qualification was not needed because all associated field sample results for bromodichloromethane and chloroform were non-detect.
- Bromodichloromethane (2.0 ug/L) and chloroform (19 ug/L) were detected in the field blank EBW-050509 above the respective laboratory detection limits. Data qualification was not needed because all associated field sample results for bromodichloromethane and chloroform were non-detect.

#### 2.4.4 Common Laboratory Contaminants

Per USEPA guidelines, common laboratory contaminants for VOC analysis are acetone, 2-butanone (MEK), cyclohexane, and methylene chloride. Common laboratory contaminants for SVOC analysis include phthalates. Analytical results were qualified for respective sample batches that corresponded to a methylene chloride detection in the method blank. The samples were qualified because the laboratory confirmed the detection is likely a result of an air circulation issue at the lab. Refer to Section 2.4.1 for samples qualified due to method blank detections most likely resulting from common laboratory contamination.

# 2.5 SURROGATE RECOVERY

Surrogate recoveries for the organic and inorganic analyses were performed at the required frequency and were within laboratory acceptance limits, with the following exceptions:

- Surrogate recovery for 2,4,6-Tribromophenol (29.4%) and Phenol-d5 (37.5%) were below the lower acceptance limits of 35.5% and 37.6%, respectively, for soil sample IM 1-3 (4-6'). Data qualification was not required because four other acid and base/neutral fractions were within acceptance limits.
- Surrogate recovery for BFB (179%) was above the upper acceptance limit of 123% for soil sample IM 1-1 (8-10'). The associated gasoline range organics (GRO) sample result was qualified "J+" for sample IM 1-1 (8-10') due to potential high bias.
- Surrogate recoveries for 4-Bromofluorobenzene (149%), Dibromofluoromethane (119%), and Toluene-d8 (120%) were above the upper acceptance limits of 111%, 105%, and 113%, respectively, for soil sample IM 1-1 (8-10'). Associated sample results were qualified "J+" to account for a potential high bias.
- Surrogate recoveries for Dibromofluoromethane (61.2%) were below the lower acceptance limit of 70% for soil sample IM 1-1 (8-10'). Data was not qualified since the other surrogates were recovered within acceptance limits.

- Surrogate recoveries for 2-Fluorophenol (24.1%), and Phenol-d5 (32.9%) were below the respective lower acceptance limits (28.1% and 37.6%) for sample IM 1-1 (8-10'). Data qualification was not required because the two other acid and base/neutral fractions were within acceptance limits.
- Surrogate recovery for BFB (156%) was above the upper acceptance limit of 123% for soil sample IM 1-1 (18-20'). Data qualification was not required because the associated sample result was non-detect.
- Surrogate recoveries for Dibromofluoromethane (110%) and Toluene-d8 (126%) were above the upper acceptance limits of 110% and 126%, respectively, for soil sample IM 1-1 (18-20'). Data was not qualified since the other surrogates were recovered within acceptance limits.
- Surrogate recovery for BFB (210%) was above the upper acceptance limit of 123% for soil sample IM 1-1 (20-22'). The associated field sample result was qualified "J+" due to potential high bias.
- Surrogate recoveries for Dibromofluoromethane (109%) was above the upper acceptance limits of 110% for soil sample IM 1-1 (20-22'). Data was not qualified since the other surrogates were recovered within acceptance limits.
- Surrogate recoveries for Dibromofluoromethane (67.8%) was below the lower acceptance limits of 70% for soil sample IM 1-1 (20-22'). Data was not qualified since the other surrogates were recovered within acceptance limits.
- Surrogate recoveries for DNOP (0%) was below the lower acceptance limit of 61.7% for soil sample IM 1-4 (2-4'). The associated field sample results were qualified "J-" due to a potential low bias.
- Surrogate recovery for BFB (125%) was above the upper acceptance limit of 123% for soil sample IM 1-4 (2-4'). The associated field sample result was qualified "J+" due to potential high bias.
- Surrogate recoveries for 1,2-Dichloroethane-d4 (67%), 4-Bromofluorobenzene (47.7%), and Toluene-d8 (58.8%) were below the lower acceptance limits of 70% for soil sample IM 1-4 (18-19.5'). The associated field sample results were qualified "J-" for detected concentrations, and "UJ" for non-detects due to potential low bias.

# 2.6 LCS RECOVERY AND RPD

LCS/LCS duplicates were performed at the required frequency and were evaluated based on the following criteria:

• If the analyte recovery was above acceptance limits for the LCS or LCS duplicate, but the analyte was not detected in the associated batch, then data qualification was not required.

- If the analyte recovery was above acceptance limits for the LCS or LCS duplicate and the analyte was detected in the associated batch, then the analyte results were qualified "J".
- If the analyte recovery was below acceptance limits for LCS or LCS duplicate then the analyte results in the associated analytical batch were qualified ("UJ" for non-detects and "J" for detected results).
- If the analyte recovery was less than 10 percent, the analyte results in the associated analytical batch were rejected and qualified "R".

LCS/LCSD percent recoveries and RPDs were within acceptance limits except for the following:

• The LCS recovery for benzene (84.2%), 1,1-Dichloroethene (92.0%), and trichloroethene (86.6%) were below the lower acceptance limit of 88%, 97.9%, and 90.5%, respectively for analytical batch R33094. Data qualification was not needed since all associated samples were for equipment and field blanks.

# 2.7 MS/MSD RECOVERY AND RPD

MS/MSD samples were performed at the required frequency and were evaluated by the following criteria:

- If the MS or MSD recovery for an analyte was above acceptance limits but the analyte was not detected in the associated analytical batch, then data qualification was not required.
- If the MS or MSD recovery for an analyte was above acceptance limits and the analyte was detected in the associated analytical batch, then analyte results were qualified "J".
- Low MS/MSD recoveries for inorganic parameters result in sample qualification of the associated analytical batch.
- Low MS/MSD recoveries for organic parameters result in the data qualification of the unspiked sample rather than the analytical batch.
- Results were not qualified based on non-project specific MS/MSD (i.e., batch QC) recoveries.

MS/MSD percent recoveries and RPDs were within acceptance limits.

## 2.8 DUPLICATES

#### 2.8.1 Field Duplicates

Field duplicates were collected at a rate of 10 percent and submitted for analysis. The RPDs between the field duplicate and its associated sample were calculated and are presented in Table C. The field duplicates were evaluated by the following criteria:

- If an analyte was detected at a concentration greater than five times the method reporting limit, the RPD should be less than 35percent for soil and 25 percent for groundwater samples.
- If an analyte was detected at a concentration that is less than five times the method reporting limit, then the difference between the sample and the field duplicate should not exceed the method reporting limit.
- Duplicate RPDs are calculated by dividing the difference of the concentrations by the average of the concentrations.

Field duplicate RPDs were within acceptance limits.

# 2.9 OTHER APPLICABLE QC PARAMETERS

#### 2.9.1 Calibration

The Method 8260B continuing calibration verification (CCV) standards were within acceptance limits.

# 3.0 COMPLETENESS SUMMARY

Two types of completeness were calculated for this project: contract and technical. The following equations were used to calculate the two types of completeness:

% Contract Completeness =  $\left(\frac{\text{Number of contract compliant results}}{\text{Number of reported results}}\right) \times 100$ 

% Technical Completeness = 
$$\left(\frac{\text{Number of usable results}}{\text{Number of reported results}}\right) \times 100$$

The overall contract completeness, which includes the evaluation of protocol and contract deviations, and includes the evaluation of the QC parameters listed in Section 2.0, was 98 percent for sample analysis (85 out of a total 1,713 results required qualification). The technical completeness attained for North and South Aeration Lagoon Closure activities was 100 percent. The completeness results are provided in Table D. The results for the performance monitoring events were considered usable for the intended purposes and the project DQOs have been met.



TABLE ASampling and Analysis Schedule



Western Refining Southwest, Inc. Bloomfield Refinery

÷

Page 9

North and South Aeration Lagoon Closure Report September 2009

#### Table A Sampling and Analysis Schedule

# North and South Aeration Lagoon Closure Report Western Refining Southwest, Inc. - Bloomfield Refinery

Sample ID	Lab ID	Date Collected	Sample Type	Parameters	
IM 1-4 (2-4')	904085-01	4/3/2009	N	VOCs, SVOCs, TPHs, Metals, Mercury	
IM 1-4 (18-19.5')	904085-02	4/3/2009	N	VOCs, TPHs, Metals, Mercury	
FB-040309	904085-03	4/3/2009	FB	VOCs, SVOCs, TPHs, Metals, Mercury	
EBS-040309	904085-04	4/3/2009	EB-Soil	VOCs, TPHs	
FB-040209	904085-05	4/2/2009	FB	VOCs, SVOCs, TPHs, Metals, Mercury	
EBS-040209	904085-06	4/2/2009	EB-Soil	VOCs, SVOCs, TPHs, Metals, Mercury	
Methanol Blank	904085-07	4/2/2009	MB	VOCs, TPHs	
Trip Blank	904085-08	4/2/2009	TB	VOCs, TPHs	
EBS-0403609	904085-09	4/3/2009	EB-Soil	Metals, Mercury	
IM 1-3 (15-17')	904032-01	3/31/2009	N	VOCs. TPHs, Metals. Mercury	
IM 1-3 (4-6')	904032-02	3/31/2009	N	VOCs, SVOCs, TPHs, Metals, Mercury	
EBS-033109	904032-03	3/31/2009	EB-Soil	VOCs, SVOCs, TPHs, Metals, Mercury	
Methanol Blank	904032-04	3/31/2009	MB	VOCs. TPHs	
Trip Blank	904032-05	3/31/2009	TB	VOCs, TPHs	
FB-033109	904032-06	3/31/2009	FB	VOCs, SVOCs, TPHs, Metals, Mercury	
EBS-0401009	904060-01	4/1/2009	EB-Soil	VOCs, SVOCs, TPHs, Metals, Mercury	
FB-040109	904060-02	4/1/2009	FB	VOCs, SVOCs, TPHs, Metals, Mercury	
Trip Blank	904060-03	4/1/2009	TB	VOCs, TPHs	
IM1-2 (16-18')	904060-04	4/1/2009	N	VOCs, TPHs, Metals, Mercury	
IM1-2 (4-6')	904060-05	4/1/2009	N	VOCs, TPHs, Metals, Mercury	
Methanol Blank	904060-06	4/1/2009	MB	VOCs, TPHs	
IM 1-1 (8-10')	904060-07	4/2/2009	N	VOCs, TPHs, Metals, Mercury	
IM 1-1 (18-20')	904060-08	4/2/2009	N	VOCs, TPHs, Metals, Mercury	
IM 1-1 (20-22')	904060-09	4/2/2009	N	VOCs, TPHs, Metals, Mercury	
EBW-050509	905067-01	5/5/2009	EB-Soil	VOCs, SVOCs, TPHs, Metals, Mercury, General Chemistry	
FB-050509	905067-02	5/5/2009	FB	VOCs, SVOCs, TPHs, Metals, Mercury, General Chemistry	
MW-58	905067-03	5/5/2009	N	VOCs. SVOCs, TPHs, Metals, Mercury, General Chemistry	
MW-57	905067-04	5/5/2009	N	VOCs, SVOCs, TPHs, Metals, Mercury, General Chemistry	
Trip Blank	905067-05	5/5/2009	ТВ	VOCs, TPH	
MW-57 (DUP)	905067-06	5/5/2009	N-dup	VOCs, SVOCs, TPHs, Metals, Mercury, General Chemistry	
MW-56	905113-01	5/5/2009	N	VOCs. SVOCs, TPHs, Metals, Mercury, General Chemistry	
MW-55	905113-02	5/5/2009	N	VOCs, SVOCs, TPHs, Metals, Mercury, General Chemistry	
Trip Blank	905113-03	5/5/2009	TB	VOCs, TPH	

Notes: VOCs = Volatile Organic Compounds N = Normal field sample FD = Field duplicate

TB = Trip Blank EB = Equipment Blank MB = Methanol Blank

FB = Field Blank TPH = Total Petroleum Hydrocarbons

TABLE BQualified Data

Western Refining Southwest, Inc. Bloomfield Refinery

North and South Aeration Lagoon Closure Report September 2009

#### Table B Qualified Data

Sample ID	Date Collected	Analyte	Result	Units	Matrix	Qualifier	Comments
IM 1-4 (18-19.5')	4/3/2009	1,1,1,2-Tetrachloroethane	< 0.949	µg/Kg-dry	Soil	ບງ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,1,1-Trichloroethane	< 0.949	µg/Kg-dry	Soil	UJ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,1,2,2-Tetrachloroethane	< 0.949	µg/Kg-dry	Soil		Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,1,2-Trichloroethane	< 0.949	µg/Kg-dry	Soil	ເບ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,1-Dichloroethane	< 0.949	µg/Kg-dry	Soil	UJ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,1-Dichloroethene	< 0.949	µg/Kg-dry	Soil	ເບ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,1-Dichloropropene	< 0.949	µg/Kg-dry	Soil	ເປ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,2,3-Trichlorobenzene	< 0.949	µg/Kg-dry	Soil	ເບ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,2,3-Trichloropropane	< 0.949	µg/Kg-dry	Soil	ບມ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,2,4-Trichlorobenzene	< 0.949	µg/Kg-dry	Soil	ບ	Qualified due to low surrogate recovery.
IM 1-1 (8-10')	4/2/2009	1,2,4-Trimethylbenzene	6.0	mg/Kg	Soil	J+	Qualified due to high surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,2,4-Trimethylbenzene	2.52	µg/Kg-dry	Soil	J.	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,2-Dibromo-3-chloropropane	< 0.949	µg/Kg-dry	Soil	ືພ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,2-Dibromoethane (EDB)	< 0.949	µg/Kg-dry	Soil	נט	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,2-Dichlorobenzene	< 0.949	µg/Kg-dry	Soil	ເມ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,2-Dichloroethane (EDC)	< 0.949	µg/Kg-dry	Soil	បរ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,2-Dichloropropane	< 0.949	µg/Kg-dry	Soil	UJ	Qualified due to low surrogate recovery.
IM 1-1 (8-10')	4/2/2009	1,3,5-Trimethylbenzene	0.17	mg/Kg	Soil	J+	Qualified due to high surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,3,5-Trimethylbenzene	< 0.949	µg/Kg-dry	Soil	ເບ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,3-Dichlorobenzene	< 0.949	µg/Kg-dry	Soil	យ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,3-Dichloropropane	< 0.949	µg/Kg-dry	Soil	ບມ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	1,4-Dichlorobenzene	< 0.949	µg/Kg-dry	Soil	បរ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	2,2-Dichloropropane	< 0.949	µg/Kg-dry	Soil	UJ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	2-Butanone	5.55	µg/Kg-dry	Soil	J.	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	2-Chlorotoluene	< 0.949	µg/Kg-dry	Soil	UJ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	2-Hexanone	< 3.80	µg/Kg-dry	Soil	បរ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	4-Chlorotoluene	< 0.949	µg/Kg-dry	Soil	ບມ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	4-Isopropyltoluene	< 0.949	µg/Kg-dry	Soil	U	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	4-Methyl-2-pentanone	< 3.80	µg/Kg-dry	Soil	ມ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Acetone	26.6	µg/Kg-dry	Soil	J-	Qualified due to low surrogate recovery.
IM 1-1 (8-10')	4/2/2009	Benzene	0.20	mg/Kg	Soil	J+	Qualified due to high surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Benzene	< 0.949	µg/Kg-dry	Soil	បរ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Bromobenzene	< 0.949	µg/Kg-dry	Soil	UJ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Bromodichloromethane	< 0.949	µg/Kg-dry	Soil	ហ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Bromoform	< 0.949	µg/Kg-dry	Soil	ບມ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Bromomethane	< 0.949	µg/Kg-dry	Soil	UJ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Carbon disulfide	< 3.80	µg/Kg-dry	Soil	U	Qualified due to low surrogate recovery.
LM 1-4 (18-19.5')	4/3/2009	Carbon tetrachloride	< 0.949	µg/Kg-dry	Soil	ບມ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Chlorobenzene	< 0.949	µg/Kg-dry	Soil	ហ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Chloroethane	< 0.949	µg/Kg-dry	Soil	ບາ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Chloroform	< 0.949	µg/Kg-dry	Soil	ហ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Chloromethane	< 0.949	µg/Kg-dry	Soil	ບ	Qualified due to low surrogate recovery.
IM 1 4 (19 10 5)	4/2/0000	cic 1 2-DCE	~0.040	un/Kadry	Soil	1 10	Qualified due to low surrorate recovery





Table	В
Qualified	Data

# North and South Aeration Lagoon Closure Report Western Refining Southwest, Inc. - Bloomfield Refinery

Sample ID	Date Collected	Analyte	Result	Units	Matrix	Qualifier	Comments
IM 1-4 (18-19.5')	4/3/2009	cis-1.3-Dichloropropene	< 0.949	ug/Kg-drv	Soil	ເບ	Oualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Dibromochloromethane	< 0.949	µg/Kg-dry	Soil	បរ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Dibromomethane	< 0.949	µg/Kg-dry	Soil	បរ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Dichlorodifluoromethane	< 0.949	µg/Kg-dry	Soil	UJ	Qualified due to low surrogate recovery.
IM 1-4 (2-4')	4/3/2009	Diesel Range Organics (DRO)	5700	mg/Kg	Soil	J.	Qualified due to low surrogate recovery.
IM 1-1 (8-10')	4/2/2009	Ethylbenzene	3.5	mg/Kg	Soil	J+	Qualified due to high surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Ethylbenzene	1.55	µg/Kg-dry	Soil	J.	Qualified due to low surrogate recovery.
IM 1-1 (20-22')	4/2/2009	Gasoline Range Organics (GRO)	35	mg/Kg	Soil	J+	Qualified due to high surrogate recovery
IM 1-1 (8-10')	4/2/2009	Gasoline Range Organics (GRO)	92	mg/Kg	Soil	J+	Qualified due to high surrogate recovery
IM 1-4 (2-4')	4/3/2009	Gasoline Range Organics (GRO)	26	mg/Kg	Soil	J+	Qualified due to high surrogate recovery
IM 1-4 (18-19.5')	4/3/2009	Hexachlorobutadiene	< 0.949	µg/Kg-dry	Soil	ເບ	Qualified due to low surrogate recovery.
IM 1-1 (8-10')	4/2/2009	Isopropylbenzene	0.56	mg/Kg	Soil	J+	Qualified due to high field duplicate RPD
IM 1-4 (18-19.5')	4/3/2009	Isopropylbenzene	< 0.949	µg/Kg-dry	Soil	UJ	Qualified due to low MS/MSD recovery
IM 1-4 (18-19.5')	4/3/2009	Methyl tert-butyl ether (MTBE)	2.63	µg/Kg-dry	Soil	j.	Qualified due to low MS recoveries
IM 1-1 (18-20')	4/2/2009	Methylene chloride	3.50	µg/Kg-dry	Soil	J+	Qualified due to detection in associated method blank
LM 1-1 (20-22')	4/2/2009	Methylene chloride	2.88	µg/Kg-dry	Soil	j+	Qualified due to detection in associated method blank
IM 1-1 (8-10)	4/2/2009	Methylene chloride	6.03	µg/Kg-dry	Soil	Ĵ+	Qualified due to detection in associated method blank
EM 1-2 (4-6')	4/1/2009	Methylene chloride	3.97	µg/Kg-dry	Soil	J+	Qualified due to detection in associated method blank
IM 1-2 (16-18')	4/1/2009	Methylene chloride	4.18	µg/Kg-dry	Soil	J+	Qualified due to detection in associated method blank
IM 1-3 (15-17')	3/31/2009	Methylene chloride	4.19	µg/Kg-dry	Soil	J+	Qualified due to detection in associated method blank
IM 1-3 (4-6')	3/31/2009	Methylene chloride	4.40	µg/Kg-dry	Soil	J+	Qualified due to detection in associated method blank
IM 1-4 (18-19.5')	4/3/2009	Methylene chloride	2.03	µg/Kg-dry	Soil	J+	Qualified due to detection in associated method blank
IM 1-4 (2-4")	4/3/2009	Methylene chloride	1.79	µg/Kg-dry	Soil	J+	Qualified due to detection in associated method blank
IM 1-4 (2-4')	4/3/2009	Motor Oil Range Organics (MRO)	3700	mg/Kg	Soil	J.	Qualified due to low surrogate recovery.
IM 1-1 (8-10')	4/2/2009	Naphthalene	0.81	mg/Kg	Soil	J+	Oualified due to high surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Naphthalene	20.7	µg/Kg-dry	Soil	J-	Qualified due to low surrogate recovery.
IM 1-1 (8-10')	4/2/2009	n-Butylbenzene	1.2	mg/Kg	Soil	J+	Oualified due to high surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	n-Butylbenzene	2.43	µg/Kg-dry	Soil	J	Qualified due to low surrogate recovery.
IM 1-1 (8-10')	4/2/2009	n-Propylbenzene	2.2	mg/Kg	Soil	J+	Qualified due to high surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	n-Propylbenzene	1.87	µg/Kg-dry	Soil	J-	Qualified due to low surrogate recovery.
IM 1-1 (8-10')	4/2/2009	sec-Butylbenzene	0.46	mg/Kg	Soil	J+	Qualified due to high surrogate recovery
IM 1-4 (18-19.5')	4/3/2009	sec-Butylbenzene	1.66	µg/Kg-dry	Soil	J.	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Styrene	< 0.949	µg/Kg-dry	Soil	ບມ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	tert-Butylbenzene	< 0.949	µg/Kg-dry	Soil	tu l	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Tetrachloroethene (PCE)	< 0.949	µg/Kg-dry	Soil	ບ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Toluene	< 0.949	µg/Kg-dry	Soil	ບ່	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	trans-1,2-DCE	< 0.949	µg/Kg-dry	Soil	បរ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	trans-1,3-Dichloropropene	< 0.949	µg/Kg-dry	Soil	ເບ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Trichloroethene (TCE)	< 0.949	µg/Kg-dry	Soil	LU I	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Trichlorofluoromethane	< 0.949	µg/Kg-dry	Soil	UJ.	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Vinyi chloride	< 0.949	µg/Kg-dry	Soil	UJ	Qualified due to low surrogate recovery.
IM 1-4 (18-19.5')	4/3/2009	Xylenes, Total	< 0.949	µg/Kg-dry	Soil	ເບ	Qualified due to low surrogate recovery.

Notes: mg/L = milligrams per liter ug/L = micrograms per liter UJ = Estimated reporting limit J- = Potential low bias

J+ = Potential high bias RPD = Relative Percent Difference MS/MSD = Matrix spike / Matrix spike duplicate



TABLE CField Duplicate Summary



Western Refining Southwest, Inc. Bloomfield Refinery

#### Table C Field Duplicate Summary

#### North and South Aeration Lagoon Closure Report Western Refining Southwest, Inc. - Bloomfield Refinery

	Parameter	MW-57 Sample Result	MW-57 (DUP) Field Duplicate	RPD (%)	
PH (mg/L):	Diesel Range Organics (DRO)	5.0	4.7	6.2	
	Motor Oil Range Organics (MRO)	< 5.0	< 5.0	NC	
	Gasoline Range Organics (GRO)	8.0	7.3	9.2	
Cs (ug/L)	1.1.1.2-Tetrachloroethane	< 5.0	< 5.0	NC	
	1.1.1-Trichloroethane	< 5.0	< 5.0	NC	
	1.1.2.2-Tetrachloroethane	< 10	< 10	NC	
	1.1.2-Tricbloroethane	< 5.0	< 5.0	NC	
	1 1-Dichloroethane	< 50	< 5.0	NC	
	1 1 Dichloroethene	< 5.0	< 5.0	NC	
	1,1-Dichlosopopopo	< 5.0	< 5.0	NC	
		< 5.0	< 5.0		
	1,2,3-1 richlorobenzene	< 5.0	< 5.0	NC	
	1,2,3-Trichloropropane	< 10	< 10	NC	
	1,2,4-Trichlorobenzene	< 5.0	< 5.0	NC	
	1,2,4-Trimethylbenzene	270	280	3.6	
	1,2-Dibromo-3-chloropropane	< 10	< 10	NC	
	1,2-Dibromoethane (EDB)	< 5.0	< 5.0	NC	
	1,2-Dichlorobenzene	< 5.0	< 5.0	NC	
	1,2-Dichloroethane (EDC)	< 5.0	< 5.0	NC	
	1,2-Dichloropropane	< 5.0	< 5.0	NC	
	1.3.5-Trimethylbenzene	36	37	2.7	
	1.3-Dichlorobenzene	< 5.0	< 5.0	NC	
	1 3-Dichloropropage	< 5.0	<u>~~</u>	NC	
	1.4-Dichlorobenzene	250	~	NC	
	1.4-Dicinorobenzene	200	250	120	
	1-Memyinapitnajene	220	200	12.8	
	2,2-Dichloropropane	< 10	< 10	NC	
	2-Butanone	< 50	< 50	NC	
	2-Chlorotoluene	< 5.0	< 5.0	NC	
	2-Hexanone	< 50	< 50	NC	
	2-Methylnaphthalene	250	290	14.8	
	4-Chlorotoluene	< 5.0	< 5.0	NC	
	4-Isopropyltoluene	9.2	9.4	2.2	
	4-Methyl-2-pentanone	< 50	< 50	NC	
	Acetone	< 50	54	7.7	
	Benzene	1100	910	18.9	
	Bromobenzene	< 5.0	~ 5.0	NC	
	Bromodichloromathana	< 5.0	< 5.0	NC	
	Dermodernoronieurane		< 5.0		
	Bromotorm	< 5.0	< 5.0	NC	
	Bromomethane	< 5.0	< 5.0	NC	
	Carbon disulfide	< 50	< 50	NC	
	Carbon Tetrachloride	< 5.0	< 5.0	NC	
	Chlorobenzene	< 5.0	< 5.0	NC	
	Chloroethane	< 10	< 10	NC	
	Chloroform	< 5.0	< 5.0	NC	
	Chloromethane	< 5.0	< 5.0	NC	
	cis-1,2-DCE	< 5.0	< 5.0	NC	
	cis-1 3-Dichloropropene	< 50	< 5.0	NC	
	Dibromochloromethane	250	~ 50	NC	
	Dibromomathana	~ 5.0	<u> </u>	NC	
	Dichlorodiflussemethers	< 3.0	<u> </u>	NC	
	Ethelhorgene	× 3.0	< J.U 200		
		200	200	0.0	
	Hexachiorobuladiene	< 0.0	0.0		
	Isopropylbenzene	4/	40	2.2	
	Methyl tert-butyl ether (MTBE)	1900	1700	11.1	
	Methylene Chloride	< 15	< 15	NC	
	Naphthalene	380	430	12.3	
	n-Butylbenzene	16	16	0.0	
	n-Propylbenzene	78	81	3.8	
	sec-Butylbenzene	12	13	8.0	
	Styrene	< 5.0	< 5.0	NC	
	tert-Butylbenzene	< 5.0	< 5.0	NC	
	Tetrachloroethene (PCE)	< 5.0	< 5.0	NC	
	Toluene	250		NC	
	trong 1.2 DCF	~	~ 5.0	NC	
		< 5.0	<u> </u>		
	urans-1,3-Dichloropropene	<u> </u>	< 5.0	NC	
	Trichloroethene (TCE)	< 5.0	< 5.0	NC	
	Trichlorofluoromethane	< 5.0	< 5.0	NC	
	Vinyl chloride	< 5.0	< 5.0	NC	
	Xylenes, Total	78	78	0.0	

· * '

#### Table C Field Duplicate Summary

	Parameter	MW-57	MW-57 (DUP)	RPD
		Sample Result	Field Duplicate	(%)
OCs (ug/L):	1,2,4-Trichlorobenzene	< 20	< 20	NC
	1,2-Dichlorobenzene	< 20	< 20	NC
	1,3-Dichlorobenzene	< 20	< 20	NC
	1,4-Dichlorobenzene	< 20	< 20	NC
	2,4,5-Trichlorophenol	< 20	< 20	NC
	2,4,6-Trichlorophenol	< 20	< 20	NC
	2,4-Dichlorophenol	< 40	< 40	NC
	2,4-Dimethylphenol	< 20	< 20	NC
	2,4-Dinitrophenol	< 40	< 40	NC
	2,4-Dinitrotoluene	< 20	< 20	NC
	2.6-Dinitrotoluene	< 20	< 20	NC
	2-Chloronaphthalene	< 20	< 20	NC
	2-Chlorophenol	< 20	< 20	NC
	2-Methylnaphthalene	100	100	0.0
	2-Methylphenol	<u> </u>	< 20	NC
	2 Nitroaniline	< 20	< 20	NC
	2 Nitrochanal	< 20	< 20	NC
	2-Nitrophenoi	< 20	< 20	
	3,3'-Dichlorobenzidine	< 20	< 20	NC NC
	3+4-Methylphenol	< 20	< 20	NC
	3-Nitroanifine	< 20	< 20	NC
	4,6-Dinitro-2-methylphenol	< 40	< 40	NC
	4-Bromophenyl phenyl ether	< 20	< 20	NC
	4-Chloro-3-methylphenol	< 20	< 20	NC
	4-Chloroaniline	< 20	< 20	NC
	4-Chlorophenyl phenyl ether	< 20	< 20	NC
	4-Nitroaniline	< 20	< 20	NC
	4-Nitrophenol	< 20	< 20	NC
	Acenaphthene	< 20	< 20	NC
	Acenanhthylene	< 20	< 20	NC
	Aniline	< 20	< 20	NC
	Anthroene	< 20	< 20	NC
	Anuracene	= 20	< 20	
	Azobenzene	< 20	< 20	NC
	Benz(a)anthracene	< 20	< 20	NC
	Benzo(a)pyrene	< 20	< 20	NC
	Benzo(b)fluoranthene	< 20	< 20	NC
	Benzo(g,h,i)perylene	< 20	< 20	NC
	Benzo(k)fluoranthene	< 20	< 20	NC
	Benzoic acid	< 40	< 40	NC
	Benzyl alcohol	< 20	< 20	NC
	Bis(2-chloroethoxy)methane	< 20	< 20	NC
	Bis(2-chloroethyl)ether	< 20	< 20	NC
	Bis(2-chloroisopropyl)ether	< 20	< 20	NC
	Bis(2-ethylbexyl)phthalate	< 20	< 20	NC
	Butyl benzyl obthalate	< 20	< 20	NC
	Carbazole	- 20	< 20	NC
	Chargene	- 20	~ 20	NC
	Dihang(a b)graduation	< 20	< 20	
	Dibenz(a,n)anthracene	< 20	< 20	
	Dibenzofuran	< 20	< 20	NC NC
	Dietnyi phinalate	< 20	< 20	NC NC
	Dimethyl phthalate	< 20	< 20	NC
	Di-n-butyl phthalate	< 20	< 20	NC NC
	Di-n-octyl phthalate	< 20	< 20	NC
	Fluoranthene	< 20	< 20 .	NC
	Fluorene	< 20	< 20	NC NC
	Hexachlorobenzene	< 20	< 20	NC
	Hexachlorobutadiene	< 20	< 20	NC
	Hexachlorocyclopentadiene	< 20	< 20	NC
	Hexachloroethane	< 20	< 20	NC
	Indeno(1,2,3-cd)pyrene	< 20	< 20	NC
	Isophorone	< 20	< 20	NC
	Naphthalene	150	150	0.0
	Nitrobenzene	< 20	< 20	NC
	N-Nitrosodimethylamine	< 20	< 20	NC
	N-Nitrosodi n-propulamine	- 20	< 20	NC
	N Nitrocodinhanylamina	- 20	~ 20	
		<u> </u>	<u>&lt; 20</u>	
	Pentachiorophenol	< 40	< 40	NC
	Phenanthrene	< 20	< 20	I NC
	Phenol	< 20	< 20	NC NC
	Pyrene	< 20	< 20	NC
	Deridine	< 20	~ 20	I NC

#### Table C Field Duplicate Summary

#### North and South Aeration Lagoon Closure Report Western Refining Southwest, Inc. - Bloomfield Refinery

	Description	MW-57	MW-57 (DUP)	RPD
	rarameter	Sample Result	Field Duplicate	(%)
Total Metals (mg/L):	Arsenic	< 0.020	< 0.020	NC
	Barium	1.8	1.7	5.7
	Cadmium	< 0.0020	< 0.0020	NC
	Calcium	120	120	0.0
	Chromium	< 0.0060	< 0.0060	NC
	Iron	3.7	3.7	0.0
	Lead	< 0.0050	< 0.0050	NC
	Magnesium	47	46	2.2
	Manganese	3.5	3.4	2.9
	Мегсшту	< 0.00020	< 0.00020	NC
	Potassium	5.0	5.0	0.0
	Selenium	< 0.25	< 0.25	NC
	Silver	< 0.0050	< 0.0050	NC
	Sodium	420	420	0.0
Dissolved Metals (mg/L):	Calcium	120	120	0.0
	Iron	0.32	0.30	6.5
	Magnesium	44	44	0.0
	Potassium	4.7	4.7	0.0
	Sodium	410	410	0.0
General Chemistry (mg/L)	Specific Conductance *	2100	2100	0.0
	Total Dissolved Solids	1700	1600	6.1
	Chloride	260	260	0.0
	Fluoride	0.56	0.57	1.8
	Nitrogen, Nitrate (As N)	0.14	0.15	6.9
	Nitrogen, Nitrite (As N)	< 1.0	< 1.0	NC
	Phosphorus, Orthophosphate (As P)	< 0.50	< 0.50	NC
	Sulfate	9.3	9.3	0.0
	Alkalinity, Total (As CaCO3)	1100	1100	0.0
	Bicarbonate	1100	1100	0.0
	Carbonate	< 2.0	< 2.0	NC

#### Notes:

RPD = Relative percent difference; [(difference)/(average)]* 100 NC = Not calculated; RPD values were not calculated for non-detects ug/L = micrograms per liter mg/kg = milligrams per liter mg/kg = milligrams per kilogram * = units are umhs/cm RPD Outlier = RPD limit > 25%


## ATTACHMENT 12 Quality Assessment / Quality Control Report

# TABLE DCompleteness Summaries



Western Refining Southwest, Inc. Bloomfield Refinery

Page 12

North and South Aeration Lagoon Closure Report September 2009

### Table D

## **Completeness Summaries**

#### North and South Aeration Lagoon Closure Report Western Refining Southwest, Inc. - Bloomfield Refinery

			Number of	Percent	Number of	Percent
	De se se stan	Total Number	Contractual	Compliance	Usable	Technical
TDU (mall)	Parameter Discol Pange Organics (DPO)	14		02 0		100
IPH (mg/L):	Meter Oil Pange Organics (MPO)	14	13 4	92.9	14	100
	Geoline Pange Organics (MRO)	14	11 h	78.6	14	100
	Gasonine Range Organics (ORO)	14	110	78.0	14	100
VOCs (ug/L):	All analytes	892	812 a c d	91.0	892	100
SVOCs (ug/L):	All analytes	557	557	100	557	100
Total Metals (mg/L):	Arsenic	14	14	100	14	100
	Barium	14	14	100	14	100
	Cadmium	14	14	100	14	100
	Calcium	5	5	100	5	100
	Chromium	14	14	100	14	100
	Iron	5	5	100	5	100
	Lead	14	14	100	14	100
	Magnesium	5	5	100	5	100
	Manganese	5	5	100	5	100
	Mercury	14	14	100	14	100
1	Potassium	5	5	100	5	100
	Selenium	14	14	100	14	100
	Sodium	5	5	100	5	100
	Silver	14	14	100	14	100
General Chemistry (mg/L):	Total Dissolved Solids	5	5	100	5	100
	Chloride	5	5	100	5	100
	Fluoride	5	5	100	5	100
	Nitrogen, Nitrate (As N)	5	5	100	5	100
	Nitrogen, Nitrite (As N)	5	5	100	5	100
	Phosphorus, Orthophosphate (As P)	5	5	100	5	100
	Sulfate	5	5	100	5	100
	Alkalinity, Total (As CaCO3)	5	5	100	5	100
	Bicarbonate	5	5	100	5	100
	Carbonate	5	5	100	5	100
	Specific Conductance (umhos/cm)	5	5	100	5	100
	Calcium	5	5	100	5	100
	Iron	5	5	100	5	100
	Magnesium	5	5	100	5	100
	Potassium	5	5	100	5	100
· ·	Sodium	5	5	100	5	100

#### Notes:

Number of samples used in completeness calculations includes field duplicates but does not include equipment rinsate, field, or trip blanks.

Percent Contractural Compliance = (number of contract compliant results / Number of reported results)*100

Percent Technial Compliance = (Number of usable results / Number of reported results) * 100

a = Qualified due to low surrogate recoveries.

b = Qualified due to high surrogate recoveries.

c = Qualified due to associated blank detection.

d = Qualified due to potential laboratory contamination.



